

Crown Pastoral Land Tenure Review

Lease name : GODLEY PEAKS

Lease number : PT 017

Preliminary Proposal

A Preliminary Proposal is advertised for public submissions as per Section 43 of the Crown Pastoral Land Act 1998.

The report attached is released under the Official Information Act 1982.

October

15

PROPOSAL FOR REVIEW OF CROWN LAND

Under Part 2 of the Crown Pastoral Land Act 1998

Date:

Parties

Holder:

Verity Farms NZ 29 Wallace St, Motueka 7120

Commissioner of Crown Lands:

Land Information New Zealand Private Bag 4721 Christchurch 8140 Attention Ashley Parris

The Land

Lease:	Godley Peaks
Legal Description:	Part Run 80 situated in Blocks VIII, XII & XVI Cook, Blocks V, IX & XIII Godley, Blocks IV & XIX Cass, Blocks I, II, V, VI, IX, X, XIII & XIV Tekapo North and Blocks I & II Tekapo Survey Districts.
Area:	14,493.0582 hectares (subject to survey)
Certificate of Title/Unique Identifier:	CB30B/914

Summary of Designations

Under this Proposal, the Land is designated as follows:

- (a) The Crown Land (shown shaded pink on the Plan) is to be restored to, or retained by, the Crown as set out in Schedules One and Two; and
- (b) The Freehold Land (shown shaded green and yellow on the Plan) is to be disposed by freehold disposal to the Holder as set out in Schedule Three.

1 The Plan

2 Conditions

2.1 This Proposal, and any agreement arising therefrom, is subject to the conditions contained in Schedule Four (if any).

3 Settlement

- 3.1 Unless otherwise agreed by the parties, the Settlement Date for the disposal of the Freehold Land to the Holder by freehold disposal will be the day that is TEN (10) working days following the day on which Land Information New Zealand notifies the Commissioner that the Final Plan and a copy of this Proposal are registered in accordance with the Act.
- 3.2 The Freehold Land will be disposed of to the Holder under the Land Act 1948.
- 3.3 Notwithstanding anything to the contrary, if, as at the Settlement Date (as determined pursuant to clause 3.1), the rent payable under the Lease is subject to a Rent Review, then the Commissioner may elect to:

(a) settle on the Settlement Date on the basis that the Commissioner may retain from the Commissioner's Payment an amount which the Commissioner, acting reasonably, estimates will be payable by the Holder to the Commissioner following agreement or determination of the Rent Review ("the Retention"). The Retention shall be held by the Crown Law Office in an on-call, interest-bearing trust account in the joint names of the parties for their respective rights and interests. Upon agreement or determination of the Rent Review, the Commissioner shall calculate the rent shortfall payable by the Holder to the Commissioner in respect of the period from the effective date of the Rent Review to the Settlement Date, both dates inclusive ("the Shortfall").If:

(i) the Shortfall is less than the Retention and the net interest earned thereon, the balance shall be paid by the Commissioner to the Holder within TEN (10) working days; or

(ii) the Shortfall is more than the Retention and the net interest earned thereon, the balance shall be paid by the Holder to the Commissioner within TEN (10) working days;

ог

(b) defer the Settlement Date until TEN (10) working days after the rent payable as a consequence of the Rent Review:

(i) has been agreed or determined; and

(ii) is not and will not be subject to any appeal, rehearing or other proceedings.

4 Holder's Payment

- 4.1 By 3.00 p.m. on the Settlement Date, the Holder must pay the Holder's Payment and all other money payable to the Commissioner or the duly appointed agent of the Commissioner by bank cheque without set-off or deduction of any kind in accordance with the settlement requirements of the Commissioner.
- 4.2 If the Holder fails to pay the Holder's Payment or any part of it or any other money to the Commissioner or to the duly appointed agent of the Commissioner on the Settlement Date clause 19 will apply.

5 Commissioner's Payment

- 5.1 The Commissioner shall pay the Commissioner's Payment to the Holder on the Settlement Date.
- 5.2 No interest shall be payable to the Holder by the Commissioner in respect of the Commissioner's Payment, including (without limitation) for the period from the Vesting Date to the Settlement Date.

6 Vesting of Crown Land

6.1 The Crown Land will vest in the Crown on the Vesting Date.

7 Issue of Certificate of Title

- 7.1 Notwithstanding any other provision in this Proposal, the Commissioner will not request that the Surveyor-General issue a certificate to the Registrar pursuant to section 116 of the Land Act 1948 (to enable a certificate of title to issue for the Freehold Land) unless and until:
 - (a) the Commissioner has received the Holder's Payment from the Holder under clause 4, and all other money payable by the Holder under this Proposal and the Notice;
 - (b) the Holder has provided to the Commissioner duplicate copies of the certificate of title relating to the Lease (if any) and/or the Lease if requested by the Commissioner;
 - (c) the Holder has signed and returned to the Commissioner all documents required by the Commissioner to be signed by the Holder to give effect to this Proposal (including, without limitation, any permit, covenant, easement and/or any other document); and
 - (d) the Holder has procured a registrable discharge of any Mortgage and provided this to the Commissioner together with any new mortgage documents to be registered against the Freehold Land.

8 Registration of Documents

8.1 Subject to clause 7, the Commissioner will lodge all documents necessary to give effect to this Proposal (including, without limitation any easement, covenant, discharge of mortgage, and/or duplicate copy of the Lease) and any new mortgage documents to be registered against the certificate of title to be issued for the Freehold Land so that the certificate of title for the Freehold Land will issue subject to the encumbrances provided in this Proposal. Any new mortgage will be registered after any other encumbrances such as any easements and/or covenants are registered.

9 Consents

- 9.1 The Holder must obtain the written consent to the Holder's acceptance of this Proposal from all persons having an interest in the Land (other than the Holder), including, but not limited to:
 - (a) any Mortgagee(s);
 - (b) any party entitled to the benefit of a land improvement agreement registered against the Lease and/or the Land; and
 - (c) any other person that the Commissioner reasonably believes has an interest in the Land or who the Holder reasonably believes has an interest in the Land, whether registered or not.
- 9.2 The consents required under clause 9.1 must be in a form acceptable to the Commissioner in all respects and be returned to the Commissioner with this Proposal on its acceptance by the Holder. Examples of the form of consents required under clause 9.1 are set out in Appendix 1.
- 9.3 The Holder must also obtain, and provide to the Commissioner if requested, all consents necessary for the Holder to accept this Proposal including (without limitation) any:
 - (a) corporate and/or trustee consents; and
 - (b) consent required under the Overseas Investment Act 2005.

- 9.4 The Holder will procure the Mortgagee to execute a registrable discharge of the Mortgage and, if required by the Mortgagee, the Holder will execute registrable new mortgage documents and forward these to the Commissioner to be registered as set out in clause 8.
- 9.5 If required by the Mortgagee, the Commissioner will provide an undertaking that, subject to the provisions of clause 7 being satisfied, the Commissioner will register the discharge of the Mortgage and register any new mortgage against the certificate of title for Freehold Land at the same time as the certificate of title for the Freehold Land issues.

10 Continuation of Lease

- 10.1 The Lease will remain in full force and effect until a certificate of title issues for the Freehold Land. Notwithstanding when Settlement Date occurs, until a certificate of title issues for the Freehold Land the Holder will duly and punctually comply with all obligations on its part under the Lease (other than as set out at clause 12.1 (b)) and the Lease will remain in full force and effect.
- 10.2 From the date that a certificate of title is issued for the Freehold Land the Lease is deemed to be surrendered and, subject to clause 10.3, the Commissioner releases and discharges the Holder from the performance and observance of all covenants, conditions and obligations under the Lease.
- 10.3 The release and discharge in clause 10.2:
 - (a) Is without prejudice to the rights, remedies and powers of the Commissioner contained in the Lease (except as varied in accordance with clause 12.1(b)); and
 - (b) will not release or discharge the Holder from any liability under the Lease,

arising prior to the date that the certificate of title for the Freehold Land is issued, under any statute or by any reason where such liability is due to the fault of the Holder.

10.4 As from the Vesting Date, the Holder will not have any estate, right or claim against any of the land, improvements, fencing, buildings, structures, fixtures, fittings or chattels on the Crown Land (subject to the provisions of any permit, easement, concession, other encumbrance or document provided under this Proposal). The Holder will not be entitled to any compensation for any of its improvements, fencing, buildings, structures, fixtures, fixtures, fittings or chattels which are on the Crown Land as at the Vesting Date.

11 Fencing and Construction Works

- 11.1 If the Holder has accepted this Proposal and that acceptance has taken effect pursuant to the Act, the Commissioner will, subject to clauses 11.2 and 14.4, erect at the Commissioner's cost new fencing:
 - (a) approximately along the line marked "New Fencing Line" on the Plan; and
 - (b) to the specifications in Appendix 3;

("the Fencing").

- 11.2 If the Fencing requires a resource consent or any other consent from any local or territorial authority ("the Fencing Consent"), the following provisions shall apply:
 - (a) The Commissioner shall use reasonable endeavours to obtain the Fencing Consent within 6 months of this Proposal taking effect pursuant to the Act.
 - (b) If the Fencing Consent:
 - (i) is not obtained within 6 months of this Proposal taking effect pursuant to the Act; and/or
 - (ii) is obtained on terms which are not satisfactory to the Commissioner in all respects;

the Commissioner may, acting reasonably, elect to do any one or more of the following:

- (iii) erect the Fencing in a position different from that shown on the Plan;
- (iv) erect the Fencing over a shorter distance than that shown on the Plan; or
- (v) erect the Fencing to specifications different from those in Appendix 3.
- 11.3 If the Commissioner has not completed the Fencing by the Settlement Date, the Holder agrees that the Commissioner may register a covenant, on terms entirely satisfactory to the Commissioner (in the Commissioner's sole discretion), over the Freehold Land to enable the Commissioner to complete the Fencing. The Holder will do all things necessary (including signing any document) to enable the Commissioner to register such a covenant.
- 11.4 The ongoing maintenance of the Fencing referred to in clauses 11.1 and 11.2 will be subject to the terms of the Fencing Act 1978.
- 11.5 If the Holder has accepted this Proposal and that acceptance has taken effect pursuant to the Act, the Commissioner will, subject to clause 11.6, undertake the construction works set out in Appendix 3 on the terms and conditions set out in Appendix 3 ("the Construction Works").
- 11.6 If any Construction Works for which the Commissioner is liable, or jointly liable with the Holder, require a resource consent or any other consent from any local or territorial authority ("the Works Consent"), the following provisions shall apply:
 - (a) The Commissioner shall use reasonable endeavours to obtain the Works Consent within 6 months of this Proposal taking effect pursuant to the Act.
 - (b) If the Works Consent:
 - (i) is not obtained within 6 months of this Proposal taking effect pursuant to the Act; and/or
 - (ii) is obtained on terms which are not satisfactory to the Commissioner in all respects;

the Commissioner may, acting reasonably, elect to vary the extent of the Construction Works in question and/or the terms and conditions upon which they are carried out.

12 Apportionments

- 12.1 Rent payable under the Lease in respect of the Freehold Land shall be apportioned as follows:
 - (a) Rent paid or payable will be apportioned on the Settlement Date as at the Settlement Date and either deducted from or added to (as the case may be) the amount required to settle.
 - (b) Notwithstanding that the Lease continues in effect until a certificate of title issues for the Freehold Land, the Holder shall not be required to pay any rent under the Lease for the Freehold Land from the Settlement Date.
- 12.2 Rent paid or payable under the Lease for the Crown Land will be apportioned on the Settlement Date as at the Vesting Date and either deducted from or added to (as the case may be) the amount required to settle.
- 12.3 All rates, levies, and all other incomings and outgoings and other charges receivable from or charged upon the Freehold Land will, unless otherwise agreed by the parties, be apportioned on the Settlement Date as at the Settlement Date.
- 12.4 All rates, levies and all other incomings and outgoings and other charges receivable from or charged upon the Crown Land will be apportioned on the Settlement Date as at the Vesting Date and either deducted from or added to (as the case may be) the amount required to settle.

12.5 Following the date that a certificate of title issues for the Freehold Land, the Commissioner will undertake a final apportionment and either the Commissioner will pay to the Holder, or the Holder will pay to the Commissioner, any additional amounts due because of any payments made or received by one party on behalf of the other for the period from the Settlement Date to the date on which a new certificate of title issues for the Freehold Land.

13 Risk

- 13.1 On and with effect from the Unconditional Date all risk of any nature in respect of the Freehold Land will pass from the Commissioner to the Holder. For the avoidance of doubt, the Holder's current risk in respect of matters arising under the Lease, including, without limitation, the Holder's risk in respect of all improvements, buildings, fencing, fixtures, fittings and chattels, will continue to remain with the Holder until the Lease is deemed to be surrendered under clause 10.2.
- 13.2 The Holder will be required to comply with its settlement obligations under this Proposal irrespective of any damage to, or destruction of, the Freehold Land prior to the Settlement Date.

14 Survey

- 14.1 All areas of the Land forming part of this Proposal and delineated on the Plan are approximate and subject to preparation of the Final Plan. The measurements of the areas may therefore alter on the Final Plan.
- 14.2 No error, misdescription or amendment of any part of the Land will annul, vary, or derogate from this Proposal, or the Holder's acceptance of this Proposal.
- 14.3 For the avoidance of doubt, the Holder will not be entitled to cancel or withdraw its acceptance of this Proposal, nor will the Holder, or any successor in title of the Holder or any party with an interest in the Land, be entitled to payment of any compensation, should any area of the Land on the Final Plan have a different measurement to the area specified in this Proposal.
- 14.4 The Commissioner does not warrant that any existing fence is erected on, or that any new fence to be erected will be on, any boundaries of the Land or any part of the Land as outlined on the Plan or the Final Plan.

15 Holder's Acknowledgements

- 15.1 If the Holder accepts this Proposal and that acceptance takes effect under the Act, the Holder acknowledges that:
 - (a) it is obtaining the freehold interest in the Freehold Land:
 - (i) "as is", solely in reliance on its own investigations and judgement; and
 - (ii) not in reliance on any representation or warranty made by the Commissioner, its employees, agents or any other person or persons directly or indirectly associated with the Commissioner;
 - (b) the Holder has carried out all inspections of the Freehold Land which the Holder considers necessary to satisfy itself as to all matters relating to the Freehold Land;
 - (c) the Holder, at its cost, is entirely responsible for all work to ensure that the Freehold Land complies with all applicable laws including (without limitation):
 - (i) the Resource Management Act 1991 any rule in any plan, resource consent or other requirement issued under the Resource Management Act 1991, and
 - (iii) the Building Act 2004; and

the Holder hereby indemnifies and will indemnify the Commissioner against all losses, damages and expenses incurred by the Commissioner and against all claims made against the Commissioner in respect of any work or costs for which the Holder is liable under this clause 15;

- (d) nothing in this Proposal is affected by, and the Commissioner has no liability of any nature in respect of, the existence or terms of any leases, licences or other occupation rights of any nature (if any) granted by the Holder in respect of the Land; and
- (e) the Holder has no claim (and will not have any claim) whatsoever against the Crown and/or Commissioner in relation to the Tenure Review and/or this Proposal, including (without limitation) any claim for any misrepresentation or for any loss or damage suffered whether in contract, tort (including negligence) or otherwise.

16 No Representations or Warranties by the Commissioner

- 16.1 The Commissioner gives no representations or warranties of any nature in respect of the Freehold Land. Without limitation, the Commissioner does not warrant:
 - (a) the accuracy of any matter in the Notice or this Proposal or in any notice, or any correspondence or other information provided to the Holder by the Commissioner or by any agent, contractor or employee of the Commissioner; or
 - (b) that the Freehold Land is or will remain suitable for the Holder's use; or
 - (c) that the Freehold Land complies with all or any statutory, territorial authority or any other legal requirements affecting or relevant to the Freehold Land.

17 Acceptance

- 17.1 The Holder's acceptance of this Proposal is irrevocable and constitutes a binding agreement between the Commissioner and the Holder.
- 17.2 If the Commissioner does not receive an acceptance to this Proposal from the Holder within three (3) months of putting it (in its substantive form) to the Holder, the Holder is deemed to have rejected this Proposal.

18 Solicitors Certificate

- 18.1 The Holder must procure the Holder's solicitors to provide the Commissioner with a solicitor's certificate (in a form satisfactory to the Commissioner, in its reasonable opinion) relating to such matters as the Holder's execution of this Proposal and the Holder's execution of any documents required to give effect to this Proposal (including, without limitation any easement, protective mechanism and/or concession). An example of the form of solicitors certificate required is set out at Appendix 2.
- 18.2 The Holder must return the completed solicitor's certificate to the Commissioner with this Proposal on its acceptance by the Holder.

19 Default

- 19.1 If from any cause whatever (except the default of the Commissioner) all or any part of the Holder's Payment or any other money payable by the Holder to the Commissioner is not paid on the due date the Holder will pay to the Commissioner interest at the Default Rate on the part of the Holder's Payment or any other money payable by the Holder to the Commissioner so unpaid from the due date until the date of actual payment in full.
- 19.2 The Commissioner's rights under this clause 19 are without prejudice to any other rights or remedies available to the Commissioner at law or in equity.

20 Goods and Services Tax

- 20.1 Unless the context otherwise requires, words and phrases used in this clause have the same meaning as in the GST Act.
- 20.2 If the supplies evidenced by the Holder's Consideration and the Commissioner's Consideration are taxable supplies under the GST Act, then:

- (a) the Commissioner and the Holder warrant to each other that they are registered for GST purposes as at the Holder's acceptance of this Proposal and that they will be so registered on the Settlement Date;
- (b) the Commissioner and the Holder confirm that as at the Settlement Date:
 - (i) each is acquiring the goods supplied with the intention of using the goods for making taxable supplies; and
 - (ii) the Commissioner and any associated person in terms of section 2A(1)(c) of the GST Act do not intend to use the Crown Land and the Holder and any associated person in terms of section 2A(1)(c) of the GST Act do not intend to use the Freehold Land as a principal place of residence; and
- (c) the Commissioner and the Holder agree that the supplies evidenced by the Holder's Consideration and the Commissioner's Consideration are to be zero-rated for GST purposes under section 11(1)(mb) of the GST Act.
- 20.3 If any of the circumstances set out in clause 20.2 change between the date of the Holder's acceptance of this Proposal and the Settlement Date, then the relevant party will notify the other of the changed circumstances as soon as practicable and in any event not later than 2 working days before the Settlement Date and such party shall warrant that the changed circumstances are correct as at the Settlement Date. If the GST treatment of the supplies evidenced by the Holder's Consideration and the Commissioner's Consideration changes as a result of the changed circumstances and a party has already provided the other with a GST invoice, then that party will issue a debit note or credit note, as the case may be, for GST purposes.
- 20.4 On the 10th working day following the Unconditional Date, the Commissioner will provide to the Holder a GST invoice in respect of the supply evidenced by the Holder's Consideration. The invoice will specify the Commissioner's GST Date.
- 20.5 The Holder will pay GST (if any) on the Holder's Consideration to the Commissioner by bank cheque on the Commissioner's GST Date, time being of the essence.
- 20.6 On the 10th working day following the Unconditional Date, the Holder will provide to the Commissioner a GST invoice in respect of the supply evidenced by the Commissioner's Consideration.
- 20.7 The Commissioner will pay GST (if any) on the Commissioner's Consideration to the Holder on the Commissioner's GST Date, time being of the essence.
- 20.8 Where any GST is not paid to the Commissioner or to the Holder (as the case may be) in accordance with this clause 20, the Holder will pay to the Commissioner, or the Commissioner will pay to the Holder (as the case may be), upon demand and together with the unpaid GST:
 - (a) interest, at the Default Rate, on the amount of the unpaid GST and which will accrue from the Commissioner's GST Date until the date of payment of the unpaid GST; and
 - (b) any Default GST.

21 Lowest price

- 21.1 The Holder's Consideration does not include any capitalised interest and the parties agree that the "lowest price" for the purposes of valuing the Freehold Land under section EW 32(3) of the Income Tax Act 2007 is equal to the Holder's Consideration.
- 21.2 The Commissioner's Consideration does not include any capitalised interest and the parties agree that the "lowest price" for the purposes of valuing the Crown Land under section EW 32(3) of the Income Tax Act 2007 is equal to the Commissioner's Consideration.

22 Costs

22.1 The Commissioner will meet the costs of the survey (if any) of the Land, including all designation areas, the Final Plan and for a certificate of title to issue for the Freehold Land.

22.2 The Holder is responsible for all costs the Holder incurs in respect of and incidental to the Tenure Review. In particular, but without limitation, the Holder shall bear all its costs in relation to the review of all documentation forming part of the Tenure Review (including this Proposal), and all professional advice provided to or sought by the Holder.

23 No nomination or assignment

23.1 The Holder is not entitled to, and is expressly prohibited from, nominating another person to perform the Holder's obligations under this Proposal or assigning to another person the Holder's interest (or any part) under this Proposal.

24 Recreation Permit

24.1 Immediately on the registration of the Final Plan and a copy of the proposal to which it relates over the Land and pursuant to s64 of the Act, any recreation permit granted over the Land shall be determined.

25 Consents for Activities

25.1 If the Holder has been granted a consent by the Commissioner to do an activity on the land under sections 15 or 16 of the Act, and the area over which the consent is exercised is designated in the proposal as Crown Land then the Holder agrees to act in good faith whilst exercising the terms of consent and not damage or destroy the Crown Land or anything thereon.

26 General

26.1 This Proposal and the Notice:

- (a) constitute the entire understanding and agreement between the Commissioner, the Crown and the Holder in relation to the Tenure Review; and
- (b) supersede and extinguish all prior agreements and understandings between the Crown, the Commissioner and the Holder relating to the Tenure Review.
- 26.2 Each provision of this Proposal will continue in full force and effect to the extent that it is not fully performed at the Settlement Date.
- 26.3 The Holder must comply with the Commissioner's requirements for the implementation and settlement of the Tenure Review contemplated by this Proposal.
- 26.4 The Commissioner and the Holder will sign and execute all deeds, agreements, schedules and other documents and do all acts and things as may be reasonably required by the other to effectively carry out and give effect to the terms and intentions of this Proposal.
- 26.5 This Proposal is governed by, and must be construed under, the laws of New Zealand and the Commissioner and the Holder irrevocably submit to the jurisdiction of the New Zealand courts or other New Zealand system of dispute resolution.
- 26.6 The illegality, invalidity or unenforceability of any provision in this Proposal will not affect the legality, validity or enforceability of any other provision.
- 26.7 In relation to notices and other communications under this Proposal:
 - (a) each notice or other communication is to be in writing, and sent by facsimile, personal delivery or by post to the addressee at the facsimile number or address, and marked for the attention of the person or office holder (if any), from time to time designated for that purpose by the addressee to the other party. Other than the address to which the Holder is to send its acceptance of this Proposal (which the Commissioner will specifically notify the Holder of) the address, person or office holder (if any) for each party is shown on the front page of this Proposal;
 - (b) no communication is to be effective until received. A communication will be deemed to be received by the addressee:

- in the case of a facsimile, on the working day on which it is despatched or, if despatched after 5.00 p.m. on a working day or, if despatched on a nonworking day, on the next working day after the date of dispatch;
- (ii) in the case of personal delivery (including, but not limited to, courier by a duly authorised agent of the person sending the communication), on the working day on which it is delivered, or if delivery is not made on a working day, on the next working day after the date of delivery; and
- (iii) in the case of a letter, on the fifth working day after mailing (postage paid).

27 Interpretation

27.1 Definitions

In this Proposal unless the context otherwise requires:

Act means the Crown Pastoral Land Act 1998;

Commissioner means the Commissioner of Crown Lands appointed under section 24AA of the Land Act 1948;

Commissioner's Consideration means the amount payable by the Commissioner to the Holder by equality of exchange for the surrender of the leasehold interest in the Lease in relation to the Crown Land, as specified in the Notice;

Commissioner's GST Date means the earlier of Settlement Date or the fifth working day before the day on which the Commissioner is due to pay to the Inland Revenue Department all GST payable by the Commissioner in respect of the supply made under this Proposal;

Commissioner's Payment means the balance of the Commissioner's Consideration payable by the Commissioner to the Holder by equality of exchange for the Crown Land, as specified in the Notice (if any);

Crown Land means the land (including any improvements) set out in Schedule One and the land (including any improvements) set out in Schedule Two (if any);

Default GST means any additional GST, penalty or other sum levied against either the Commissioner or the Holder under the Goods and Services Tax Act 1985 or the Tax Administration Act 1994 by reason of either the Commissioner or the Holder failing to pay GST as required by this Proposal. It does not include any sum levied against the Commissioner or the Holder by reason of a default by the Commissioner after payment of GST to the Commissioner by the Holder or by reason of a default by the Holder after payment of GST to the Holder by the Commissioner;

Default Rate means the rate of 11 per cent per annum;

Fencing means any stock proof farm fence.

Fencing Consent means any and all consents required for fencing under the Resource Management Act 1991.

Final Plan means the final plan for the Land prepared and submitted by the Commissioner to the Surveyor-General under sections 62(4)(c) and (d) of the Act;

Freehold Land means the land set out in Schedule Three;

GST means all goods and services tax payable by the Commissioner or the Holder under the Goods and Services Tax Act 1985 in respect of their respective supplies evidenced by this Proposal;

GST Act means the Goods and Services Tax Act 1985;

Holder means holder shown on the front page of this Proposal (being the lessee under the Lease);

Holder's Consideration means the amount payable by the Holder to the Commissioner by equality of exchange for the freehold of the Freehold Land, as specified in the Notice;

Holder's Payment means the balance of the Holder's Consideration payable by the Holder to the Commissioner by equality of exchange for the freehold of the Freehold Land, as specified in the Notice (if any);

Land means the land subject to the Tenure Review identified on the front page of this Proposal;

Lease means the lease described on the front page of this Proposal;

Mortgage means any mortgage (registered or unregistered) over the Land;

Mortgagee means the holder of any Mortgage;

Notice means the notice to the Holder setting out:

- (a) the Holder's Consideration;
- (b) the Commissioner's Consideration; and
- (c) the Holder's Payment or the Commissioner's Payment (as the case may be);

which includes amounts proposed to be paid by way of equality of exchange and accompanies this Proposal, but is not part of this Proposal;

Plan means the plan of the Land showing all designations on page 2 of this Proposal;

Registrar means the Registrar-General of Lands appointed pursuant to section 4 of the Land Transfer Act 1952;

Rent Review means the process for determination of the rent payable under the Lease as set out in sections 6 - 8 of the Act;

Settlement Date means the settlement date defined in clause 3.1;

Surveyor-General means the Surveyor-General appointed under section 5 of the Cadastral Survey Act 2002;

Tenure Review means the tenure review of the Land being undertaken by the Commissioner under the Act;

Unconditional Date means the date that the Commissioner receives from the Holder an executed copy of this Proposal signed by the Holder containing the signed consents of all persons having an interest in the Land to the Holder's acceptance of this Proposal which results in the acceptance taking effect under the Act;

Vesting Date means the date on which the Crown Land vests in the Crown pursuant to the Act;

Working day means a day that is not a Saturday, a Sunday, Waitangi Day, Good Friday, Easter Monday, Anzac Day, the Sovereign's birthday, Labour Day, or a day during the period commencing on any Christmas Day and ending with the 15th day of the following January or a day which is a provincial holiday in the place where the obligation is to be performed.

Works Consent means any and all consents required under the Resource Management Act 1991; and/or the Building Act 2004.

27.2 Construction of certain references

In this Proposal, unless inconsistent with the context:

- (a) a reference to a certificate of title includes a reference to a computer register;
- (b) words importing a gender include all genders;
- (c) reference to a statute includes reference to all enactments that amend or are passed in substitution for the relevant statute and to all regulations relating to that statute;
- (d) words in the singular include the plural and vice versa;

- (e) reference to a month means a calendar month;
- (f) reference to a person means an individual, a body corporate, an association of persons (whether corporate or not), a trust or a state or agency of a state (in each case, whether or not having separate legal personality);
- (g) references to sections, clauses, sub-clauses, parts, annexures, attachments, appendices, schedules, paragraphs and sub-paragraphs are references to such as they appear in this Proposal and form part of this Proposal;
- (h) headings are included for ease of reference only and will not affect the construction or interpretation of this Proposal;
- (i) all monetary amounts are expressed in New Zealand currency;
- (j) references to obligations includes reference to covenants, undertakings, warranties and, generally, obligations or liabilities of any nature properly arising whether directly or indirectly, under or in respect of the relevant contract, agreement or arrangement;
- (k) all references to times are references to times in New Zealand;
- (I) if the Holder comprises more than one person, each of those person's obligations, as Holder, will be both joint and several.

Schedule One: Provisions relating to the Schedule One Land

1 Details of Designation

- 1.1 Under this Proposal the land shaded pink on the Plan and labelled "RR", being 27 hectares (approximately) is designated as land to be restored to or retained in full Crown ownership and control as reserve held for the purpose of recreation.
- 1.2 Under this Proposal the land shaded pink on the Plan and labelled "CA2", being 172 hectares (approximately) is designated as land to be restored to or retained in full Crown ownership and control as conservation area.

2 Schedule One Improvements

Nil

Schedule Two: Provisions relating to the Schedule Two Land

1 Details of designation

- 1.1 Under this Proposal the land shaded pink on the Plan and labelled "CA1", being 10,629 hectares (approximately) is designated as land to be restored to or retained in Crown control as conservation area subject to:
 - (a) The granting of a guiding concession for guided walking, tramping, rock climbing, mountaineering and ski touring, and guided commercial ground based hunting, as set out in Appendix 4.
 - (b) The granting of a concession for irregular helicopter landings for the purpose of scenic landing, heli-landings for the purpose of positioning recreationists and commercially guided groups, and heli-landings for the purpose of drop off and/or pick up of heli-hiking clients, as set on in Appendix 5.
 - (c) The granting of a concession for helicopter landings for the purpose of heli snowboarding and heli-skiing, and guided skiing and guided snowboarding, as set out in Appendix 6.
 - (d) The granting of a concession for commercial filming and photography, as set out in Appendix 7.
 - (e) The granting of a concession to operate, maintain and repair a telecommunications facility as set out in Appendix 8;

2 Information Concerning Proposed Concessions

2.1 Guiding concession

2.1.1 Description of the proposed activity:

The guiding licence provides for the following activities:

- (a) Guided walking, tramping, climbing, rock climbing, mountaineering and ski touring
- (b) Guided commercial ground based hunting
- 2.1.2 Description of area where proposed activity to be carried out and proposed status:

The Concession Activity is over the area designated as Conservation Area CA1 on the Hall Range on the plan attached to the proposal. The area covers the Hall Range adjacent to Lake Tekapo and the Godley River from Godley Pass at the northern end down to and including Mistake Peak and a small area of the southern end of the Haszard Range.

2.1.3 Description of potential effects of proposed activity and any actions proposed to avoid, mitigate or remedy any adverse effect:

DOC staff will periodically visit the concession area of the proposed conservation land for a range of activities including surveillance and control of wild animals and weeds and will therefore be aware if the concession activity is having any negative effects.

2.1.4 Details of the proposed type of concession:

The proposed guiding licence will be issued under Section 17Q(1) of the Conservation Act 1987.

- 2.1.5 Proposed duration of concession and reason for proposed duration:
 - (a) Proposed duration 10 years
 - (b) Reason for proposed duration To provide commercial certainty for the proposed activities.
- 2.1.6 Relevant information about the proposed grantee including information relevant to the Concessionaire's ability to carry out the proposed activity:

Proposed grantee: Verity Farms Limited.

Relevant information: The proposed concessionaire currently already operates on the area, which is part of Godley Peaks Pastoral Lease. The licence was agreed to as part of the tenure review negotiations.

2.2 Concession for irregular helicopter landings

2.2.1 Description of the proposed activity:

The licence provides for irregular helicopter landings for the purpose of:

- (a) Scenic landings
- (b) Heli-landings for the purpose of positioning recreationists (including ground-based hunters), and commercially guided groups other than for helicopter-snow activities.
- (c) Heli-landings for the purpose of drop off and/or pick up of heli-hiking clients
- 2.2.2 Description of area where proposed activity to be carried out and proposed status:

The Concession Activity is over the area designated as Conservation Area CA1 on the Hall Range on the plan attached to the proposal. The area covers the Hall Range adjacent to Lake Tekapo and the Godley River from Godley Pass at the northern end down to and including Mistake Peak and a small area of the southern end of the Haszard Range.

2.2.3 Description of potential effects of proposed activity and any actions proposed to avoid, mitigate or remedy any adverse effect:

DOC staff will periodically visit the concession area of the proposed conservation land for a range of activities including surveillance and control of wild animals and weeds and will therefore be aware if the concession activity is having any negative effects.

2.2.4 Details of the proposed type of concession:

The proposed guiding licence will be issued under Section 17Q(1) of the Conservation Act 1987.

- 2.2.5 Proposed duration of concession and reason for proposed duration:
 - (a) Proposed duration 10 years
 - (b) Reason for proposed duration To provide commercial certainty for the proposed activities.
- 2.2.6 Relevant information about the proposed grantee including information relevant to the Concessionaire's ability to carry out the proposed activity:

Proposed grantee: Verity Farms Limited.

Relevant information: The proposed concessionaire currently already operates on the area, which is part of Godley Peaks Pastoral Lease. The licence was agreed to as part of the tenure review negotiations.

2.3 Concession for heli skiing and snowboarding

2.3.1 Description of the proposed activity:

The licence provides for helicopter landings for the purpose of heli snowboarding and heli skiing, and also for guided skiing and guided snowboarding

2.3.2 Description of area where proposed activity to be carried out and proposed status:

The Concession Activity is over the area designated as Conservation Area CA1 on the Hall Range on the plan attached to the proposal. The area covers the Hall Range adjacent to Lake Tekapo and the Godley River from Godley Pass at the northern end down to and including Mistake Peak and a small area of the southern end of the Haszard Range.

2.3.3 Description of potential effects of proposed activity and any actions proposed to avoid, mitigate or remedy any adverse effect:

DOC staff will periodically visit the concession area of the proposed conservation land for a range of activities including surveillance and control of wild animals and weeds and will therefore be aware if the concession activity is having any negative effects.

2.3.4 Details of the proposed type of concession:

The proposed guiding licence will be issued under Section 17Q(1) of the Conservation Act 1987.

- 2.3.5 Proposed duration of concession and reason for proposed duration:
 - (a) Proposed duration 10 years
 - (b) Reason for proposed duration To provide commercial certainty for the activities.
- 2.3.6 Relevant information about the proposed grantee including information relevant to the Concessionaire's ability to carry out the proposed activity:

Proposed grantee: Verity Farms Limited.

Relevant information: The proposed concessionaire currently already operates on the area, which is part of Godley Peaks Pastoral Lease. The licence was agreed to as part of the tenure review negotiations.

2.4 Concession for commercial filming and photography

2.4.1 Description of the proposed activity:

The general licence provides for the use of the land for the purposes of commercial filming and photography.

2.4.2 Description of area where proposed activity to be carried out and proposed status:

The Concession Activity is over the area designated as Conservation Area CA1 on the Hall Range on the plan attached to the proposal. The area covers the Hall Range adjacent to Lake Tekapo and the Godley River from Godley Pass at the northern end down to and including Mistake Peak and a small area of the southern end of the Haszard Range.

2.4.3 Description of potential effects of proposed activity and any actions proposed to avoid, mitigate or remedy any adverse effect:

DOC staff will periodically visit the concession area of the proposed conservation land for a range of activities including surveillance and control of wild animals and weeds and will therefore be aware if the concession activity is having any negative effects.

2.4.4 Details of the proposed type of concession:

The proposed guiding licence will be issued under Section 17Q(1) of the Conservation Act 1987.

- 2.4.5 Proposed duration of concession and reason for proposed duration:
 - (a) Proposed duration 10 years
 - (b) Reason for proposed duration To provide commercial certainty for the activities.
- 2.4.6 Relevant information about the proposed grantee including information relevant to the Concessionaire's ability to carry out the proposed activity:

Proposed grantee: Verity Farms Limited.

Relevant information: The proposed concessionaire currently already operates on the area, which is part of Godley Peaks Pastoral Lease. The licence was agreed to as part of the tenure review negotiations.

2.4 Concession for telecommunications facility:

2.4.1 Description of the proposed activity:

To establish, operate, maintain and repair a telecommunications facility for the provision of communications services and the right of helicopter access to the site labelled as "TA" on the Plan attached to the Proposal.

The site will provide for the establishment and continued operation of a new communications facility to provide for essential farm communications and health and safety management for the local area.

2.4.2 Description of area where proposed activity is to be carried out and proposed status:

The site is located on the top of Mistake Peak at the southern end of the Hall Range. This site is included in the proposed Conservation Area extending along the Hall Range and will be at the location marked "TA" on the proposal plan.

2.4.3 Description of potential affects of proposed activity and any actions proposed to avoid, mitigate or remedy any adverse affect:

DOC staff will periodically visit the concession area of the proposed conservation land for a range of activities including surveillance and control of wild animals and weeds and will therefore be aware if the concession activity is having any negative effects.

2.4.4 Details of the proposed type of concession:

A Telecommunications Aerial Concession under S. 17Q (1) Conservation Act 1987.

- 2.4.5 Proposed duration of concession and reason for proposed duration:
 - (a) Proposed duration 30 years
 - (b) Reason for proposed duration To provide commercial certainty for telecommunications operations.
- 2.4.6 Relevant information:

Proposed grantee: Verity Farms Limited.

Relevant information: The licence was agreed to as part of the tenure review negotiations.

Schedule Three: Provisions relating to the Schedule Three Land

1 Details of designation

- 1.1 Under this Proposal the land shaded green or yellow on the Plan, being 3,665.0582 hectares (approximately) is designated as land to be disposed of by freehold disposal to the Holder subject to:
 - (a) Part IVA of the Conservation Act 1987;
 - (b) Section 11 of the Crown Minerals Act 1991;
 - (c) the easement marked as "a-b, c-d, e-f, g-h. g-i, j-k, I-m, b-n, o-p, and q-r" on the Plan and substantially as set out in Appendix 9; and
 - (d) the covenant shaded yellow on the Plan and labelled "CC1", and substantially as set out in Appendix 10;

Schedule Four: Conditions

- 1 The Commissioner is under no obligation, and may decide, in its sole discretion, not to proceed further with the Tenure Review unless and until:
 - (a) the Commissioner considers that sufficient funds will be obtained in order to complete the Tenure Review;
 - (b) the Director General of Conservation has completed all actions required under Part IVA of the Conservation Act 1987;
 - (c) the Commissioner has reviewed, and is satisfied, in its sole discretion that the easement referred to in Appendix 9 is in an acceptable form, has been executed, consented by the necessary parties and has been registered against the Lease.
 - (d) the Commissioner has reviewed, and is satisfied, in its sole discretion that the covenant referred to in Appendix 10 is in an acceptable form, has been executed, consented by the necessary parties and has been registered against the Lease.

Appendix 1: Consents – Example of Mortgagee Consent

[] as Mortgagee under Mortgage [

] ("the Mortgage"), hereby:

- (a) consents to acceptance of the Proposal dated [] ("the Proposal") by [the Holder] ("the Holder") pursuant to the Crown Pastoral Land Act 1998 and agrees and consents to the registration of the documents affecting the Freehold Land referenced in the Proposal prior to the registration of any new mortgage to be granted in its favour over the Freehold Land; and
- (b) agrees to sign and execute all deeds, agreements, schedules and other documents and do all acts and things as may be reasonably required by the Holder or the Commissioner to register a discharge of the Mortgage and any new mortgage over the Freehold Land.

Dated:

SIGNED by [])	
in the presence of:)	

Witness Signature:

Witness Name:
Occupation:
Address:

Appendix 1: Consents (continued) - Example of "Other" Consent

[], being the party entitled to the benefit of [] regi	stered
against Lease [], hereby consents to the acceptance of the Proposa	al dated [] by
[the Holder] pursuar	nt to the Crown Pastoral Land Act 1998.		

Dated:

SIGNED for and on behalf of [] in the presence of:)))	
Witness Signature:		

Witness Name: Occupation: Address:

Appendix 2: Example of Solicitors Certificate

Certifications

- I [] hereby certify as follows:
- [[insert name of Holder] ("the Holder") is a duly incorporated company under the Companies Act 1993. The Holder's entry into and performance of its obligations under the Proposal dated [] ("the Proposal") have been duly authorised by the directors and, if required, by the shareholders of the Holder. The Holder has executed the Proposal in accordance with its constitution.] OR

The entry into the Proposal dated [] ("the Proposal") by [insert name of Holder] ("the Holder") and performance of the Holder's obligations under the Proposal have been duly authorised by the trustees of the [insert name of trust] in accordance with its trust deed **OR**

[[insert name of Holder] ("the Holder") has delegated responsibility for signing the Proposal on its behalf to an attorney in accordance with its constitution. The attorney of the Holder has properly executed the Proposal in accordance with this power of attorney and in accordance with the Holder's constitution and a certificate of non-revocation is enclosed.]

- 2. The consent of each person that has an interest (registered or unregistered) in the Land (as that term is defined in the Proposal), to the Holder's acceptance of the Proposal has been obtained and included in the copy of the Proposal, signed by the Holder, that has been provided to the Commissioner.
- 3. [No consent, licence, approval or authorisation by any court, regulatory authority or governmental agency is required to enable the Holder to accept the Proposal, perform the Holder's obligations under the Proposal and to acquire the freehold interest in the Land (as defined in the Proposal).] *OR*

[All necessary consents, licences, approvals and authorisations required to enable the Holder to accept the Proposal, perform its obligations under it and to acquire the freehold interest in the Land (as defined in the Proposal) have been obtained. Evidence of the consents, licences, approvals and authorisations are attached to this letter.]

Yours faithfully [signed by principal of law firm]

Appendix 3: Indicative Fencing and Construction Requirements

Length and location:	X-Y	3400m (approximately)
	R-S	3400m (approximately)
	T-U	1400m (approximately)
	V-W	1400m (approximately)

Type: Fence to be a standard 7 wire sheep/cattle fence

Specifications:

- Fence to be constructed of 7 x 2.5mm h/t wires.
- 1 x 3.6m Heavy duty gate in each line midway at nominated waypoint. Gate to be swung to open fully and close onto opposite strainer, secured with a full wrap around type chain.
- 2.1m x 175mm treated timber strainers for end of strains and intermediate corners, stayed or tied back. All strainers either footed or tied down.
- 2.4m x 125mm stay posts
- All tiebacks are to have a minimum of 2 anchor points and all tiedown and tieback wire to be 4.00mm soft wire remaining clear of ground.
- Tiebacks permitted on both sides of fence.
- All wires strained to manufacturers specifications, tied off neatly and securely and remain between 100 and 150mm clear of ground.
- Permanent wire strainers on all strains and a max strain length of 300m.
- 1.50m y-post at 5m max spacing. 2.00kg/m min weight.
- 1.8m x 125mm treated timber posts on high points and at max spacing of 21m
- 1.65m T-Irons acceptable as intermediate strainers and intermediate posts if ground conditions do not allow for timber posts to be used.
- 50 x 4 mm barbed staples driven well in but allow wire to run through, with insulators for one wire.

The Construction Works

Certain sections of fence may require mechanical clearance of rocks or other debris. Such clearance will be minimized and will require an undertaking of:

- Minimal vegetation disturbance
- Not to cause slope instability
- Not to cause erosion or siltation

Works Consents from ECAN will be obtained where required.

If in the course of fencing work it is considered that additional mechanical clearance or earthworks are required beyond work specified in the final fencing specifications, then a separate consent from LINZ

will be required prior to any work being undertaken. Such consent is to be sought by LINZ's implementation contractor and approval will require an undertaking in relation to the above matters, with Works Consents obtained from ECAN as appropriate.

Appendix 4: Guiding concession (DOCDM-1599933 (1))



Department of Conservation Te Papa Atawhai

Concession Number:

Concession Document (Guiding Licence)

THIS CONCESSION is made this day of

PARTIES:

- **1. Minister of Conservation** (the Grantor)
- 2. Verity Farms Limited (the Concessionaire)

BACKGROUND

- **A.** The Department of Conservation ("Department") Te Papa Atawhai is responsible for managing and promoting conservation of the natural and historic heritage of New Zealand on behalf of, and for the benefit of, present and future New Zealanders, and is under the control of the Grantor.
- **B.** The land described in Item 1 of Schedule 1 as the Land is a Conservation Area or a Reserve under the management of the Grantor.
- **C.** Sections 66 and 68 of the Crown Pastoral Land Act 1998 authorise the Grantor to grant a Concession for a Concession Activity in a Conservation Area and a Reserve under section 17Q(1) of the Conservation Act 1987 or section 59A of the Reserves Act 1977 (whichever is relevant in the circumstances).
- **D.** The Concessionaire wishes to carry out the Concession Activity on the Land subject to the terms and conditions of this Concession.
- **E.** The parties wish to record the terms and conditions of this Concession and its Schedules.

OPERATIVE PARTS

1. In exercise of the Grantor's powers under the Crown Pastoral Land Act 1998, and the Conservation Act 1987 or the Reserves Act 1977 as relevant, the Grantor **GRANTS** to the Concessionaire a **GUIDING LICENCE** to carry out the Concession Activity on the Land subject to the terms and conditions contained in this Concession and its Schedules.

its

SCHEDULE 1

1.	Land (Schedule 3)	As shown on the Proposed Designations Plan marked (CA1) attached in Schedule 4
		Conservation Area of 10,629 hectares approximately on the Hall Range and Mistake Peak situated in the Land District of Canterbury
2.	Concession Activity (clause 2)	a) Guided walking, tramping, climbing, rock climbing, mountaineering and ski touring.b) Guided commercial ground based hunting.
3.	Term (clause 3)	10 years commencing on the date that an approved plan is registered, affecting Certificate of Title CB 30B/914 (Canterbury Registry), vesting the land in the Crown as a conservation area or reserve, being the day of 20 (the commencement date)
4.	Renewal(s) (clause 3)	None
5.	Final Expiry Date (clause 3)	The 10th anniversary of the commencement date.
6.	Concession Fee	Annual Activity Fees:
0.	(clause 4)	a) Guiding walking, tramping, climbing, rock climbing, mountaineering and ski touring
		\$10.00 plus GST per adult client guided per full day (defined as a period of more than 4 hours but no more than 24 hours);
		\$5.00 plus GST per adult client guided per half day (a half day being defined as more than 1 hour but no more than 4 hours); and
		\$1.00 plus GST per adult client guided per hour or less.
		For any client charged by the Concessionaire at a reduced rate as a 'child', the annual activity fee is to be half the adult client concession activity fee.
		b) Guided commercial ground based hunting
		\$15.00 plus GST per person guided
		\$5.00 for accompanying companion per day Animal fees
		\$150.00 plus GST per Tahr (males only)
		\$100.00 plus GST per Chamois
		\$1.00 plus GST per pig
		\$1.00 plus GST per deer
		Annual Management Fee:
		\$500 per annum plus GST
		Annual Environmental Monitoring Fee
		To be negotiated between Grantor and Concessionaire if required (see special condition 30)

Concessionaire's initials

Grantor's initials

7.		a. Annually, in arrears upon commencement of the term set out in clause 3 for payment instalments		
	Instalments (clauses 4 and 6.1)	b. Annually, in arrears upon commencement of the term set out in clause 3 for activity returns, using the form attached in Schedule 5		
8.		On or before the date specified on the invoice issued by the Grantor.		
9.	· · · · ·	Double the current Official Cash Rate (OCR). See Reserve Bank of New Zealand website		
10.	ConcessionFeeReview Date(s)(clause 5)	On the third anniversary of the date of commencement of this Concession set out in item 3 above and the corresponding date every 3 years thereafter until the expiry of the term.		
11.	Health and Safety (clause 12)	Audited Safety Plan: Required (unless Qualmark or Outdoorsmark certified) Auditors certificate of approval to be provided to Grantor		
12.	Concessionaire Identification (Clause 23)	Required		
13.	Insurance (To be obtained by Concessionaire) (clause 11)	 Types and amounts: Public Liability Insurance for (a) General indemnity for an amount no less than \$2,000,000.00; and (b) Forest and Rural Fires Act extension for an amount no less than \$2,000,000.00; and (c) Third party vehicle liability for an amount no less than \$500,000.00 (d) Statutory liability for an amount no less than \$500,000.00 		
14.	Addresses for Notices (clause 21)	Subject to review on each Concession Fee Review Date The Grantor's address is: Physical Address: Department of Conservation Conservation House 77 Lower Stuart Street Dunedin 9016		
		Postal Address: PO Box 5244 Moray Place Dunedin 9058 Phone: (03) 477 0677 Email: <u>permissionsdunedin@doc.govt.nz</u>		

Concessionaire's initials

Grantor's initials

		The Concessionaire's address in New Zealand is:	
		Address:	
		C/- Milnes Beatson Ld	
		29 Wallace Street	
		Motueka 7120	
		Phone: (03) 528 8760	
		Email: david@mba.net.nz	
		Director's Address:	
		210 Patons Rock Road	
		RD2	
		Takaka 7182	
		Phone: 021 531 227	
		Email: ann@verityfarms.co.nz	
15.	Special Conditions	See Schedule 3.	
	(clause 25)		

Note: The clause references are to the Minister of Conservation's Standard Terms and Conditions for Guiding Licences set out in Schedule 2.

Note: Please initial each page of Schedule 1

SCHEDULE 2

STANDARD TERMS AND CONDITIONS FOR GUIDING LICENCES

1. Interpretation

1.1 "Background" means the matters referred to under the heading 'Background" on the first page of this Document, and words used in the Background have the meaning given to them in this clause 1.1.

"Commencement date" means the date that an approved plan is registered vesting the Land in the Crown as a conservation area or reserve.

"Concession" means this document and any subsequent amendments and all schedules, annexures, and plans attached to it.

"Concession Activity" means the activity described in item 2 of Schedule 1.

"Conservation Area" has the same meaning as "conservation area" in section 2 of the Conservation Act 1987.

"Director-General" means the Director-General of Conservation

"**Reserve**" has the same meaning as "reserve" in section 59A of the Reserves Act 1977.

"Term" means the period of time specified in Item 5 of Schedule 1 during which this Concession operates.

1.2 The Concessionaire is responsible for the acts and omissions of its employees, contractors, agents, clients and invitees (excluding other members of the public accessing the Land). The Concessionaire is liable under this Concession for any breach of the terms of the Concession by its employees, contractors, agents, clients and invitees (excluding other members of the public accessing the Land), as if the breach had been committed by the Concessionaire.

2. What is being authorised?

2.1 The Concessionaire is only allowed to use the Land for the Concession Activity.

3. How long is the Concession for - the Term?

- 3.1 This Concession commences on the date set out in Item 3 of Schedule 1 and ends on the Final Expiry Date specified in Item 5 of Schedule 1.
- 3.2 No renewals of this Concession are permitted.

4. What are the fees and when are they to be paid?

4.1 The Concessionaire must pay to the Grantor in the manner directed by the Grantor the Concession Fee (which includes the Annual Activity Fees, the Management Fee, and the Environmental Monitoring Fee) plus GST in the

instalments and on the Concession Fee Payment Dates specified in Items 6, 7 and 8 of Schedule 1.

4.2 If the Concessionaire fails to make payment within 14 days of the Concession Fee Payment Date then the Concessionaire is to pay interest on the unpaid Concession Fee from the Concession Fee Payment Date until the date of payment at the Penalty Interest Rate specified in Item 9 of Schedule 1.

5. When can the fees be reviewed?

- 5.1 The Grantor is to review the Concession Fee on the Concession Fee Review Date stated in Item 10 of Schedule 1. The new Concession Fee is to be the market value of the Concession Activity carried out on the Land having regard to the matters set out in section 17Y(2) of the Conservation Act 1987.
- 5.2 Both parties are to agree on the new fee within 30 working days of the Grantor giving the Concessionaire written notice of the review.
- 5.3 If the parties cannot so agree then each party is to appoint a Registered Valuer who must meet and agree on the new fee. If the Registered Valuers fail to reach agreement the new fee is to be determined by an umpire appointed by the two Registered Valuers. Each party is to bear that party's own costs and half the costs of the umpire (if any).

6. What about Activity return forms?

6.1 The Concessionaire must complete a Client Activity Return form in the format required by the Grantor, and return them to the Grantor on the Activity Return Dates stated in Item 7 of Schedule 1. The Grantor may request further or different activity related information to better monitor and determine any effects of the Concession Activity on the Land.

7. When can the Concession be assigned?

- 7.1 The Concessionaire must not transfer, sub licence, assign, mortgage or otherwise dispose of the Concessionaire's interest under this Concession or any part of it (which includes the Concessionaire entering into a contract or any other arrangement whatsoever whereby the Concession Activity would be carried out by a person (called the assignee) other than the Concessionaire) without the prior written consent of the Grantor.
- 7.2 The Grantor may in the Grantor's discretion decline any application for consent under clause 7.1.
- 7.3 Sections 17P, 17S, 17T, 17U, 17W, 17X, 17ZB and 17ZC of the Conservation Act 1987 apply to applications for consent under this clause unless the Grantor, in the Grantor's discretion, decides otherwise.
- 7.4 If the Grantor gives consent under this clause then the Concessionaire remains liable to observe and perform the terms and conditions of this Concession throughout the Term and is to procure from the Assignee a covenant to be bound by the terms and conditions of this Concession.
- 7.5 The Concessionaire must pay the costs reasonably incurred by the Grantor incidental to any application for consent, whether or not such consent is granted.

7.6 Any change in the shareholding of the Concessionaire altering the effective control of the Concessionaire requires the approval of the Grantor.

8. What are the obligations to protect the environment?

- 8.1 The Concessionaire must not cut down or damage any vegetation; or damage any natural feature or historic resource on the Land; or light any fire on the Land without the prior consent of the Grantor.
- 8.2 Having regard to the Concession Activity, the Concessionaire must ensure that it adheres to the international "Leave No Trace" Principles (www.leavenotrace.org.nz).
- 8.3 The Concessionaire must not bury:
 - (a) any toilet waste within 50 metres of a water source on the Land; or
 - (b) any animal or fish or any part thereof within 50 metres of any water body, water source or public road or track.

9. When can structures be erected?

- 9.1 The Concessionaire must not place any structures on the Land without the prior written consent of the Grantor.
- 9.2 The Concessionaire must keep all structures, buildings, fences, gates, drains and other improvements now or hereafter upon the Land, in good order, condition and repair.

10. What if the Concessionaire wishes to surrender the Concession?

10.1 If the Concessionaire wishes to surrender this Concession during the currency of the Term, then the Grantor may accept that surrender on such conditions as the Grantor considers appropriate.

11. What are the liabilities and who insures?

- 11.1 The Concessionaire agrees to use the Land at the Concessionaire's own risk and releases to the full extent permitted by law the Grantor and the Grantor's employees and agents from all claims and demands of any kind and from all liability which may arise in respect of any accident, damage or injury occurring to any person or property in or about the Land.
- 11.2 The Concessionaire must indemnify the Grantor against all claims, actions, losses and expenses of any nature which the Grantor may suffer or incur or for which the Grantor may become liable arising from the Concessionaire's performance of the Concession Activity.
- 11.3 This indemnity is to continue after the expiry or termination of this Concession in respect of any acts or omissions occurring or arising before its expiry or termination.
- 11.4 Without prejudice to or in any way limiting its liability under this clause 11 the Concessionaire at the Concessionaire's expense must take out and keep current policies for insurance and for the amounts not less than the sums stated in Item 13 of Schedule 1 with a substantial and reputable insurer.
- 11.5 The Grantor may on each Concession Fee Review Date on giving 10 working day's notice to the Concessionaire alter the amounts of insurance required under clause 11.4. On receiving such notice the Concessionaire must within 10 working days take out and keep current policies for insurance and for the amounts not less than the sums specified in that notice.
- 11.6 The Concessionaire must, provide to the Grantor within 5 working days of the Grantor so requesting:
 - (a) details of any insurance policies required to be obtained under this Concession, including any renewal policies if such renewal occurs during the Term; and/ or;
 - (b) a copy of the current certificate of such policies.

12. What about Health and Safety?

- 12.1 The Concessionaire must exercise the rights granted by this Concession in a safe and reliable manner and must comply with the Health and Safety in Employment Act 1992 and its regulations and all other provisions or requirements of any competent authority relating to the exercise of this Concession. The Concessionaire must comply with its safety plan (if one is required in Item 11 of Schedule 1), and with any safety directions of the Grantor.
- 12.2 Before commencing the Concession Activity the Concessionaire must where the Concessionaire has Qualmark or Outdoorsmark certification provide the Grantor with a copy of that certification.
- 12.3 If the Concessionaire does not hold Qualmark or Outdoorsmark certification then before commencing the Concession Activity the Concessionaire must, if required by Item 11 of Schedule 1:
 - (a) prepare a safety plan;

- (b) have it audited by a suitably qualified person approved by the Grantor; and forward to the Grantor a certificate from the auditor certifying that the safety plan is suitable for the Concession Activity; and
- (c) The Concessionaire must obtain from the auditor details as to when the safety plan is to be re-audited. The Concessionaire must comply with any such requirement to re-audit and forward a copy of the re-audit certificate to the Grantor within 5 working days of the certificate being issued.
- 12.4 The Grantor may at any time request the Concessionaire to provide the Grantor with a copy of the current safety plan in which case the Concessionaire must provide the copy within 10 working days of receiving the request.
- 12.5 The Concessionaire must:
 - 12.5.1 take all reasonable steps to protect the safety of all persons present on the Land in the vicinity of the Concession Activity; and
 - 12.5.2 take all reasonable steps to eliminate any dangers to the public of which the Concessionaire is aware; and
 - 12.5.3 record and report to the Grantor all accidents or incidents involving serious harm within 24 hours of their occurrence and forward an investigation report to the Grantor within 3 working days of the accident or incident occurring;
 - 12.5.4 at the request of the Grantor make available any of the Concessionaire's directors, employees, servants or agents who in the opinion of the Grantor might assist any investigation by the Grantor into the cause of any such serious harm accident or incident.
- 12.6 The Concessionaire is requested to notify the Grantor as soon as practicable of any natural events or activities on the Land or the surrounding area which may endanger the public or the environment.

13. What are the compliance obligations of the Concessionaire?

- 13.1 The Concessionaire must comply where relevant:
 - (a) with the provisions of any conservation management strategy or conservation management plan under the Conservation Act 1987 or Part 2A of the Reserves Act 1977, or any general policy statement made under the Conservation Act 1987, Reserves Act 1977, National Parks Act 1980, or Wildlife Act 1953, or management plan under section 45 of the National Parks Act 1980, whichever is appropriate to the Land, together with any amendment or review of any policy, strategy or plan whether approved before, on, or after the date on which this Concession takes effect; and
 - (b) with the Conservation Act 1987, the Reserves Act 1977, the National Parks Act 1980, Wildlife Act 1953 and any other statute, ordinance, regulation, bylaw, or other enactment (collectively the "Legislation") affecting or relating to the Land or affecting or relating to the Concession Activity, including any regulations made under the Conservation Act 1987 and Wildlife Act 1953 or bylaws made under the Reserves Act 1977 or the National Parks Act 1980; and

- (c) with all notices and requisitions of any competent authority affecting or relating to the land or affecting or relating to the conduct of the Concession Activity; and
- (d) with all Department signs and notices placed on or affecting the Land
- 13.2 The Concessionaire must comply with this Concession.
- 13.3 A breach or contravention by the Concessionaire of a relevant conservation management strategy, conservation management plan, management plan or any statement of general policy referred to in clause 13.1 (a) is deemed to be a breach of this Concession.
- 13.4 A breach or contravention by the Concessionaire of any Legislation affecting or relating to the Land or affecting or relating to the Concession Activity is deemed to be a breach of this Concession.

14. When can the Concession be suspended?

- 14.1 If, in the Grantor's opinion, there is a temporary risk to any natural or historic resource on or in the vicinity of the Land or to public safety whether arising from natural events such as earthquake, land slip, volcanic activity, flood, or arising in any other way, whether or not from any breach of the terms of this Concession on the part of the Concessionaire, then the Grantor may suspend this Concession.
- 14.2 If, in the Grantor's opinion, the activities of the Concessionaire is having or may have an adverse effect on the natural, historic or cultural values or resources of the Land and the Grantor considers that the effect can be avoided, remedied or mitigated to an extent satisfactory to the Grantor, then the Grantor may suspend this Concession until the Concessionaire avoids, remedies or mitigates the adverse effect to the Grantor's satisfaction.
- 14.3 The Grantor may suspend the Concession for such period as the Grantor determines where the Concessionaire has breached any terms of this Concession.
- 14.4 The Grantor may suspend this Concession while the Grantor investigates any of the circumstances contemplated in clauses 14.1 and 14.2 and also while the Grantor investigates any potential breach or possible offence by the Concessionaire, whether or not related to the Concession Activity under the Conservation Act 1987 or any of the Acts mentioned in the First Schedule of that Act.
- 14.5 The word "investigates" in clause 14.4 includes the laying of charges and awaiting the decision of the Court.
- 14.6 During any period of temporary suspension arising under clauses 14.1 or 14.2 the Concession Fee payable by the Concessionaire is to abate in fair proportion to the loss of use by the Concessionaire of the Land.
- 14.7 The Grantor is not to be liable to the Concessionaire for any loss sustained by the Concessionaire by reason of the suspension of the Concession under this clause 14 including loss of profits.

15. When can the Concession be terminated?

- 15.1 If:
 - (a) the Concessionaire breaches any of the conditions of this Concession; or
 - (b) the whole or any part of the Land is required for the Grantor's use

the Grantor may terminate this Concession at any time in respect of the whole or any part of the Land. Before so terminating the Grantor must give the Concessionaire either:

- (c) one calendar month's notice in writing; or
- (d) such other time period which in the sole opinion of the Grantor appears reasonable and necessary

of the Grantor's intention so to terminate this Concession. If this Concession is terminated then the Grantor, at the Grantor's sole discretion, may adjust the Concession Fee payable or refund any Concession Fee paid in advance.

16. What are the Grantor's Rights to remedy defaults?

16.1 The Grantor may choose to remedy at any time without notice any default by the Concessionaire under this Concession. Where that occurs, the Concessionaire must pay forthwith on demand all reasonable costs incurred by the Grantor in remedying such default.

17. What happens on termination or expiry of the Concession?

17.1 Upon the expiry or earlier termination of this Concession, either as in whole or in part, the Concessionaire is not entitled to compensation for any structure or other improvement erected or carried out by the Concessionaire. The Concessionaire must within such time as the Grantor determines, remove all such structures or other improvements making good at the Concessionaire's expense any damage caused by such removal and leaving the Land in a clean and tidy condition.

18. When is the Grantor's consent required?

18.1 Where the Grantor's consent or approval is expressly required under this Concession then the Concessionaire must seek that approval or consent for each separate time it is required even though the Grantor may have given approval or consent for a like purpose on a prior occasion. Any such consent or approval may be made on such conditions as the Grantor considers appropriate.

19. Are there limitations on public access and closure?

19.1 The Concessionaire acknowledges that the Land is open to the public for access and that the Grantor may close public access during periods of high fire hazard or for reasons of public safety or emergency.

20. How will disputes be resolved?

20.1 If a dispute arises between the parties in connection with this Concession the parties must, without prejudice to any other rights or entitlements they may have, attempt to resolve the dispute by agreement using informal dispute resolution techniques such as negotiation, mediation, independent expert

appraisal or any other alternative dispute resolution technique. The rules governing any such technique adopted are to be agreed between the parties.

- 20.2 If the dispute cannot be resolved by agreement within 14 days of written notice by one party to the other (or such further period as the parties may agree to in writing) either party may refer the dispute to the Disputes Tribunal, where relevant, or to arbitration which arbitration is to be carried out in accordance with the provisions of the Arbitration Act 1996.
- 20.3 If the parties do not agree on an arbitrator within 10 working days of a party giving written notice of the requirement to appoint an arbitrator the President of the New Zealand Law Society is to appoint the arbitrator. In either case the arbitrator must not be a person who has participated in an informal dispute resolution procedure in respect of the dispute.
- 20.4 The arbitrator must include in the arbitration award reasons for the determination.
- 20.5 Despite the existence of a dispute, each party must continue to perform its obligations under this Concession.

21. How are notices sent and when are they received?

- 21.1 Any notice to be given under this Concession is to be in writing and made by personal delivery, fax, by pre paid post or email to the receiving party at the address, fax number or email address specified in Item 14 of Schedule 1. Any such notice is to be deemed to have been received:
 - (a) in the case of personal delivery, on the date of delivery;
 - (b) in the case of post, on the 3rd working day after posting;
 - (c) in the case of email, on the date the email is sent provided that day is a working day and the email is sent before 5 pm, otherwise on the next working day.
- 21.2 If either party's details stated out in Item 14 of Schedule 1 change then the party whose details change must within 5 working days of such change provide the other party with the changed details.

22. What about the payment of costs?

- 22.1 The Concessionaire must pay the Grantor's legal costs and expenses of and incidental to preparing and signing any extension or variation of this Concession.
- 22.2 The Concessionaire must pay in full immediately and on demand all costs and fees (including solicitor's costs and fees of debt collecting agencies engaged by the Grantor) arising out of and associated with steps taken by the Grantor to enforce or attempt to enforce the Grantor's rights and powers under this Concession. This includes the right to recover outstanding money owed to the Grantor.

23. What about Identification cards and the Grantor's Approved Label?

23.1 Before commencing the Concession Activity the Concessionaire must, if required by the Grantor in Item 12 of Schedule 1, obtain Concessionaire

Godley Peaks Pastoral Lease - Guiding concession - May 2015 - DOCDM-1599933 (1)

Identification cards from the Grantor. The Grantor is to supply such cards to the Concessionaire on a cost recovery basis.

- 23.2 The Concessionaire and any person acting under the authority of the Concession must carry and display a Concession Identification card when carrying out the Concession Activity.
- 23.3 The Concessionaire must obtain sufficient cards to ensure all people acting under the authority of the Concession can carry and display such cards when undertaking the Concession Activity.
- 23.4 The Concessionaire may also access, use and/ or display the Grantor's "Approved Label". This right only exists once the Concessionaire agrees to comply with the Grantor's Approved Label terms and conditions and while the Concession remains operative. When the Concessionaire so requests the Grantor is to forward the Concessionaire an electronic link to the Approved Label. This electronic link is to contain the Approved Label terms and conditions.
- 23.5 The right under this clause 23.4 does not affect the obligation in this clause 23 to carry and display a Concession Identification card.

24. When can the conditions of the Concession be varied?

- 24.1 The Grantor may on each Concession Fee Review Date, after first consulting with the Concessionaire, vary any condition of this Concession to make the condition more effective in addressing adverse effects resulting from the Concession Activity.
- 24.2 Nothing in clause 24.1 otherwise affects the Grantor's rights to vary the Concession under section 17ZC of the Conservation Act 1987.

25. Are there any Special Conditions?

25.1 Special conditions are specified in Schedule 3. If there is a conflict between this Schedule 2 and the Special Conditions in Schedule 3, the Special Conditions shall prevail.

26. The Law

26.1 This Concession is to be governed by, and interpreted in accordance with the laws of New Zealand.

SCHEDULE 3

SPECIAL CONDITIONS

Use of toilets

1. Toilets must be used when camping in the vicinity of huts.

Hut Use

- 2. Hut use is on a first come first served basis and the Concessionaire and their clients together with all other concessionaires and their clients must not be occupy more than half the bunk space of any hut unless the bunks would otherwise be unoccupied. The Concessionaire or their clients must carry alternative accommodation on all overnight trips operated under this Concession and must not use a hut for more than two consecutive nights unless authorised in writing by the Department's appropriate Manager.
- 3. Where the Concessionaire makes use of Department administered huts, the Concessionaire must enter relevant details of the activity into any hut book provided by the Grantor.

Hut fees/ permits

4. Standard hut fees and camping fees must be paid by both guides and clients either before the trip begins or to the hut warden or camp manager, or if unavailable, to the relevant District office on completion of the trip.

Private land

5. This Concession does not confer any right of access over any private land or public conservation land leased by the Grantor. Any arrangements necessary for access over private land or leased land are the responsibility of the Concessionaire. In granting this Concession the Grantor does not warrant that such access can be obtained.

DOC staff

6. The Grantor may send any officer of the Department on any of the activities authorised during the term of this Concession for the purpose of assessing the impact on conservation values, the standard of service offered and compliance with the terms and conditions of the Concession, at no expense to the Grantor.

Use of tracks

7. The Concessionaire must ensure that, where provided, clients remain on formed tracks or well-used routes designed to protect natural and historic features of the Land, do not enter caves and do not exceed any loading limitations placed on facilities and structures.

Camp sites

8. The Concessionaire must ensure that no permanent camp sites are created nor stores or cache of any equipment is left on the Land or in any hut without the specific authority of the Department's appropriate Manager.

Wāhi Tapu

9. The Concessionaire must recognise the sensitivity of wāhi tapu and urupa and seek guidance of iwi who claim mana whenua over any parts of the Land prior to providing interpretation on matters of iwi cultural significance and recognise the sensitivity of wāhi tapu and urupa.

In respect to Ngāi Tahu

- 10. The Concessionaire is requested to consult the relevant Papatipu Runanga (www.ngaitahu.iwi.nz) if they wish to use Ngāi Tahu cultural information. If the Concessionaire wishes to use the Tōpuni or statutory acknowledgement information contained in schedules 14-108 of the Ngāi Tahu Claims Settlement Act 1998, or any Department produced interpretative material in respect to Ngāi Tahu cultural information, they are requested to notify the relevant Papatipu Rūnanga, as a matter of courtesy.
- 11. The Concessionaire must, as far as practicable, attend any workshops held by the Department for the purpose of providing information to concessionaires, which is to include the Ngāi Tahu values associated with Tōpuni areas.
- 12. The Concessionaire must ensure any persons employed by the Concessionaire are requested to recognise and provide for Ngãi Tahu values in the conduct of their activities.

Vehicle use

13. The Concessionaire must ensure that none of its vehicles or vehicles of its clients are taken off formed roads.

Vehicle parking

14. The Concessionaire must ensure that its vehicles and the vehicles of its clients are only parked only in designated parking areas.

Animals

15. The Concessionaire must not take, and must ensure that its clients do not take, any animals, including dogs or any domestic pets, onto the Land.

Weeds

16. The Concessionaire must take all precautions to ensure weeds are not introduced to the Land; this includes ensuring that all tyres, footwear, gaiters and packs used by the Concessionaire, its staff and clients are clean before entering the Land.

Interpretation materials

- 17. The Concessionaire must consult with and seek the guidance of iwi claiming mana whenua over any parts of the Land prior to providing interpretation on matters of cultural significance to such iwi.
- 18. The Concessionaire must provide detailed information of any historical, cultural or natural science interpretation provided by the Concessionaire to its clients in the course of the Concession Activity, to the Grantor within thirty days of the date of any such written request by the Grantor.

19. If the Grantor considers the interpretative material provided by the Concessionaire above unsatisfactory, the Concessionaire must prepare an interpretation plan for approval by the Grantor within 60 days of advice from the Grantor that this is required.

Recordings of bird songs

20. The Concessionaire must not and must ensure that its clients do not play recordings of bird songs on the Land.

Didymo

21. The Concessionaire must comply with all guidelines and notices put out by Biosecurity New Zealand regarding measures to avoid spreading the pest organism Didymosphenia geminate ("Didymo"), and or any other pest organism identified during the term of this Concession.

Other Special Conditions applying to All Concession Activities

- 22. The public have unrestricted access rights to the Land. The Grantor may erect signposts and mark the access routes through the Land for the benefit of the public.
- 23. This Licence authorises the Concessionaire to carry out the Concession Activity only, but does not authorise the Concessionaire to do anything on the Land which would not otherwise be permitted or which would require some other form of authority from the Grantor or any other person. Without limiting the effect of this clause and by way of example the Concessionaire is not authorised under this Concession to:
 - i. Access any part of the Land by air unless such access is with an air operator who is otherwise authorised to fly into the area concerned.
 - ii. Breach any law, regulation, by law, restriction, requirement, notice, sign or provision of any Conservation Management Strategy or management plan.
 - iii. Access any Land or water other than that listed in Schedule 1.
- 24. The Grantor reserves the right to request further or different activity related information in order to best monitor and determine any effects of the Concession Activity on the Land.
- 25. The Concessionaire shall ensure that guides brief clients on the historic and natural values of sites visited and take all practical measures to ensure that these values are not damaged or compromised in any way.
- 26. The Concessionaire shall ensure that guided groups show courtesy and consideration to other users of public conservation land at all times, and do not do not impede or hinder use of tracks and facilities by other users.
- 27. The Concessionaire shall ensure that in relation to camping activities;
 - i. Informal camping must ensure minimum impact is caused to the campsite. Existing campsites shall be used in preference to new sites.

- ii. Campsites (designated or informal) are not to be used for more than two consecutive nights.
- iii. Permanent or semi-permanent structures are not to be erected at any formal or informal campsite.
- 28. The Grantor will have no obligation to maintain any of the tracks within the Land. The Grantor shall have all rights and powers whether on the track or necessarily adjacent thereto to take all reasonable steps to repair and maintain the tracks to a standard suitable for Conservation management use and public pedestrian access. The Concessionaire may maintain any existing 4WD tracks within the Land. Maintenance of existing 4WD tracks is to be done in consultation with the Twizel Te Manahuna Office Manager.
- 30. If the Grantor determines that the conditions of this Document or the effects of Concession Activity should be monitored, the Concessionaire shall meet: either the full costs of any monitoring programme that is implemented; or, if the Grantor determines that the costs should be apportioned among several Concessionaires who use the same locations, part of the costs of the monitoring programme. These costs will include the Department's standard charge-out rates for staff time and the mileage rates for vehicle use associated with the monitoring programme.

A Guided Walking and Tramping Conditions:

DOC Facilities (eg huts) or informal campsites	Size (incl.	number of	Duration of visit (half or full days?)
	8	 2	

A1. The activity is limited to the following;

A2. The Concessionaire is to limit the party size to the maximum party size (including guide) identified in the table above.

B Guided Climbing and Rock Climbing Conditions:

B1. The activity is limited to the following;

DOC Facilities (eg huts) or informal campsites		number of	Duration of visit (half or full days?)
	8	2	_

B2. The Concessionaire is to limit the party size to the maximum party size (including guide) identified in the table above.

B3. The Grantor reserves the right to request further or different activity related information in order to best monitor and determine any effects of the Concession Activity on the Land.

B4. The Concessionaire is to ensure that there is no brushing or preparing of any rock surfaces.

B5. The Concessionaire is to ensure that any cracks, clefts or other hollows or depressions in the rock surface that support vegetative growth are not cleaned out or "gardened".

B6. The Concessionaire is to ensure that rock bolts are not placed on new climbing routes and existing bolts are not replaced.

B7. The Concessionaire is to ensure that minimal trampling of plants at rock bases occurs.

B8. The Concessionaire is to ensure all equipment is removed from the Land.

C Guided Mountaineering Conditions:

DOC Facilities (eg huts) or informal campsites		Frequency of Use (trips) per Year		
	8		2	

C1. The activity is limited to the following;

C2. The Concessionaire is to limit the party size to the maximum party size (including guide) identified in the table above.

C3. All persons employed to conduct high guiding activity will only take clients out on routes on which they are suitably qualified or experienced to do so.

D Guided Commercial Ground Based Hunting Conditions:

1.	DOC Facilities		<u>0;</u>		
		3	20	2	

1. D1. The activity is limited to the following;

D2. The Concessionaire is to limit the party size to the maximum party size (including guide) identified in the table above.

D3. The Grantor reserves the right to authorise any hunter who holds a valid hunting permit issued by the Director-General of Conservation to hunt on the Land.

D4. The Concessionaire shall not guide any hunting party on the Land without having first obtained hunting permits to cover all members of the specific hunting party, including the guide.

D5. The Grantor will be responsible for controlling the tahr numbers on the Land. The tahr numbers are to be maintained at a level that is in keeping with current Department of Conservation policies on tahr number control. The Grantor reserves the right to monitor the tahr numbers from time to time on the Land as they consider appropriate and if numbers are not at an acceptable level the Grantor reserves the right to undertake wild animal control management actions on the Land to bring the numbers down to a figure that is appropriate to current Departmental wild animal control policies.

SCHEDULE 4

Proposed Designations Plan

Godley Peaks Pastoral Lease - Guiding concession - May 2015 - DOCDM-1599933 (1) - 21 -

SCHEDULE 5 - Client Activity Return Form

Month/Year

cumulative impacts on conservation and recreation values. We need to know how many trips you make to each site and the party size of each trip. If you make more than one trip to the same site on the same day please record each trip on a new line (see example below). We appreciate your time and efforts to supply this information accurately and promptly. Supplying this information to the Department not only helps us to collect the fees for your activity, more importantly it helps us to understand the visitor use patterns and trends on public conservation lands. This information is critically important and helps us to manage the effects of all activities including the

y Solar (1 T									

Godley Peaks Pastoral Lease - Guiding concession - May 2015 - DOCDM-1599933 (1) - 22 -

ary
E
Ξ
2
Ψ.

		Total number of Clients per less 1 hour	Fee/Client per less 1 hour	Total number of Clients per ½ day	Fee/Client per ½ day	Total number of Clients per whole day	Fee/Client per whole day	Total Fees Payable
Guided walking, tramping, climbing, rock climbing, mountaineering, and ski touring (adult)	nping, Ing, ski touring		S1.00		S5.00		\$10.00	
Guided walking, tramping, climbing, rock climbing, mountaineering, and ski touring (Cbild)	nping, ing, ski touring		\$0.50		\$2.50		\$5.00	
Guided Commercial Ground Based Hunting Fee	Ground Based						\$15.00	
Guided Commercial Ground Based Hunting Companion fee	Ground Based fee						\$5.00	
	Plus GST	ST						69
	SUB TOTAL	OTAL						59
	Less M	Less Minimum Fees Paid						59
	TOTA	TOTAL FEE PAYABLE						59
		I certify th	I certify that the above figures	are a true and accur	figures are a true and accurate copy of records held by the Company.	ld by the Company.		

Signed by:

Thanks again for your help

Date / /

Godley Peaks Pastoral Lease - Guiding concession - May 2015 - DOCDM-1599933 (1) - 23 -

Appendix 5: Concession for irregular helicopter landings (DOCDM-1599935)



Department of Conservation *Te Papa Atawbai*

Concession Document (General Licence)

Concession Number:

THIS	CONCESSION	is made this	day of
TITIO	CONCESSION	is made this	uay UI

PARTIES:

Minister of Conservation (the Grantor)

Verity Farms Limited (the Concessionaire)

BACKGROUND

- **A.** The Department of Conservation ("Department") Te Papa Atawhai is responsible for managing and promoting conservation of the natural and historic heritage of New Zealand on behalf of, and for the benefit of, present and future New Zealanders.
- **B.** The land described in Item 1 of Schedule 1 as the Land is a Conservation Area or a Reserve under the management of the Grantor.
- **C.** Sections 66 and 68 of the Crown Pastoral Land Act 1998 authorise the Grantor to grant a Concession for a Concession Activity in a Conservation Area and a Reserve under section 17Q(1) of the Conservation Act 1987 or section 59A of the Reserves Act 1977 (whichever is relevant in the circumstances).
- **D.** The Concessionaire wishes to carry out the Concession Activity on the Land subject to the terms and conditions of this Concession.
- **E.** The parties wish to record the terms and conditions of this Concession and its Schedules.

OPERATIVE PARTS

F. In exercise of the Grantor's powers under the Conservation legislation the Grantor **GRANTS** to the Concessionaire a **LICENCE** to carry out the Concession Activity on the Land subject to the terms and conditions contained in this Concession and its Schedules.

Godley Peaks Pastoral Lease - Aircraft landing concession - May 2015 - DOCDM-1599935 Previous template docdm-1406816

SIGNED on behalf of the Minister of Conservation by [insert name and title of delegate]	
acting under delegated authority in the presence of:	in the presence of: Witness Signature:
Witness Signature:	Witness Name:
Witness Name:	Witness Occupation:
Witness Occupation:	Witness Address:
Witness Address:	
A copy of the Instrument of Delegation may be inspected at the Director-General's office at 18-22 Manners Street, Wellington.	

	Concessionaire's initials	Grantor's initials	
Godley Peaks Pastoral Lease - Aircraft landing concession Godley Peaks Pastoral Lease - Aircraft landing concession - Previous template docdm-14606816			

SCHEDULE 1

		As shown on the Drop and Designations Disc marks I (CAA)		
	Land	As shown on the Proposed Designations Plan marked (CA1) attached in Schedule 4		
1.	(clause 2)	Conservation Area of 10,629 hectares approximately on the Hall Range and Mistake Peak situated in the Land District of Canterbury		
		Irregular helicopter landings for the purpose of: a. Scenic landings		
2.	Concession Activity (clause 2)	 b. Heli-landings for the purpose of positioning recreationists (including ground-based hunters), and commercially guided groups other than for helicopter-snow activities. 		
		c. Heli-landings for the purpose of drop off and/or pick up of heli-hiking clients		
3.	Term (clause 3)	10 years commencing on the date that an approved plan is registered, affecting Certificate of Title CB 30B/914 (Canterbury Registry), vesting the land in the Crown as a conservation area or reserve, being the day of		
4.	Renewal(s) (clause 3)	None		
5.	Final Expiry Date (clause 3)	The 10th anniversary of the commencement date.		
		Activity Fee:		
		\$15.00 plus GST per person landed and picked up at a single site;		
		\$26.00 plus GST per person landed and picked up at multiple sites per day		
		For any client charged by the Concessionaire at a reduced rate as a 'child', the annual activity fee is to be half the adult client concession activity fee.		
6.	Concession Fee (clause 4)	Management Fee:		
	(ontable 4)	\$500.00 per annum plus GST, payment of this fee is subject to special condition 18 in Schedule 3.		
		Environmental Monitoring Fee (clause 9)		
		To be negotiated between Grantor and Concessionaire if required (see condition 9.2 in Schedule 2)		
		Community Services Contribution (clause 6) Not required		
	Concession Fee			
7.	Payment Instalments	out in clause 3 for payment instalments.		
	(clause 4)	b. Annually, in arrears upon commencement of the term set		

Concessionaire's initials

Grantor's initials

Godley Peaks Pastoral Lease - Aircraft landing concession - May 2015 - DOCDM-1599935 Godley Peaks Pastoral Lease - Aircraft landing concession - May 2015 - DOCDM-1599935 Previous template docdm-14606816

		out in clause 3 for activity returns, using the form attached in Schedule 5.
8.	Concession Fee Payment Date(s) (clause 4)	On or before the date specified on the invoice issued by the Grantor.
9.	PenaltyInterestRate(clause 4)	Double the current Official Cash Rate (OCR). <u>See Reserve Bank of New Zealand website</u>
10.	Concession Review Date(s)Fee(clause 5)	On the third anniversary of the date of commencement of this Concession set out in item 3 above and the corresponding date every 3 years thereafter until the expiry of the term.
11.	Health and Safety (clause 13)	Audited Safety Plan: Not required - covered by Civil Aviation Regulations
12.	Concessionaire Identification (clause 31)	Required
13.	Insurance (To be obtained by Concessionaire) (clause 12)	 (c) Aviation Legal Liability insurance for an amount no less than \$1,0000,000.00 (d) Statutory liability for an amount no less than \$500,000.00
14.	Addresses for Notices (clause 24)	Subject to review on each Concession Fee Review Date The Grantor's address is: Physical Address: Department of Conservation Conservation House 77 Lower Stuart Street Dunedin 9016 Postal Address: PO Box 5244 Moray Place Dunedin 9058 Phone: (03) 477 0677 Email: permissionsdunedin@doc.govt.nz
		The Concessionaire's address in New Zealand is:

Concessionaire's initials

Godley Peaks Pastoral Lease - Aircraft landing concession - May 2015 - DOCDM-1599935 Godley Peaks Pastoral Lease - Aircraft landing concession - May 2015 - DOCDM-1599935 Previous template docdm-14606816

		Address:
ļ		C/- Milnes Beatson Ld
		29 Wallace Street
		Motueka 7120
		Phone: (03) 528 8760
		Email: david@mba.net.nz
		Director's Address:
		210 Patons Rock Road
		RD2
		Takaka 7182
		Phone: 021 531 227
		Email: ann@verityfarms.co.nz
	Special Conditions	See Schedule 3.
15.	(clause 34)	

Note: The clause references are to the Grantors Standard Terms and Conditions of Licence set out in Schedule 2.

Note: Please initial each page of Schedule 1

,

Concessionaire's initials

SCHEDULE 2

STANDARD TERMS AND CONDITIONS OF LICENCE

1. Interpretation

- 1.1 Where the Grantor's consent or approval is expressly required under a provision of this Concession, the Concessionaire must seek the consent or approval of the Grantor for each separate occasion it is required notwithstanding that the Grantor has granted consent or approval for a like purpose on a prior occasion.
- 1.2 The Concessionaire is responsible for the acts and omissions of its employees, contractors, agents, clients and invitees (excluding other members of the public accessing the Land). The Concessionaire is liable under this Concession for any breach of the terms of the Concession by its employees, contractors, agents, clients and invitees (excluding other members of the public accessing the Land), as if the breach had been committed by the Concessionaire.
- 1.3 Where this Concession requires the Grantor to exercise a discretion or give any approval or provides for any other actions by the Grantor, then the Grantor must act reasonably and within a reasonable time. When a consent is required under this Concession such consent must not be unreasonably withheld.
- 1.4 Where this Concession provides for approvals, directions, reports and consents to be given by one party to the other, those approvals, directions, reports and consents must be given by notice in writing and clause 24 is to apply.

2. What is being authorised?

- 2.1 The Concessionaire is only allowed to use the Land for the Concession Activity.
- 2.2 The Concessionaire must exercise reasonable skill, care and diligence in carrying out the Concession Activity, in accordance with standards of skill, care and diligence normally practised by suitably qualified and experienced people in carrying out such activities.
- 2.3 The Concessionaire must provide the Grantor with evidence of the competency and qualifications of its employees and contractors if the Grantor so requests.

3. How long is the Concession for - the Term?

3.1 This Concession commences on the date set out in Item 3 of Schedule 1 and ends on the Final Expiry Date specified in Item 5 of Schedule 1.

4. What are the fees and when are they to be paid?

- 4.1 The Concessionaire must pay to the Grantor in the manner directed by the Grantor the Concession Fee and any other payment comprised in the Total Payment specified in Item 6 of Schedule 1 in the instalments and on the Concession Fee Payment Date specified in Items 7, and 8 of Schedule 1.
- 4.2 If the Concessionaire fails to make payment within 14 days of the Concession

Fee Payment Date then the Concessionaire is to pay interest on the unpaid Concession Fee from the Concession Fee Payment Date until the date of payment at the Penalty Interest Rate specified in Item 9 of Schedule 1.

5. When can the fee be reviewed?

- 5.1 The Grantor is to review the Concession Fee on the Concession Fee Review Dates in the following manner:
 - (a) The Grantor must commence the review not earlier than 3 months before a Concession Fee Review Date and no later than 9 months following the Concession Fee Review Date by giving notice to the Concessionaire.
 - (b) Subject to clause 5.1(e) the notice must specify the Concession Fee which the Grantor considers to be the market value for the Concession Activity as at the Concession Fee Review Date having regard to the matters specified in section 17Y(2) of the Conservation Act 1987.
 - (c) If, within 28 days of receipt of the Grantor's notice, the Concessionaire gives notice to the Grantor that the Concessionaire disputes the proposed new Concession Fee, the new Concession Fee is to be determined in accordance with clause 5.2(a) or (b).
 - (d) If the Concessionaire does not give notice to the Grantor under clause 5.1(c) the Concessionaire is to be deemed to have accepted the Concession Fee specified in the Grantor's notice.
 - (e) Notwithstanding clause 5.1(b) the new Concession Fee so determined or accepted must not be less than the Concession Fee payable during the year preceding the particular Concession Fee Review Date and is to be the Concession Fee payable by the Concessionaire from the Concession Fee Review Date.
 - (f) Until determination of the new Concession Fee, the Concession Fee payable by the Concessionaire from the Concession Fee Review Date is to be the Concession Fee specified in the Grantor's notice. On determination of the new Concession Fee an adjustment is to be made and paid, either by the Grantor or by the Concessionaire, whichever is applicable.
- 5.2 Immediately the Concessionaire gives notice to the Grantor under clause 5.1(c) the parties are to endeavour to agree on a new Concession Fee. If the parties are unable to reach agreement within 28 days the new Concession Fee is to be determined either:
 - (a) By one party giving notice to the other requiring the new Concession Fee to be determined by the Disputes clause (clause 22) or, if the parties agree,
 - (b) by registered valuers acting as experts and not as arbitrators as follows:
 - (i) Each party must appoint a valuer and give notice of the appointment to the other party within 14 days of the parties agreeing to determine the new Concession Fee by this means.
 - (ii) If the party receiving a notice does not appoint a valuer within the 14 day period the valuer appointed by the other party is to

DOCDM – Godley Peaks Pastoral Lease – Aircraft Landing Concession – May 2015 Godley Peaks Pastoral Lease - Aircraft landing concession - May 2015 - DOCDM-1599935 Previous template docdm-1406816 determine the new Concession Fee and that valuer's determination is to be binding on both parties.

- (iii) Before commencing their determination the respective valuers must appoint an umpire who need not be a registered valuer.
- (iv) The valuers are to determine the new Concession Fee which they consider to be the market value for the Concession Activity as at the Concession Fee Review Date having regard to the matters specified in section 17Y(2) of the Conservation Act 1987 but in no case is the new Concession Fee to be less than the Concession Fee payable during the year preceding the particular Concession Fee Review Date. If they fail to agree the Concession Fee is to be determined by the umpire.
- (v) In determining the Concession Fee the valuers or umpire are to disregard the annual cost to the Concessionaire to maintain or provide access to the Land.
- (vi) Each party is to be given the opportunity to make written or oral representations or submissions to the valuers or the umpire subject to such reasonable time and other limits as the valuers or the umpire may prescribe.
- (vii) The valuers or the umpire must have regard to any such representations but are not bound by them.
- (c) The valuers or umpire must give written notice to the parties once they have determined the new Concession Fee. The notice is to be binding on the parties and is to provide how the costs of the determination are to be borne.
- (d) (i) If a Concession Fee Review Date is postponed because of a moratorium imposed by law the Concession Fee Review is to take place at the date the moratorium is lifted or so soon afterwards as is practicable; and
 - (ii) the Concession Fee Review is to establish the market value for the Concession Activity as at that date instead of the date fixed under clause 5.1 having regard to the matters specified in section 17Y(2) of the Conservation Act 1987 but in no case is the new Concession Fee to be less than the Concession Fee payable during the year preceding the particular Concession Fee Review Date; and
 - (iii) each subsequent Concession Fee Review is to take place in accordance with the procedure fixed in clause 5.1.

6. Are there any other charges?

- 6.1 The Concessionaire must pay all levies rates and other charges, including utility charges payable in respect of the Land or for the services provided to the Land which relate to the Concessionaire's use of the Land or the carrying on of the Concession Activity.
- 6.2 The Grantor is not liable for any cost incurred in re-establishing the supply of any utilities in the event of any of them becoming unavailable for any reason.
- 6.3 Where the Grantor has paid such levies, rates or other charges the Concessionaire must on receipt of an invoice from the Grantor pay such sum

DOCDM – Godley Peaks Pastoral Lease – Aircraft Landing Concession – May 2015 Godley Peaks Pastoral Lease - Aircraft landing concession - May 2015 - DOCDM-1599935 Previous template docdm-1406816 to the Grantor within 14 days of receiving the invoice. If payment is not made within the 14 days then the Concessionaire is to pay interest on the unpaid sum from the date payment was due until the date of payment at the Penalty Interest Rate specified in Item 9 of Schedule 1.

6.4 Where the Grantor or Director-General has provided a community service, benefit or facility for the benefit of the Concessionaire under section 17ZH of the Conservation Act 1987, the Concessionaire must pay the Grantor the amount specified in Item 7 of Schedule 1 as part of the Total Payment specified in Item 6 of Schedule 1 on the Concession Fee Payment Dates specified in Item 8 of Schedule 1.

7. When can the Concession be assigned?

- 7.1 The Concessionaire must not transfer, sub licence, assign, mortgage or otherwise dispose of the Concessionaire's interest under this Concession or any part of it (which includes the Concessionaire entering into a contract or any other arrangement whatsoever whereby the Concession Activity would be carried out by a person (called the assignee) other than the Concessionaire) without the prior written consent of the Grantor.
- 7.2 The Grantor may in the Grantor's discretion decline any application for consent under clause 7.1.
- 7.3 Sections 17P, 17S, 17T, 17U, 17W, 17X, 17ZB and 17ZC of the Conservation Act 1987 apply to applications for consent under this clause unless the Grantor, in the Grantor's discretion, decides otherwise.
- 7.4 If the Grantor gives consent under this clause then the Concessionaire remains liable to observe and perform the terms and conditions of this Concession throughout the Term and is to procure from the Assignee a covenant to be bound by the terms and conditions of this Concession.
- 7.5 The Concessionaire must pay the costs reasonably incurred by the Grantor incidental to any application for consent, whether or not such consent is granted.
- 7.6 If the Concessionaire is not a publicly listed company then any change in the shareholding of the Concessionaire altering the effective control of the Concessionaire requires the approval of the Grantor.

8. What are the obligations to protect the environment?

- 8.1 The Concessionaire must not cut down or damage any vegetation; or damage any natural feature or historic resource on the Land; or light any fire on the Land without the prior consent of the Grantor.
- 8.2 The Concessionaire must at its cost keep the Land in a clean and tidy condition and free of weeds and all organisms specified as pests in a relevant pest management strategy.
- 8.3 The Concessionaire must not store hazardous materials on the Land nor store other materials on the Land where they may obstruct the public or create a nuisance
- 8.4 If directed by the Grantor, the Concessionaire must take all steps necessary to control, or, at the Grantor's option, contribute to the cost of controlling any pest, insect or rodent infestation occurring in or emanating from the Land or any structure or facility on the Land, and if directed by the Grantor, engage a

pest exterminator approved by the Grantor.

- 8.5 The Concessionaire must not bury:
 - (a) any toilet waste within 50 metres of a water source on the Land; or
 - (b) any animal or fish or any part thereof within 50 metres of any water body, water source or public road or track.

9. What about Environmental Monitoring?

- 9.1 The Concessionaire must, during the Term, if the Grantor so directs, design in consultation with the Grantor and undertake a programme to monitor and report on the environmental effects of the Concessionaire's use of the Land and conduct of the Concession Activity.
- 9.2 If the Grantor does not issue a direction under clause 9.1 the Concessionaire must, during the Term, pay to the Grantor the annual Environmental Monitoring Contribution specified in Item 6 of Schedule 1 to enable the Grantor to design and undertake a programme to monitor the environmental effects of the Concessionaire's use of the Land and conduct of the Concession Activity.

10. When can new structures be erected or land alterations occur?

10.1 The Concessionaire must not erect, alter or bring on to the Land any structure.

11. What about advertising?

11.1 The Concessionaire must not erect or display any signs or advertising on the Land.

12. What are the liabilities and who insures?

- 12.1 The Concessionaire agrees to use the Land at the Concessionaire's own risk and releases to the full extent permitted by law the Grantor and the Grantor's employees and agents from all claims and demands of any kind and from all liability which may arise in respect of any accident, damage or injury occurring to any person or property in or about the Land.
- 12.2 The Concessionaire must indemnify the Grantor against all claims, actions, losses and expenses of any nature which the Grantor may suffer or incur or for which the Grantor may become liable arising from the Concessionaire's performance of the Concession Activity.
- 12.3 This indemnity is to continue after the expiry or termination of this Concession in respect of any acts or omissions occurring or arising before its expiry or termination.
- 12.4 The Concessionaire has no responsibility or liability for costs, loss, or damage of whatsoever nature arising from any act or omission or lack of performance or any negligent or fraudulent act or omission by the Grantor, or any contractor or supplier to the Grantor, or any employee or agent of the Grantor.
- 12.5 Despite anything else in clause 12 the Concessionaire is not liable for any indirect or consequential damage or loss howsoever caused.

DOCDM – Godley Peaks Pastoral Lease – Aircraft Landing Concession – May 2015 Godley Peaks Pastoral Lease - Aircraft landing concession - May 2015 - DOCDM-1599935 Previous template docdm-1406816

- 12.6 The Grantor is not liable and does not accept any responsibility for damage to or interference with the Land , the Concession Activity, or to any structures, equipment or facilities on the Land or any other indirect or consequential damage or loss due to any natural disaster, vandalism, sabotage, fire, or exposure to the elements except where, subject to clause 12.7, such damage or interference is caused by any wilful act or omission of the Grantor, the Grantor's employees, agents or contractors.
- 12.7 Where the Grantor is found to be liable in accordance with clause 12.6, the total extent of the Grantor's liability is limited to \$1,000,000 in respect of the Concessionaire's structures, equipment and facilities.
- 12.8 Despite anything else in clause 12 the Grantor is not liable for any indirect or consequential damage or loss howsoever caused.
- 12.9 Without prejudice to or in any way limiting its liability under this clause 12 the Concessionaire at the Concessionaire's expense must take out and keep current policies for insurance and for the amounts not less than the sums specified in Item 13 of Schedule 1 with a substantial and reputable insurer.
- 12.10 After every three year period of the Term the Grantor may, on giving 10 working day's notice to the Concessionaire, alter the amounts of insurance required under clause 12.9. On receiving such notice the Concessionaire must within 10 working days take out and keep current policies for insurance and for the amounts not less than the sums specified in that notice.
- 12.11 The Concessionaire must provide to the Grantor within 5 working days of the Grantor so requesting:
 - (a) details of any insurance policies required to be obtained under this Concession, including any renewal policies if such renewal occurs during the Term; and/ or;
 - (b) a copy of the current certificate of such policies.

13. What about Health and Safety?

- 13.1 The Concessionaire must exercise the rights granted by this Concession in a safe and reliable manner and must comply with the Health and Safety in Employment Act 1992 and its regulations and all other provisions or requirements of any competent authority relating to the exercise of this Concession. The Concessionaire must comply with its safety plan (if one is required in Item 11 of Schedule 1), and with any safety directions of the Grantor.
- 13.2 Before commencing the Concession Activity the Concessionaire must, where the Concessionaire has Qualmark or Outdoorsmark certification, provide the Grantor with a copy of that certification.
- 13.3 If the Concessionaire does not hold Qualmark or Outdoorsmark certification then before commencing the Concession Activity the Concessionaire must, if required by Item 11 of Schedule 1:
 - (a) prepare a safety plan;
 - (b) have it audited by a suitably qualified person approved by the Grantor; and forward to the Grantor a certificate from the auditor certifying that the safety plan is suitable for the Concession Activity; and

DOCDM – Godley Peaks Pastoral Lease – Aircraft Landing Concession – May 2015 Godley Peaks Pastoral Lease - Aircraft landing concession - May 2015 - DOCDM-1599935 Previous template docdm-1406816

- (c) the Concessionaire must obtain from the auditor details as to when the safety plan is to be re-audited. The Concessionaire must comply with any such requirement to re-audit and forward a copy of the reaudit certificate to the Grantor within 5 working days of the certificate being issued.
- 13.4 For any Concession Activity that is subject to the Health and Safety in Employment (Adventure Activities) Regulations 2011, proof of registration with WorkSafe New Zealand will satisfy the Grantor's requirement under clause 13.3(b).
- 13.4 The Grantor may at any time request the Concessionaire to provide the Grantor with a copy of the current safety plan in which case the Concessionaire must provide the copy within 10 working days of receiving the request.
- 13.5 Receipt of the safety plan/audit certificate by the Grantor is not in any way to limit the obligations of the Concessionaire under clause 13 and is not to be construed as implying any responsibility or liability on the part of the Grantor.
- 13.6 The Concessionaire must:
 - (a) notify the Grantor of any natural events or activities on the Land or the surrounding area which may endanger the public or the environment;
 - (b) take all practicable steps to protect the safety of all persons present on the Land and must, where necessary, erect signposts warning the public of any dangers they may encounter as a result of the Concessionaire's operations;
 - (c) take all practicable steps to eliminate any dangers to the public and must clearly and permanently mark any that remain and of which the Concessionaire is aware;
 - (d) record and report to the Grantor all accidents involving serious harm within 24 hours of their occurrence and forward an investigation report within 3 days of the accident occurring;
 - (e) at the request of the Grantor make available any of the Concessionaire's directors, employees, servants or agents who in the opinion of the Grantor might assist any investigation by the Grantor into the cause of any such serious harm accident or incident;
 - (f) ensure that all contracts between the Concessionaire and any contractors contain, at a minimum, the same requirements as clause 13;
 - (g) be satisfied that facilities or equipment provided by the Grantor to enable the Concession Activity to be carried out meet the safety requirements of the Concessionaire;
 - (h) not bring onto the Land or any land administered by the Department any dangerous or hazardous material or equipment which is not required for purposes of the Concession Activity; and if such material or equipment is required as part of the Concession Activity, the Concessionaire must take all practicable steps at all times to ensure that the material or equipment is treated with due and proper care.

14. What are the compliance obligations of the Concessionaire?

- 14.1 The Concessionaire must comply where relevant:
 - (a) with the provisions of any conservation management strategy or conservation management plan under the Conservation Act 1987 or Part IIA of the Reserves Act 1977, or any general policy statement made under the Conservation Act 1987, Reserves Act 1977, National Parks Act 1980, or Wildlife Act 1953, or management plan under section 45 of the National Parks Act 1980, whichever is appropriate to the Land, together with any amendment or review of any policy, strategy or plan whether approved before, on, or after the date on which this Concession takes effect; and
 - (b) with the Conservation Act 1987, the Reserves Act 1977, the National Parks Act 1980, Wildlife Act 1953 and any other statute, ordinance, regulation, bylaw, or other enactment (collectively the "Legislation") affecting or relating to the Land or affecting or relating to the Concession Activity, including any regulations made under the Conservation Act 1987 and Wildlife Act 1953 or bylaws made under the Reserves Act 1977 or the National Parks Act 1980; and
 - (c) with all notices and requisitions of any competent authority affecting or relating to the Land or affecting or relating to the conduct of the Concession Activity; and
 - (d) with all Department signs and notices placed on or affecting the Land; and
 - (e) with all reasonable notices and directions of the Grantor concerning the Concession Activity on the Land.
- 14.2 The Concessionaire must comply with this Concession.
- 14.3 A breach or contravention by the Concessionaire of a relevant conservation management strategy, conservation management plan, management plan or any statement of general policy referred to in clause 14.1(a) is deemed to be a breach of this Concession.
- 14.4 A breach or contravention by the Concessionaire of any Legislation affecting or relating to the Land or affecting or relating to the Concession Activity is deemed to be a breach of this Concession.

15. What are the Grantor's rights to remedy defaults?

- 15.1 The Grantor may elect to remedy at any time, after giving notice, if practicable, any default by the Concessionaire under this Concession. Before electing to so remedy in accordance with this clause, the Grantor must, if practicable, first give the Concessionaire notice of the default and a reasonable opportunity to remedy the default.
- 15.2 The Concessionaire must pay to the Grantor forthwith on demand all reasonable costs and expenses incurred by the Grantor, including legal costs and expenses as between solicitor and client, in remedying such default. The Concessionaire is to pay interest on such costs and expenses if payment is not made within 14 day's of the Grantor's demand from the date of the demand until the date of payment at the Penalty Interest Rate specified in Item 12 of Schedule 1.

16. When can the Concession be suspended?

- 16.1 If, in the Grantor's opinion, there is a temporary risk to any natural or historic resource on or in the vicinity of the Land or to public safety whether arising from natural events such as earthquake, land slip, volcanic activity, flood, or arising in any other way, whether or not from any breach of the terms of this Concession on the part of the Concessionaire, then the Grantor may suspend this Concession.
- 16.2 If, in the Grantor's opinion, the activities of the Concessionaire are having or may have an adverse effect on the natural, historic or cultural values or resources of the Land and the Grantor considers that the effect can be avoided, remedied or mitigated to an extent satisfactory to the Grantor, then the Grantor may suspend this Concession until the Concessionaire avoids, remedies or mitigates the adverse effect to the Grantor's satisfaction.
- 16.3 The Grantor may suspend the Concession for such period as the Grantor determines where the Concessionaire has breached any terms of this Concession.
- 16.4 The Grantor may suspend this Concession while the Grantor investigates any of the circumstances contemplated in clauses 16.1 and 16.2 and also while the Grantor investigates any potential breach or possible offence by the Concessionaire, whether or not related to the Concession Activity under the Cónservation Act 1987 or any of the Acts mentioned in the First Schedule of that Act.
- 16.5 The word "investigates" in clause 16.4 includes the laying of charges and awaiting the decision of the Court.
- 16.6 During any period of temporary suspension arising under clauses 16.1 or 16.2 the Concession Fee payable by the Concessionaire is to abate in fair proportion to the loss of use by the Concessionaire of the Land.
- 16.7 The Grantor is not to be liable to the Concessionaire for any loss sustained by the Concessionaire by reason of the suspension of the Concession under this clause 16 including loss of profits.

17. When can the Concession be terminated?

- 17.1 The Grantor may terminate this Concession either in whole or in part:
 - (a) by 14 days notice to the Concessionaire if the Concession Fee or any other money payable to the Grantor under this Concession is in arrears and unpaid for 14 days after any of the days appointed for payment whether it has been lawfully demanded or not; or
 - (b) by 14 days notice to the Concessionaire or such sooner period as it appears necessary and reasonable to the Grantor if.
 - (i) the Concessionaire breaches any terms of this Concession and in the Grantor's sole opinion the breach is able to be rectified; and
 - (ii) the Grantor has notified the Concessionaire of the breach; and
 - (iii) the Concessionaire does not rectify the breach within 7 days of receiving notification; or such earlier time as specified by the Grantor; or

- (c) by notice in writing to the Concessionaire where the Concessionaire breaches any terms of this Concession and in the sole opinion of the Grantor the breach is not capable of being rectified; or
- (d) immediately by notice in writing to the Concessionaire where the Concessionaire breaches clauses 12.9 and 13; or
- (e) by notice in writing to the Concessionaire if the Concessionaire ceases to conduct the Concession Activity or, in the reasonable opinion of the Grantor, the conduct of the Concession Activity is manifestly inadequate; or
- (f) by notice in writing to the Concessionaire if the Concessionaire is convicted of an offence under the Conservation Act 1987 or any of the Acts listed in the First Schedule to that Act or any statute, ordinance, regulation, bylaw, or other enactment affecting or relating to the Land or which in the Grantor's sole opinion affects or relates to the Concession Activity; or
- (g) by notice in writing to the Concessionaire if the Concessionaire or the Guarantor is dissolved; or enters into any composition with or assignment for the benefit of its creditors; or is adjudged bankrupt; or being a company, has a receiver appointed; or is put into liquidation; or is placed under statutory management; or has a petition for winding up presented against it; or is otherwise unable to pay its debts as they fall due; or the estate or interest of the Concessionaire is made subject to a Writ of Sale or charging order; or the Concessionaire ceases to function or operate; or
- (h) immediately if there is, in the opinion of the Grantor, a permanent risk to public safety or to the natural and historic resources of the Land whether arising from the conduct of the Concession Activity or from natural causes such as earthquake, land slip, volcanic activity, flood, or arising in any other way, whether or not from any breach of the terms of this Concession on the part of the Concessionaire.
- 17.2 The Grantor may exercise its power to terminate under 17.1(h) without giving notice.
- 17.3 The Grantor may exercise the Grantor's right under this clause to terminate the Concession notwithstanding any prior waiver or failure to take action by the Grantor or any indulgence granted by the Grantor for any matter or default.
- 17.4 Termination of the Concession is not to prejudice or affect the accrued rights or claims and liabilities of the parties.

18. What happens on termination or expiry of the Concession?

- 18.1 On expiry or termination of this Concession, either as to all or part of the Land, the Concessionaire is not entitled to compensation for any structures or other improvements placed or carried out by the Concessionaire on the Land.
- 18.2 The Concessionaire may, with the Grantor's written consent, remove any specified structures and other improvements on the Land. Removal under this clause must occur within the time specified by the Grantor and the Concessionaire is to make good any damage and leave the Land and other public conservation land affected by the removal in a clean and tidy condition.

18.3 The Concessionaire must, if the Grantor gives written notice, remove any specified structures and other improvements on the Land. Removal under this clause must occur within the time specified by the Grantor and the Concessionaire is to make good any damage and leave the Land and other public conservation land affected by the removal in a clean and tidy condition and replant the Land with indigenous vegetation of a similar abundance and diversity as at the commencement of the Term. If before the expiry of the Term the Concessionaire makes an application for a further concession in respect of the same Concession Activity on the Land then the Grantor can not require such removal and reinstatement until such time as that concession application has been determined. If a new concession is granted then removal and reinstatement can not be required until the expiry or termination of the new concession.

19. When is the Grantor's consent required?

19.1 Where the Grantor's consent or approval is expressly required under this Concession then the Concessionaire must seek that approval or consent for each separate time it is required even though the Grantor may have given approval or consent for a like purpose on a prior occasion. Any such consent or approval may be made on such conditions as the Grantor considers appropriate.

20. Are there limitations on public access and closure?

20.1 The Concessionaire acknowledges that the Land is open to the public for access and that the Grantor may close public access during periods of high fire hazard or for reasons of public safety or emergency.

21. What about other concessions?

21.1 Nothing expressed or implied in this Concession is to be construed as preventing the Grantor from granting other concessions, whether similar or not, to other persons provided that the Grantor must not grant another concession that would derogate in any material way from the Concessionaire's ability to carry out the Concession Activity.

22. How will disputes be resolved?

- 22.1 If a dispute arises between the parties in connection with this Concession the parties must, without prejudice to any other rights or entitlements they may have, attempt to resolve the dispute by agreement using informal dispute resolution techniques such as negotiation, mediation, independent expert appraisal or any other alternative dispute resolution technique. The rules governing any such technique adopted are to be agreed between the parties.
- 22.2 If the dispute cannot be resolved by agreement within 14 days of written notice by one party to the other (or such further period as the parties may agree to in writing) either party may refer the dispute to the Disputes Tribunal, where relevant, or to arbitration, which arbitration is to be carried out in accordance with the provisions of the Arbitration Act 1996.
- 22.3 If the parties do not agree on an arbitrator within 10 working days of a party giving written notice of the requirement to appoint an arbitrator the President of the New Zealand Law Society is to appoint the arbitrator. In either case the

DOCDM – Godley Peaks Pastoral Lease – Aircraft Landing Concession – May 2015 Godley Peaks Pastoral Lease - Aircraft landing concession - May 2015 - DOCDM-1599935 Previous template docdm-1406816

arbitrator must not be a person who has participated in an informal dispute resolution procedure in respect of the dispute.

- 22.4 The arbitrator must include in the arbitration award reasons for the determination.
- 22.5 Despite the existence of a dispute, each party must continue to perform its obligations under this Concession.

23. What about prosecution for offences?

- 23.1 Where any breach of this Concession by the Concessionaire also constitutes an offence under the Resource Management Act 1991, the Conservation Act 1987, or any of the Acts listed in the First Schedule to that Act:
 - (a) no waiver or failure to act by the Grantor under this Concession is to preclude the Grantor from prosecuting the Concessionaire; and
 - (b) no failure by the Grantor to prosecute the Concessionaire is to preclude the Grantor from exercising the Grantor's remedies under this Concession; and
 - (c) any action of the Grantor in prosecuting the Concessionaire is not to preclude the Grantor from exercising the Grantor's remedies under this Concession.

24. How are notices sent and when are they received?

- 24.1 Any notice to be given under this Concession is to be in writing and made by personal delivery, fax, by pre paid post or email to the receiving party at the address, fax number or email address specified in Item 14 or 15 of Schedule 1. Any such notice is to be deemed to have been received:
 - (a) in the case of personal delivery, on the date of delivery;
 - (b) in the case of post, on the 3^{rd} working day after posting;
 - (c) in the case of email, on the day the email was sent provided it is sent before 5 pm on a working day, otherwise on the next working day.
- 24.2 If any party's details specified in Item 14 or 15 of Schedule 1 change then the party whose details change must within 5 working days of such change provide the other party with the changed details.

25. What is the scope of the Concession?

25.1 Except as provided by legislation, this Concession and any written variation agreed by the parties contain the entire understanding between the parties with reference to the subject matter of this Concession and there is no other agreement, representation or warranty whether it is expressed or implied which in any way extends, defines or otherwise relates to the provisions of this Concession.

26. Can provisions be severed?

26.1 Any illegality, or invalidity or unenforceability of any provision in this Concession is not to affect the legality, validity or enforceability of any other provisions.

DOCDM -- Godley Peaks Pastoral Lease -- Aircraft Landing Concession -- May 2015 Godley Peaks Pastoral Lease -- Aircraft landing concession -- May 2015 -- DOCDM-1599935 Previous template docdm-1406816

27. What about the payment of costs?

- 27.1 The Concessionaire must pay the Grantor's legal costs and expenses of and incidental to preparing and signing any extension or variation of this Concession.
- 27.2 The Concessionaire must pay in full immediately and on demand all costs and fees (including solicitor's costs and fees of debt collecting agencies engaged by the Grantor) arising out of and associated with steps taken by the Grantor to enforce or attempt to enforce the Grantor's rights and powers under this Concession including the right to recover outstanding money owed to the Grantor.

28. What is the relationship of parties?

- 28.1 Nothing expressed or implied in this Concession is to be construed as:
 - (a) constituting the parties as partners or joint venturers;
 - (b) conferring on the Concessionaire any right of exclusive occupation or use of the Land;
 - (c) granting any exclusive estate or interest in the Land to the Concessionaire;
 - (d) affecting the rights of the Grantor and the public to have access across the Land.

29. What about a Guarantee?

- 29.1 Where the Grantor has in Item 15 of Schedule 1 required this Concession to be guaranteed by a third party the following clauses are to apply.
- 29.2 In consideration of the Grantor entering into this Concession at the Guarantor's request the Guarantor:
 - (a) guarantees payment of the Concession Fee and the performance by the Concessionaire of the covenants in this Concession; and
 - (b) indemnifies the Grantor against any loss the Grantor might suffer should the Concession be lawfully disclaimed or abandoned by any liquidator, receiver or other persons.
- 29.3 The Guarantor covenants with the Grantor that:
 - (a) no release, delay, or other indulgence given by the Grantor to the Concessionaire, to the Concessionaire's successors or assigns, or any other thing whereby the Guarantor would have been released had the Guarantor been merely a surety is to release, prejudice, or affect the liability of the Guarantor as a Guarantor or as indemnifier;
 - (b) as between the Guarantor and Grantor the Guarantor may, for all purposes, be treated as the Concessionaire and the Grantor is under no obligation to take proceedings against the Concessionaire before taking proceedings against the Guarantor;
 - (c) the guarantee is for the benefit of and may be enforced by any person entitled for the time being to receive the Concession Fee;
 - (d) any assignment of this Concession and any Concession Fee Review in accordance with this Concession are not to release the Guarantor from liability;

DOCDM – Godley Peaks Pastoral Lease – Aircraft Landing Concession – May 2015 Godley Peaks Pastoral Lease - Aircraft landing concession - May 2015 - DOCDM-1599935 Previous template docdm-1406816 (e) should there be more than one Guarantor the liability of each Guarantor under this Guarantee is to be joint and several.

30. What about Co-Siting? [clauses deleted – not-relevant]

31. What about Identification cards?

- 31.1 Before commencing the Concession Activity the Concessionaire must, if required by the Grantor in Item 12 of Schedule 1, obtain Concessionaire Identification cards from the Grantor. The Grantor is to supply such cards to the Concessionaire on a cost recovery basis.
- 31.2 The Concessionaire and any person acting under the authority of the Concession must carry and display a Concession Identification card when carrying out the Concession Activity.
- 31.3 The Concessionaire must obtain sufficient cards to ensure all people acting under the authority of the Concession can carry and display such cards when undertaking the Concession Activity.
- 31.4 The Concessionaire may also access, use and/ or display the Grantor's "Approved Label". This right only exists once the Concessionaire agrees to comply with the Grantor's Approved Label terms and conditions and while the Concession remains operative. When the Concessionaire so requests the Grantor is to forward the Concessionaire an electronic link to the Approved Label. This electronic link is to contain the Approved Label terms and conditions.
- 31.5 The right under this clause 31.4 does not affect the obligation in this clause 31 to carry and display a Concession Identification card.

32. Which clauses survive termination?

32.1 Clauses 12 and 24 survive the termination of this Concession.

33. When can the conditions of the Concession be varied?

- 33.1 The Grantor may on each Concession Fee Review Date, after first consulting with the Concessionaire, vary any condition of this Concession to make the condition more effective in addressing adverse effects resulting from the Concession Activity.
- 33.2 Nothing in clause 33.1 otherwise affects the Grantor's rights to vary the Concession under section 17ZC of the Conservation Act 1987.

34. Are there any Special Conditions?

34.1 Special conditions are specified in Schedule 3.

35. The Law

35.1 This Concession is to be governed by, and interpreted in accordance with the laws of New Zealand

SPECIAL CONDITIONS

In respect to Ngāi Tahu

- 1. The Concessionaire is requested to consult the relevant Papatipu Runanga (www.ngaitahu.iwi.nz) if they wish to use Ngāi Tahu cultural information. If the Concessionaire wishes to use the Tōpuni or statutory acknowledgement information contained in schedules 14-108 of the Ngāi Tahu Claims Settlement Act 1998, or any Department produced interpretative material in respect to Ngāi Tahu cultural information, they are requested to notify the relevant Papatipu Runanga, as a matter of courtesy.
- 2. The Concessionaire must, as far as practicable, attend any workshops held by the Department for the purpose of providing information to concessionaires, which is to include the Ngāi Tahu values associated with Tōpuni areas.
- 3. The Concessionaire must ensure any persons employed by the Concessionaire are requested to recognise and provide for Ngāi Tahu values in the conduct of their activities.

Didymo

4. The Concessionaire must comply with all guidelines and notices put out by Biosecurity New Zealand regarding measures to avoid spreading the pest organism Didymosphenia geminate ("Didymo"), and or any other pest organism identified during the term of this Concession.

Aircraft Use

- 5. The Concessionaire must ensure that aircraft idle times on the ground are kept to a practicable minimum.
- 6. The Concessionaire must not refuel, leave any fuel drums or construct any fuel dumps on the Land, unless in an emergency situation.
- 7. During the term of the concession, where Grantor believes that the effects of aircraft noise should be further reduced, the Grantor may, by notice, require the concessionaire to either undertake measures to minimise the effects of noise on conservation values or become accredited to a recognised noise abatement and disputes resolution programme. If such notice is given by the Grantor, the concessionaire must:

(i) if required to undertake measures to minimise the effects of noise on conservation values within 3 months from receiving the notice undertake those measures to the satisfaction of the Grantor until the Final Expiry Date.

(ii) if required to become accredited to a recognised noise abatement and disputes resolution programme within 3 months from receiving the notice provide proof to the Grantor that such accreditation has been completed and must keep their participation in that programme or training current until the Final Expiry Date.

The Grantor may, at any time, issue a subsequent notice(s) requiring the Concessionaire to implement the other option.

- 8. The Concessionaire shall not land within 250m of any hut where possible.
- 9. The Grantor may close access to any part of the Land for management purposes on a temporary basis, and shall give notice in writing to the Concessionaire prior to the temporary closure. The Concessionaire shall not land at any site that has been closed under this clause until given notice in writing by the Grantor that the temporary closure has been lifted.
- 10. The concessionaire shall submit activity return forms which clearly illustrate where the landings were made e.g. grid reference or GPS points, and declaration of fees form, attached as Schedule 5 on a yearly basis or on demand as requested by the Department.
- 11. The concessionaire must avoid, where possible, overflying tramping routes, tracks and other visitor facilities.
- 12. All helicopter operations under this licence shall be in compliance with Civil Aviation Authority regulations.
- 13. The concessionaire must advise any of their clients that are anglers or hunters that they require a valid hunting permit or fishing licence to conduct their activities on public conservation land.
- 14. The Concessionaire shall not carry under slung loads into Public Conservation Lands. Sling loads (for example deer or tahr shot by recreational hunters) may be removed from the Land.
- 15. The Concessionaire must adhere to the Helicopter Association International 'Fly Neighborly Guide'.
- 16. The Concessionaire must possess a current Civil Aviation Authority Air Operator Certificate, or Transitional Air Operator Certificate (Air Service Operations), as appropriate, as well as appropriate airworthiness certificates for all aircraft used, pursuant to Clause 11 of this concession. Before commencing this activity, copies of these certifications must be submitted to the Grantor.
- 17. Safety and survival equipment is to be carried at all times to the standards of the licensee's Flight Operations manual and as specified by CAA regulations.

Other Special Conditions

18. Provided that the Concessionaire for this concession being Verity Farms Limited is also the Concessionaire for the Guiding Licence concession being _______[insert reference], then the concession Annual Management Fee in Schedule 1, item 6 is not payable. This is on the basis that while the two concessions are both held by Verity Limited, one Annual Management Fee is sufficient.

- 19. This Licence authorises the Concessionaire to carry out the Concession Activity only, but does not authorise the Concessionaire to do anything which would not otherwise be permitted or which would require some other form of authority from the Grantor or any other person. Without limiting the effect of this clause and by way of example the Concessionaire is not authorised under this Concession to:
 - i. Access any part of the Land by air unless such access is with an air operator who is otherwise authorised to fly into the area concerned.
 - ii. Breach any law, regulation, by law, restriction, requirement, notice, sign or provision of any Conservation Management Strategy or management plan.
 - iii. Access any Land or water other than that listed in Schedule 1.
- 20. The Grantor reserves the right to request further or different activity related information in order to best monitor and determine any effects of the Concession Activity on the Land.
- 21. The Concessionaire shall ensure that guides brief clients on the historic and natural values of sites visited and take all practical measures to ensure that these values are not damaged or compromised in any way.

Designations Plan or map Any further plan if required DOCDM – Godley Peaks Pastoral Lease – Aircraft Landing Concession – May 2015 Godley Peaks Pastoral Lease - Aircraft landing concession - May 2015 - DOCDM-1599935 Previous template docdm-1406816 -23 -

.

Anna Anna Anna Anna Anna Anna Anna Anna											
in an											
· · · · ·					 						
									-		

DOCDM – Godley Peaks Pastoral Lease – Aircraft Landing Concession – May 2015 Godley Peaks Pastoral Lease - Aircraft landing concession - May 2015 - DOCDM-1599935 Previous template docdm-1406816 - 24 -

Appendix 6: Concession for heli skiing and snowboarding (DOCDM-1599937)



Department of Conservation Te Papa Atawhai

Concession Document (General Licence)

Concession Number:

r	
THIS	CONCESSION is made this day of
PART	TIES:
Minis	ster of Conservation (the Grantor)
Veriț	y Farms Limited (the Concessionaire)
BACK	KGROUND
А.	The Department of Conservation ("Department") Te Papa Atawhai is responsible for managing and promoting conservation of the natural and historic heritage of New Zealand on behalf of, and for the benefit of, present and future New Zealanders.
В.	The land described in Item 1 of Schedule 1 as the Land is a Conservation Area or a Reserve under the management of the Grantor.
C.	Sections 66 and 68 of the Crown Pastoral Land Act 1998 authorise the Grantor to grant a Concession for a Concession Activity in a Conservation Area and a Reserve under section $17Q(1)$ of the Conservation Act 1987 or section 59A of the Reserves Act 1977 (whichever is relevant in the circumstances).
D.	The Concessionaire wishes to carry out the Concession Activity on the Land subject to the terms and conditions of this Concession.
Е.	The parties wish to record the terms and conditions of this Concession and its Schedules.
OPER	RATIVE PARTS
F.	In exercise of the Grantor's powers under the Conservation legislation the Grantor GRANTS to the Concessionaire a LICENCE to carry out the Concession Activity on the Land subject to the terms and conditions contained in this Concession and its Schedules.

SIGNED on behalf of the Minister of Conservation by [insert name and title of delegate]	SIGNED by Verity Farms Limited by its Director Ann Poindexter:
acting under delegated authority in the presence of:	in the presence of: Witness Signature:
Witness Signature:	Witness Name:
Witness Name:	Witness Occupation:
Witness Occupation:	Witness Address:
Witness Address:	
A copy of the Instrument of Delegation may be inspected at the Director-General's office at 18-22 Manners Street, Wellington.	

		As shown on the Proposed Designations Plan marked (CA1
	Land	attached in Schedule 4
1.	(clause 2)	Conservation Area of 10,629 hectares approximately on the Hall Range and Mistake Peak situated in the Land District o Canterbury
	Concession Activity	Helicopter landings for the purpose of heli snowboarding
2.	(clause 2)	and heli-skiing; and Guided skiing and guided snowboarding
		10 years commencing on the date that an approved plan is
3.	Term (clause 3)	registered, affecting Certificate of Title CB 30B/914 (Canterbury Registry), vesting the land in the Crown as a conservation area or reserve, being the day of
4.	Renewal(s)	None
•	(clause 3)	
_	Final Expiry Date	The 10th anniversary of the commencement date.
5.	(clause 3)	
		Activity Fee:
		\$27.00 plus GST per person per day
		Management Fee:
_	Concession Fee	\$500.00 per annum plus GST, payment of this fee is subject to special condition 14 in Schedule 3.
6.	(clause 4)	Environmental Monitoring Fee (clause 9)
		To be negotiated between Grantor and Concessionaire if required (see condition 9.2 in Schedule 2)
		Community Services Contribution (clause 6)
		Not required
	Concession Fee	a. Annually, in arrears upon commencement of the term se out in clause 3 for payment instalments.
7.	Payment Instalments (clause 4)	b. Annually, in arrears upon commencement of the term se out in clause 3 for activity returns, using the form attached in
		Schedule 5.
	Concession Fee Payment Date(s)	On or before the date specified on the invoice issued by th Grantor.
8.		
	(clause 4)	
	Penalty Interest	
	D-+-	
9.	Rate (clause 4)	Double the current Official Cash Rate (OCR).

Concessionaire's initials

. . . .

11.		of the term.
	Health and Safety (clause 13)	Audited Safety Plan: Not required - covered by Civil Aviation Regulations
12.	Concessionaire Identification	Required
	(clause 31)	
13.	Insurance (To be obtained by Concessionaire) (clause 12)	Types and amounts: Public Liability Insurance for: (a) General indemnity for an amount no less than \$2,000,000.00; and (b) Forest and Rural Fires Act extension for an amount no less than \$2,000,000.00; and (c) Aviation Legal Liability insurance for an amount no less than \$1,0000,000.00 (d) Statutory liability for an amount no less than \$500,000.00
		Subject to review on each Concession Fee Review DateThe Grantor's address is:Physical Address:Department of ConservationConservation House77 Lower Stuart Street
14.	Addresses for Notices	Dunedin 9016 Postal Address:
	(clause 24)	PO Box 5244 Moray Place Dunedin 9058
		Phone: (03) 477 0677 Email: permissionsdunedin@doc.govt.nz
		The Concessionaire's address in New Zealand is: C/- Milnes Beatson Ld 29 Wallace Street Motueka 7120 Phone: (03) 528 8760 Email: david@mba.net.nz
		Director's Address: 210 Patons Rock Road RD2 Takaka 7182

Concessionaire's initials

Grantor's initials

		Phone: 021 531 227 Email: ann@verityfarms.co.nz
15.	Special Conditions	See Schedule 3.
	(clause 34)	

Note: The clause references are to the Grantors Standard Terms and Conditions of Licence set out in Schedule 2.

Note: Please initial each page of Schedule 1

Concessionaire's initials

- .---

STANDARD TERMS AND CONDITIONS OF LICENCE

1. Interpretation

1.1 **"Background"** means the matters referred to under the heading 'Background" on the first page of this Document, and words used in the Background have the meaning given to them in this clause 1.1.

"Commencement date" means the date that an approved plan is registered vesting the Land in the Crown as a conservation area or reserve.

"Concession" means this document and any subsequent amendments and all schedules, annexures, and plans attached to it.

"Concession Activity" means the activity described in item 2 of Schedule 1.

"Conservation Area" has the same meaning as "conservation area" in section 2 of the Conservation Act 1987.

"Director-General" means the Director-General of Conservation

"Reserve" has the same meaning as "reserve" in section 59A of the Reserves Act 1977.

"**Term**" means the period of time specified in Item 5 of Schedule 1 during which this Concession operates.

- 1.2 Where the Grantor's consent or approval is expressly required under a provision of this Concession, the Concessionaire must seek the consent or approval of the Grantor for each separate occasion it is required notwithstanding that the Grantor has granted consent or approval for a like purpose on a prior occasion.
- 1.3 The Concessionaire is responsible for the acts and omissions of its employees, contractors, agents, clients and invitees (excluding other members of the public accessing the Land). The Concessionaire is liable under this Concession for any breach of the terms of the Concession by its employees, contractors, agents, clients and invitees (excluding other members of the public accessing the Land), as if the breach had been committed by the Concessionaire.
- 1.4 Where this Concession requires the Grantor to exercise a discretion or give any approval or provides for any other actions by the Grantor, then the Grantor must act reasonably and within a reasonable time. When a consent is required under this Concession such consent must not be unreasonably withheld.
- 1.5 Where this Concession provides for approvals, directions, reports and consents to be given by one party to the other, those approvals, directions, reports and consents must be given by notice in writing and clause 24 is to apply.

2. What is being authorised?

- 2.1 The Concessionaire is only allowed to use the Land for the Concession Activity.
- 2.2 The Concessionaire must exercise reasonable skill, care and diligence in carrying out the Concession Activity, in accordance with standards of skill, care and diligence normally practised by suitably qualified and experienced people in carrying out such activities.

2.3 The Concessionaire must provide the Grantor with evidence of the competency and qualifications of its employees and contractors if the Grantor so requests.

3. How long is the Concession for - the Term?

3.1 This Concession commences on the date set out in Item 3 of Schedule 1 and ends on the Final Expiry Date specified in Item 5 of Schedule 1.

4. What are the fees and when are they to be paid?

- 4.1 The Concessionaire must pay to the Grantor in the manner directed by the Grantor the Concession Fee and any other payment comprised in the Total Payment specified in Item 6 of Schedule 1 in the instalments and on the Concession Fee Payment Date specified in Items 7, and 8 of Schedule 1.
- 4.2 If the Concessionaire fails to make payment within 14 days of the Concession Fee Payment Date then the Concessionaire is to pay interest on the unpaid Concession Fee from the Concession Fee Payment Date until the date of payment at the Penalty Interest Rate specified in Item 9 of Schedule 1.

5. When can the fee be reviewed?

- 5.1 The Grantor is to review the Concession Fee on the Concession Fee Review Dates in the following manner:
 - (a) The Grantor must commence the review not earlier than 3 months before a Concession Fee Review Date and no later than 9 months following the Concession Fee Review Date by giving notice to the Concessionaire.
 - (b) Subject to clause 5.1(e) the notice must specify the Concession Fee which the Grantor considers to be the market value for the Concession Activity as at the Concession Fee Review Date having regard to the matters specified in section 17Y(2) of the Conservation Act 1987.
 - (c) If, within 28 days of receipt of the Grantor's notice, the Concessionaire gives notice to the Grantor that the Concessionaire disputes the proposed new Concession Fee, the new Concession Fee is to be determined in accordance with clause 5.2(a) or (b).
 - (d) If the Concessionaire does not give notice to the Grantor under clause 5.1(c) the Concessionaire is to be deemed to have accepted the Concession Fee specified in the Grantor's notice.
 - (e) Notwithstanding clause 5.1(b) the new Concession Fee so determined or accepted must not be less than the Concession Fee payable during the year preceding the particular Concession Fee Review Date and is to be the Concession Fee payable by the Concessionaire from the Concession Fee Review Date.
 - (f) Until determination of the new Concession Fee, the Concession Fee payable by the Concessionaire from the Concession Fee Review Date is to be the Concession Fee specified in the Grantor's notice. On determination of the new Concession Fee an adjustment is to be made and paid, either by the Grantor or by the Concessionaire, whichever is applicable.

- 5.2 Immediately the Concessionaire gives notice to the Grantor under clause 5.1(c) the parties are to endeavour to agree on a new Concession Fee. If the parties are unable to reach agreement within 28 days the new Concession Fee is to be determined either:
 - (a) By one party giving notice to the other requiring the new Concession Fee to be determined by the Disputes clause (clause 22) or, if the parties agree,
 - (b) by registered valuers acting as experts and not as arbitrators as follows:
 - (i) Each party must appoint a valuer and give notice of the appointment to the other party within 14 days of the parties agreeing to determine the new Concession Fee by this means.
 - (ii) If the party receiving a notice does not appoint a valuer within the 14 day period the valuer appointed by the other party is to determine the new Concession Fee and that valuer's determination is to be binding on both parties.
 - (iii) Before commencing their determination the respective valuers must appoint an umpire who need not be a registered valuer.
 - (iv) The valuers are to determine the new Concession Fee which they consider to be the market value for the Concession Activity as at the Concession Fee Review Date having regard to the matters specified in section 17Y(2) of the Conservation Act 1987 but in no case is the new Concession Fee to be less than the Concession Fee payable during the year preceding the particular Concession Fee Review Date. If they fail to agree the Concession Fee is to be determined by the umpire.
 - (v) In determining the Concession Fee the valuers or umpire are to disregard the annual cost to the Concessionaire to maintain or provide access to the Land.
 - (vi) Each party is to be given the opportunity to make written or oral representations or submissions to the valuers or the umpire subject to such reasonable time and other limits as the valuers or the umpire may prescribe.
 - (vii) The valuers or the umpire must have regard to any such representations but are not bound by them.
 - (c) The valuers or umpire must give written notice to the parties once they have determined the new Concession Fee. The notice is to be binding on the parties and is to provide how the costs of the determination are to be borne.
 - (d) (i) If a Concession Fee Review Date is postponed because of a moratorium imposed by law the Concession Fee Review is to take place at the date the moratorium is lifted or so soon afterwards as is practicable; and
 - (ii) the Concession Fee Review is to establish the market value for the Concession Activity as at that date instead of the date fixed under clause 5.1 having regard to the matters specified in section 17Y(2) of the Conservation Act 1987 but in no case is the new Concession Fee to be less than the Concession Fee payable during the year preceding the particular Concession Fee Review Date; and

(iii) each subsequent Concession Fee Review is to take place in accordance with the procedure fixed in clause 5.1.

6. Are there any other charges?

- 6.1 The Concessionaire must pay all levies rates and other charges, including utility charges payable in respect of the Land or for the services provided to the Land which relate to the Concessionaire's use of the Land or the carrying on of the Concession Activity.
- 6.2 The Grantor is not liable for any cost incurred in re-establishing the supply of any utilities in the event of any of them becoming unavailable for any reason.
- 6.3 Where the Grantor has paid such levies, rates or other charges the Concessionaire must on receipt of an invoice from the Grantor pay such sum to the Grantor within 14 days of receiving the invoice. If payment is not made within the 14 days then the Concessionaire is to pay interest on the unpaid sum from the date payment was due until the date of payment at the Penalty Interest Rate specified in Item 9 of Schedule 1.
- 6.4 Where the Grantor or Director-General has provided a community service, benefit or facility for the benefit of the Concessionaire under section 17ZH of the Conservation Act 1987, the Concessionaire must pay the Grantor the amount specified in Item 7 of Schedule 1 as part of the Total Payment specified in Item 6 of Schedule 1 on the Concession Fee Payment Dates specified in Item 8 of Schedule 1.

7. When can the Concession be assigned?

- 7.1 The Concessionaire must not transfer, sub licence, assign, mortgage or otherwise dispose of the Concessionaire's interest under this Concession or any part of it (which includes the Concessionaire entering into a contract or any other arrangement whatsoever whereby the Concession Activity would be carried out by a person (called the assignee) other than the Concessionaire) without the prior written consent of the Grantor.
- 7.2 The Grantor may in the Grantor's discretion decline any application for consent under clause 7.1.
- 7.3 Sections 17P, 17S, 17T, 17U, 17W, 17X, 17ZB and 17ZC of the Conservation Act 1987 apply to applications for consent under this clause unless the Grantor, in the Grantor's discretion, decides otherwise.
- 7.4 If the Grantor gives consent under this clause then the Concessionaire remains liable to observe and perform the terms and conditions of this Concession throughout the Term and is to procure from the Assignee a covenant to be bound by the terms and conditions of this Concession.
- 7.5 The Concessionaire must pay the costs reasonably incurred by the Grantor incidental to any application for consent, whether or not such consent is granted.
- 7.6 If the Concessionaire is not a publicly listed company then any change in the shareholding of the Concessionaire altering the effective control of the Concessionaire requires the approval of the Grantor.

8. What are the obligations to protect the environment?

8.1 The Concessionaire must not cut down or damage any vegetation; or damage any natural feature or historic resource on the Land; or light any fire on the

Land without the prior consent of the Grantor.

- 8.2 The Concessionaire must at its cost keep the Land in a clean and tidy condition and free of weeds and all organisms specified as pests in a relevant pest management strategy.
- 8.3 The Concessionaire must not store hazardous materials on the Land nor store other materials on the Land where they may obstruct the public or create a nuisance
- 8.4 If directed by the Grantor, the Concessionaire must take all steps necessary to control, or, at the Grantor's option, contribute to the cost of controlling any pest, insect or rodent infestation occurring in or emanating from the Land or any structure or facility on the Land, and if directed by the Grantor, engage a pest exterminator approved by the Grantor.
- 8.5 The Concessionaire must not bury:
 - (a) any toilet waste within 50 metres of a water source on the Land; or
 - (b) any animal or fish or any part thereof within 50 metres of any water body, water source or public road or track.

9. What about Environmental Monitoring?

- 9.1 The Concessionaire must, during the Term, if the Grantor so directs, design in consultation with the Grantor and undertake a programme to monitor and report on the environmental effects of the Concessionaire's use of the Land and conduct of the Concession Activity.
- 9.2 If the Grantor does not issue a direction under clause 9.1 the Concessionaire must, during the Term, pay to the Grantor the annual Environmental Monitoring Contribution specified in Item 6 of Schedule 1 to enable the Grantor to design and undertake a programme to monitor the environmental effects of the Concessionaire's use of the Land and conduct of the Concession Activity.

10. When can new structures be erected or land alterations occur?

10.1 The Concessionaire must not erect, alter or bring on to the Land any structure.

11. What about advertising?

11.1 The Concessionaire must not erect or display any signs or advertising on the Land.

12. What are the liabilities and who insures?

- 12.1 The Concessionaire agrees to use the Land at the Concessionaire's own risk and releases to the full extent permitted by law the Grantor and the Grantor's employees and agents from all claims and demands of any kind and from all liability which may arise in respect of any accident, damage or injury occurring to any person or property in or about the Land.
- 12.2 The Concessionaire must indemnify the Grantor against all claims, actions, losses and expenses of any nature which the Grantor may suffer or incur or for which the Grantor may become liable arising from the Concessionaire's performance of the Concession Activity.

- 12.3 This indemnity is to continue after the expiry or termination of this Concession in respect of any acts or omissions occurring or arising before its expiry or termination.
- 12.4 The Concessionaire has no responsibility or liability for costs, loss, or damage of whatsoever nature arising from any act or omission or lack of performance or any negligent or fraudulent act or omission by the Grantor, or any contractor or supplier to the Grantor, or any employee or agent of the Grantor.
- 12.5 Despite anything else in clause 12 the Concessionaire is not liable for any indirect or consequential damage or loss howsoever caused.
- 12.6 The Grantor is not liable and does not accept any responsibility for damage to or interference with the Land , the Concession Activity, or to any structures, equipment or facilities on the Land or any other indirect or consequential damage or loss due to any natural disaster, vandalism, sabotage, fire, or exposure to the elements except where, subject to clause 12.7, such damage or interference is caused by any wilful act or omission of the Grantor, the Grantor's employees, agents or contractors.
- 12.7 Where the Grantor is found to be liable in accordance with clause 12.6, the total extent of the Grantor's liability is limited to \$1,000,000 in respect of the Concessionaire's structures, equipment and facilities.
- 12.8 Despite anything else in clause 12 the Grantor is not liable for any indirect or consequential damage or loss howsoever caused.
- 12.9 Without prejudice to or in any way limiting its liability under this clause 12 the Concessionaire at the Concessionaire's expense must take out and keep current policies for insurance and for the amounts not less than the sums specified in Item 13 of Schedule 1 with a substantial and reputable insurer.
- 12.10 After every three year period of the Term the Grantor may, on giving 10 working day's notice to the Concessionaire, alter the amounts of insurance required under clause 12.9. On receiving such notice the Concessionaire must within 10 working days take out and keep current policies for insurance and for the amounts not less than the sums specified in that notice.
- 12.11 The Concessionaire must provide to the Grantor within 5 working days of the Grantor so requesting:
 - (a) details of any insurance policies required to be obtained under this Concession, including any renewal policies if such renewal occurs during the Term; and/ or;
 - (b) a copy of the current certificate of such policies.

13. What about Health and Safety?

13.1 The Concessionaire must exercise the rights granted by this Concession in a safe and reliable manner and must comply with the Health and Safety in Employment Act 1992 and its regulations and all other provisions or requirements of any competent authority relating to the exercise of this Concession. The Concessionaire must comply with its safety plan (if one is required in Item 11 of Schedule 1), and with any safety directions of the Grantor.

- 13.2 Before commencing the Concession Activity the Concessionaire must, where the Concessionaire has Qualmark or Outdoorsmark certification, provide the Grantor with a copy of that certification.
- 13.3 If the Concessionaire does not hold Qualmark or Outdoorsmark certification then before commencing the Concession Activity the Concessionaire must, if required by Item 11 of Schedule 1:
 - (a) prepare a safety plan;
 - (b) have it audited by a suitably qualified person approved by the Grantor; and forward to the Grantor a certificate from the auditor certifying that the safety plan is suitable for the Concession Activity; and
 - (c) the Concessionaire must obtain from the auditor details as to when the safety plan is to be re-audited. The Concessionaire must comply with any such requirement to re-audit and forward a copy of the reaudit certificate to the Grantor within 5 working days of the certificate being issued.
- 13.4 For any Concession Activity that is subject to the Health and Safety in Employment (Adventure Activities) Regulations 2011, proof of registration with WorkSafe New Zealand will satisfy the Grantor's requirement under clause 13.3(b).
- 13.4 The Grantor may at any time request the Concessionaire to provide the Grantor with a copy of the current safety plan in which case the Concessionaire must provide the copy within 10 working days of receiving the request.
- 13.5 Receipt of the safety plan/audit certificate by the Grantor is not in any way to limit the obligations of the Concessionaire under clause 13 and is not to be construed as implying any responsibility or liability on the part of the Grantor.
- 13.6 The Concessionaire must:
 - (a) notify the Grantor of any natural events or activities on the Land or the surrounding area which may endanger the public or the environment;
 - (b) take all practicable steps to protect the safety of all persons present on the Land and must, where necessary, erect signposts warning the public of any dangers they may encounter as a result of the Concessionaire's operations;
 - (c) take all practicable steps to eliminate any dangers to the public and must clearly and permanently mark any that remain and of which the Concessionaire is aware;
 - (d) record and report to the Grantor all accidents involving serious harm within 24 hours of their occurrence and forward an investigation report within 3 days of the accident occurring;
 - (e) at the request of the Grantor make available any of the Concessionaire's directors, employees, servants or agents who in the opinion of the Grantor might assist any investigation by the Grantor into the cause of any such serious harm accident or incident;

- (f) ensure that all contracts between the Concessionaire and any contractors contain, at a minimum, the same requirements as clause 13;
- (g) be satisfied that facilities or equipment provided by the Grantor to enable the Concession Activity to be carried out meet the safety requirements of the Concessionaire;
- (h) not bring onto the Land or any land administered by the Department any dangerous or hazardous material or equipment which is not required for purposes of the Concession Activity; and if such material or equipment is required as part of the Concession Activity, the Concessionaire must take all practicable steps at all times to ensure that the material or equipment is treated with due and proper care.

14. What are the compliance obligations of the Concessionaire?

- 14.1 The Concessionaire must comply where relevant:
 - (a) with the provisions of any conservation management strategy or conservation management plan under the Conservation Act 1987 or Part IIA of the Reserves Act 1977, or any general policy statement made under the Conservation Act 1987, Reserves Act 1977, National Parks Act 1980, or Wildlife Act 1953, or management plan under section 45 of the National Parks Act 1980, whichever is appropriate to the Land, together with any amendment or review of any policy, strategy or plan whether approved before, on, or after the date on which this Concession takes effect; and
 - (b) with the Conservation Act 1987, the Reserves Act 1977, the National Parks Act 1980, Wildlife Act 1953 and any other statute, ordinance, regulation, bylaw, or other enactment (collectively the "Legislation") affecting or relating to the Land or affecting or relating to the Concession Activity, including any regulations made under the Conservation Act 1987 and Wildlife Act 1953 or bylaws made under the Reserves Act 1977 or the National Parks Act 1980; and
 - (c) with all notices and requisitions of any competent authority affecting or relating to the Land or affecting or relating to the conduct of the Concession Activity; and
 - (d) with all Department signs and notices placed on or affecting the Land; and
 - (e) with all reasonable notices and directions of the Grantor concerning the Concession Activity on the Land.
- 14.2 The Concessionaire must comply with this Concession.
- 14.3 A breach or contravention by the Concessionaire of a relevant conservation management strategy, conservation management plan, management plan or any statement of general policy referred to in clause 14.1(a) is deemed to be a breach of this Concession.
- 14.4 A breach or contravention by the Concessionaire of any Legislation affecting or relating to the Land or affecting or relating to the Concession Activity is deemed to be a breach of this Concession.

15. What are the Grantor's rights to remedy defaults?

- 15.1 The Grantor may elect to remedy at any time, after giving notice, if practicable, any default by the Concessionaire under this Concession. Before electing to so remedy in accordance with this clause, the Grantor must, if practicable, first give the Concessionaire notice of the default and a reasonable opportunity to remedy the default.
- 15.2 The Concessionaire must pay to the Grantor forthwith on demand all reasonable costs and expenses incurred by the Grantor, including legal costs and expenses as between solicitor and client, in remedying such default. The Concessionaire is to pay interest on such costs and expenses if payment is not made within 14 day's of the Grantor's demand from the date of the demand until the date of payment at the Penalty Interest Rate specified in Item 12 of Schedule 1.

16. When can the Concession be suspended?

- 16.1 If, in the Grantor's opinion, there is a temporary risk to any natural or historic resource on or in the vicinity of the Land or to public safety whether arising from natural events such as earthquake, land slip, volcanic activity, flood, or arising in any other way, whether or not from any breach of the terms of this Concession on the part of the Concessionaire, then the Grantor may suspend this Concession.
- 16.2 If, in the Grantor's opinion, the activities of the Concessionaire are having or may have an adverse effect on the natural, historic or cultural values or resources of the Land and the Grantor considers that the effect can be avoided, remedied or mitigated to an extent satisfactory to the Grantor, then the Grantor may suspend this Concession until the Concessionaire avoids, remedies or mitigates the adverse effect to the Grantor's satisfaction.
- 16.3 The Grantor may suspend the Concession for such period as the Grantor determines where the Concessionaire has breached any terms of this Concession.
- 16.4 The Grantor may suspend this Concession while the Grantor investigates any of the circumstances contemplated in clauses 16.1 and 16.2 and also while the Grantor investigates any potential breach or possible offence by the Concessionaire, whether or not related to the Concession Activity under the Conservation Act 1987 or any of the Acts mentioned in the First Schedule of that Act.
- 16.5 The word "investigates" in clause 16.4 includes the laying of charges and awaiting the decision of the Court.
- 16.6 During any period of temporary suspension arising under clauses 16.1 or 16.2 the Concession Fee payable by the Concessionaire is to abate in fair proportion to the loss of use by the Concessionaire of the Land.
- 16.7 The Grantor is not to be liable to the Concessionaire for any loss sustained by the Concessionaire by reason of the suspension of the Concession under this clause 16 including loss of profits.

17. When can the Concession be terminated?

- 17.1 The Grantor may terminate this Concession either in whole or in part:
 - (a) by 14 days notice to the Concessionaire if the Concession Fee or any other money payable to the Grantor under this Concession is in

arrears and unpaid for 14 days after any of the days appointed for payment whether it has been lawfully demanded or not; or

- (b) by 14 days notice to the Concessionaire or such sooner period as it appears necessary and reasonable to the Grantor if.
 - (i) the Concessionaire breaches any terms of this Concession and in the Grantor's sole opinion the breach is able to be rectified; and
 - (ii) the Grantor has notified the Concessionaire of the breach; and
 - (iii) the Concessionaire does not rectify the breach within 7 days of receiving notification; or such earlier time as specified by the Grantor; or
- (c) by notice in writing to the Concessionaire where the Concessionaire breaches any terms of this Concession and in the sole opinion of the Grantor the breach is not capable of being rectified; or
- (d) immediately by notice in writing to the Concessionaire where the Concessionaire breaches clauses 12.9 and 13; or
- (e) by notice in writing to the Concessionaire if the Concessionaire ceases to conduct the Concession Activity or, in the reasonable opinion of the Grantor, the conduct of the Concession Activity is manifestly inadequate; or
- (f) by notice in writing to the Concessionaire if the Concessionaire is convicted of an offence under the Conservation Act 1987 or any of the Acts listed in the First Schedule to that Act or any statute, ordinance, regulation, bylaw, or other enactment affecting or relating to the Land or which in the Grantor's sole opinion affects or relates to the Concession Activity; or
- (g) by notice in writing to the Concessionaire if the Concessionaire or the Guarantor is dissolved; or enters into any composition with or assignment for the benefit of its creditors; or is adjudged bankrupt; or being a company, has a receiver appointed; or is put into liquidation; or is placed under statutory management; or has a petition for winding up presented against it; or is otherwise unable to pay its debts as they fall due; or the estate or interest of the Concessionaire is made subject to a Writ of Sale or charging order; or the Concessionaire ceases to function or operate; or
- (h) immediately if there is, in the opinion of the Grantor, a permanent risk to public safety or to the natural and historic resources of the Land whether arising from the conduct of the Concession Activity or from natural causes such as earthquake, land slip, volcanic activity, flood, or arising in any other way, whether or not from any breach of the terms of this Concession on the part of the Concessionaire.
- 17.2 The Grantor may exercise its power to terminate under 17.1(h) without giving notice.
- 17.3 The Grantor may exercise the Grantor's right under this clause to terminate the Concession notwithstanding any prior waiver or failure to take action by the Grantor or any indulgence granted by the Grantor for any matter or default.

17.4 Termination of the Concession is not to prejudice or affect the accrued rights or claims and liabilities of the parties.

18. What happens on termination or expiry of the Concession?

- 18.1 On expiry or termination of this Concession, either as to all or part of the Land, the Concessionaire is not entitled to compensation for any structures or other improvements placed or carried out by the Concessionaire on the Land.
- 18.2 The Concessionaire may, with the Grantor's written consent, remove any specified structures and other improvements on the Land. Removal under this clause must occur within the time specified by the Grantor and the Concessionaire is to make good any damage and leave the Land and other public conservation land affected by the removal in a clean and tidy condition.
- 18.3 The Concessionaire must, if the Grantor gives written notice, remove any specified structures and other improvements on the Land. Removal under this clause must occur within the time specified by the Grantor and the Concessionaire is to make good any damage and leave the Land and other public conservation land affected by the removal in a clean and tidy condition and replant the Land with indigenous vegetation of a similar abundance and diversity as at the commencement of the Term. If hefore the expiry of the Term the Concessionaire makes an application for a further concession in respect of the same Concession Activity on the Land then the Grantor can not require such removal and reinstatement until such time as that concession application has been determined. If a new concession is granted then removal and reinstatement can not be required until the expiry or termination of the new concession.

19. When is the Grantor's consent required?

19.1 Where the Grantor's consent or approval is expressly required under this Concession then the Concessionaire must seek that approval or consent for each separate time it is required even though the Grantor may have given approval or consent for a like purpose on a prior occasion. Any such consent or approval may be made on such conditions as the Grantor considers appropriate.

20. Are there limitations on public access and closure?

20.1 The Concessionaire acknowledges that the Land is open to the public for access and that the Grantor may close public access during periods of high fire hazard or for reasons of public safety or emergency.

21. What about other concessions?

21.1 Nothing expressed or implied in this Concession is to be construed as preventing the Grantor from granting other concessions, whether similar or not, to other persons provided that the Grantor must not grant another concession that would derogate in any material way from the Concessionaire's ability to carry out the Concession Activity.

22. How will disputes be resolved?

22.1 If a dispute arises between the parties in connection with this Concession the parties must, without prejudice to any other rights or entitlements they may have, attempt to resolve the dispute by agreement using informal dispute resolution techniques such as negotiation, mediation, independent expert appraisal or any other alternative dispute resolution technique. The rules governing any such technique adopted are to be agreed between the parties.

- 22.2 If the dispute cannot be resolved by agreement within 14 days of written notice by one party to the other (or such further period as the parties may agree to in writing) either party may refer the dispute to the Disputes Tribunal, where relevant, or to arbitration, which arbitration is to be carried out in accordance with the provisions of the Arbitration Act 1996.
- 22.3 If the parties do not agree on an arbitrator within 10 working days of a party giving written notice of the requirement to appoint an arbitrator the President of the New Zealand Law Society is to appoint the arbitrator. In either case the arbitrator must not be a person who has participated in an informal dispute resolution procedure in respect of the dispute.
- 22.4 The arbitrator must include in the arbitration award reasons for the determination.
- 22.5 Despite the existence of a dispute, each party must continue to perform its obligations under this Concession.

23. What about prosecution for offences?

- 23.1 Where any breach of this Concession by the Concessionaire also constitutes an offence under the Resource Management Act 1991, the Conservation Act 1987, or any of the Acts listed in the First Schedule to that Act:
 - (a) no waiver or failure to act by the Grantor under this Concession is to preclude the Grantor from prosecuting the Concessionaire; and
 - (b) no failure by the Grantor to prosecute the Concessionaire is to preclude the Grantor from exercising the Grantor's remedies under this Concession; and
 - (c) any action of the Grantor in prosecuting the Concessionaire is not to preclude the Grantor from exercising the Grantor's remedies under this Concession.

24. How are notices sent and when are they received?

- 24.1 Any notice to be given under this Concession is to be in writing and made by personal delivery, fax, by pre paid post or email to the receiving party at the address, fax number or email address specified in Item 14 or 15 of Schedule 1. Any such notice is to be deemed to have been received:
 - (a) in the case of personal delivery, on the date of delivery;
 - (b) in the case of post, on the 3rd working day after posting;
 - (c) in the case of email, on the day the email was sent provided it is sent before 5 pm on a working day, otherwise on the next working day.
- 24.2 If any party's details specified in Item 14 or 15 of Schedule 1 change then the party whose details change must within 5 working days of such change provide the other party with the changed details.

25. What is the scope of the Concession?

25.1 Except as provided by legislation, this Concession and any written variation agreed by the parties contain the entire understanding between the parties

with reference to the subject matter of this Concession and there is no other agreement, representation or warranty whether it is expressed or implied which in any way extends, defines or otherwise relates to the provisions of this Concession.

26. Can provisions be severed?

26.1 Any illegality, or invalidity or unenforceability of any provision in this Concession is not to affect the legality, validity or enforceability of any other provisions.

27. What about the payment of costs?

- 27.1 The Concessionaire must pay the Grantor's legal costs and expenses of and incidental to preparing and signing any extension or variation of this Concession.
- 27.2 The Concessionaire must pay in full immediately and on demand all costs and fees (including solicitor's costs and fees of debt collecting agencies engaged by the Grantor) arising out of and associated with steps taken by the Grantor to enforce or attempt to enforce the Grantor's rights and powers under this Concession including the right to recover outstanding money owed to the Grantor.

28. What is the relationship of parties?

- 28.1 Nothing expressed or implied in this Concession is to be construed as:
 - (a) constituting the parties as partners or joint venturers;
 - (b) conferring on the Concessionaire any right of exclusive occupation or use of the Land;
 - (c) granting any exclusive estate or interest in the Land to the Concessionaire;
 - (d) affecting the rights of the Grantor and the public to have access across the Land.

29. What about a Guarantee?

- 29.1 Where the Grantor has in Item 15 of Schedule 1 required this Concession to be guaranteed by a third party the following clauses are to apply.
- 29.2 In consideration of the Grantor entering into this Concession at the Guarantor's request the Guarantor:
 - (a) guarantees payment of the Concession Fee and the performance by the Concessionaire of the covenants in this Concession; and
 - (b) indemnifies the Grantor against any loss the Grantor might suffer should the Concession be lawfully disclaimed or abandoned by any liquidator, receiver or other persons.
- 29.3 The Guarantor covenants with the Grantor that:
 - (a) no release, delay, or other indulgence given by the Grantor to the Concessionaire, to the Concessionaire's successors or assigns, or any other thing whereby the Guarantor would have been released had the Guarantor been merely a surety is to release, prejudice, or affect the liability of the Guarantor as a Guarantor or as indemnifier;

- (b) as between the Guarantor and Grantor the Guarantor may, for all purposes, be treated as the Concessionaire and the Grantor is under no obligation to take proceedings against the Concessionaire before taking proceedings against the Guarantor;
- (c) the guarantee is for the benefit of and may be enforced by any person entitled for the time being to receive the Concession Fee;
- (d) any assignment of this Concession and any Concession Fee Review in accordance with this Concession are not to release the Guarantor from liability;
- (e) should there be more than one Guarantor the liability of each Guarantor under this Guarantee is to be joint and several.

30. What about Co-Siting? [clause deleted – not relevant]

31. What about Identification cards?

- 31.1 Before commencing the Concession Activity the Concessionaire must, if required by the Grantor in Item 12 of Schedule 1, obtain Concessionaire Identification cards from the Grantor. The Grantor is to supply such cards to the Concessionaire on a cost recovery basis.
- 31.2 The Concessionaire and any person acting under the authority of the Concession must carry and display a Concession Identification card when carrying out the Concession Activity.
- 31.3 The Concessionaire must obtain sufficient cards to ensure all people acting under the authority of the Concession can carry and display such cards when undertaking the Concession Activity.
- 31.4 The Concessionaire may also access, use and/ or display the Grantor's "Approved Label". This right only exists once the Concessionaire agrees to comply with the Grantor's Approved Label terms and conditions and while the Concession remains operative. When the Concessionaire so requests the Grantor is to forward the Concessionaire an electronic link to the Approved Label. This electronic link is to contain the Approved Label terms and conditions.
- 31.5 The right under this clause 31.4 does not affect the obligation in this clause 31 to carry and display a Concession Identification card.

32. Which clauses survive termination?

32.1 Clauses 12 and 24 survive the termination of this Concession.

33. When can the conditions of the Concession be varied?

- 33.1 The Grantor may on each Concession Fee Review Date, after first consulting with the Concessionaire, vary any condition of this Concession to make the condition more effective in addressing adverse effects resulting from the Concession Activity.
- 33.2 Nothing in clause 33.1 otherwise affects the Grantor's rights to vary the Concession under section 17ZC of the Conservation Act 1987.

34. Are there any Special Conditions?

34.1 Special conditions are specified in Schedule 3.

35. The Law

35.1 This Concession is to be governed by, and interpreted in accordance with the laws of New Zealand

SPECIAL CONDITIONS

In respect to Ngāi Tahu

- 1. The Concessionaire is requested to consult the relevant Papatipu Runanga (www.ngaitahu.iwi.nz) if they wish to use Ngāi Tahu cultural information. If the Concessionaire wishes to use the Tōpuni or statutory acknowledgement information contained in schedules 14-108 of the Ngāi Tahu Claims Settlement Act 1998, or any Department produced interpretative material in respect to Ngāi Tahu cultural information, they are requested to notify the relevant Papatipu Runanga, as a matter of courtesy.
- 2. The Concessionaire must, as far as practicable, attend any workshops held by the Department for the purpose of providing information to concessionaires, which is to include the Ngāi Tahu values associated with Tōpuni areas.
- 3. The Concessionaire must ensure any persons employed by the Concessionaire are requested to recognise and provide for Ngāi Tahu values in the conduct of their activities.

Didymo

4. The Concessionaire must comply with all guidelines and notices put out by Biosecurity New Zealand regarding measures to avoid spreading the pest organism Didymosphenia geminate ("Didymo"), and or any other pest organism identified during the term of this Concession.

Aircraft Use

5. The activity is limited to the following;

Location	Activity	Max Party size (Incl Guides)	Runs per day per trip	Landings per day per trip per party	Max trips per year
	Heli-skiing, Heliboarding	(to be split into groups of no more than 5)	5	10	

- 6. The Concessionaire shall ensure that all helicopters keep at least 500 feet from ground parties encountered during flying and landing operations.
- 7. The licence shall be operable only in conjunction with Civil Aviation Authority, Resource Management Act consent, and any other approvals necessary for such operation. The Concessionaire, or it's chosen operator, must possess a current Civil Aviation Authority Air Operator Certificate, or Transitional Air Operator Certificate (Air Service Operations), as appropriate, as well as appropriate airworthiness certificates for all aircraft used, pursuant to Clause 13 of this concession. Before commencing this activity, copies of these certifications must be submitted to the Grantor.

- 8. The Concessionaire shall be permitted to land only if the Land is clear of other users including recreationists.
- 9. The Concessionaire shall complete the Client Activity Return forms, yearly from the commencement date of the concession, in each and every year and must contain the following information
 - i. The number of landings/sites visited undertaken on public conservation lands
 - ii. The sites at which landings/sites visited occurred
 - iii. The number of clients landed/sites visited and the reason for landing or visit
 - iv. The days on which the landings/sites visited occurred.
 - v. In the case of Heliskiing a map showing the runs that were under taken.
- 10. The Concessionaire must adhere to the Helicopter Association International 'Fly Neighborly Guide'.
- 11. The Concessionaire must not refuel, leave any fuel drums or construct any fuel dumps on the Conservation Area.
- 12. The Concessionaire must ensure that aircraft idle times on the ground are kept to a practicable minimum.
- 13. During the term of the concession, where Grantor believes that the effects of aircraft noise should be further reduced, the Grantor may, by notice, require the concessionaire to either undertake measures to minimise the effects of noise on conservation values or become accredited to a recognised noise abatement and disputes resolution programme. If such notice is given by the Grantor, the concessionaire must

(i) if required to undertake measures to minimise the effects of noise on conservation values within 3 months from receiving the notice undertake those measures to the satisfaction of the Grantor until the Final Expiry Date.

(ii) if required to become accredited to a recognised noise abatement and disputes resolution programme within 3 months from receiving the notice provide proof to the Grantor that such accreditation has been completed and must keep their participation in that programme or training current until the Final Expiry Date.

The Grantor may, at any time, issue a subsequent notice(s) requiring the Concessionaire to implement the other option.

Other Special Conditions

- 14. Provided that the Concessionaire for this concession being Verity Farms Limited is also the Concessionaire for the Guiding Licence concession being ______ [insert reference], then the concession Annual Management Fee in Schedule 1, item 6 is not payable. This is on the basis that while the two concessions are both held by Verity Limited, one Annual Management Fee is sufficient.
- 15. This Licence authorises the Concessionaire to carry out the Concession Activity only, but does not authorise the Concessionaire to do anything which

would not otherwise be permitted or which would require some other form of authority from the Grantor or any other person. Without limiting the effect of this clause and by way of example the Concessionaire is not authorised under this Concession to:

- i. Access any part of the Land by air unless such access is with an air operator who is otherwise authorised to fly into the area concerned.
- ii. Breach any law, regulation, by law, restriction, requirement, notice, sign or provision of any Conservation Management Strategy or management plan.
- iii. Access any Land or water other than that listed in Schedule 1.
- 16. The Grantor reserves the right to request further or different activity related information in order to best monitor and determine any effects of the Concession Activity on the Land.

Designations Plan or map

Any further plan if required

Godley Peaks Pastoral Lease - Heli-ski concession - May 2015 - DOCDM-1599937 Previous template docdm-1406816

- 24 -

					<u> </u>		 <u> </u>	 	-			
	_											
										_		
21 22												

Appendix 7: Concession for commercial filming and photography (May 2015 (1))



Department of Conservation Te Papa Atawhai

Concession Number:

Concession Document (Guiding Licence)

THIS CONCESSION is made this day of

PARTIES:

- 1. Minister of Conservation (the Grantor)
- 2. Verity Farms Limited (the Concessionaire)

BACKGROUND

- **A.** The Department of Conservation ("Department") Te Papa Atawhai is responsible for managing and promoting conservation of the natural and historic heritage of New Zealand on behalf of, and for the benefit of, present and future New Zealanders.
- **B.** The land described in Item 1 of Schedule 1 as the Land is a Conservation Area or a Reserve under the management of the Grantor.
- **C.** Sections 66 and 68 of the Crown Pastoral Land Act 1998 authorise the Grantor to grant a Concession for a Concession Activity in a Conservation Area and a Reserve under section 17Q(1) of the Conservation Act 1987 or section 59A of the Reserves Act 1977 (whichever is relevant in the circumstances).
- **D.** The Concessionaire wishes to carry out the Concession Activity on the Land subject to the terms and conditions of this Concession.
- **E.** The parties wish to record the terms and conditions of this Concession and its Schedules.

OPERATIVE PARTS

1. In exercise of the Grantor's powers under the Crown Pastoral Land Act 1998, and the Conservation Act 1987 or the Reserves Act 1977 as relevant, the Grantor **GRANTS** to the Concessionaire a **GUIDING LICENCE** to carry out the Concession Activity on the Land subject to the terms and conditions contained in this Concession and its Schedules.

SIGNED on behalf of the Minister of Conservation by [insert name and title of delegate]	
acting under delegated authority in the presence of:	in the presence of:
Witness Signature:	Witness Signature: Witness Name:
Witness Name:	Witness Occupation:
Witness Occupation:	Witness Address:
Witness Address:	
A copy of the Instrument of Delegation may be inspected at the Director-General's office at 18-22 Manners Street, Wellington.	

Godley Peaks Pastoral Lease - filming concession - May 2015 (1) Previous template docdm-1406816

1.	Land (Schedule 3)	As shown on the Proposed Designations Plan marked (CA1) attached in Schedule 4
	(cenedule 3)	Conservation Area of 10,629 hectares approximately on the Hall Range and Mistake Peak situated in the Land District of Canterbury
2.	Concession Activity	Commercial Filming and Photography
	(clause 2)	The use of the land for the purposes of commercial filming and photography
3.	Term (clause 3)	10 years commencing on the date that an approved plan is registered, affecting Certificate of Title CB 30B/914 (Canterbury Registry), vesting the land in the Crown as a conservation area or reserve, being the day of 20 (the commencement date)
4.	Renewal(s) (clause 3)	None
5.	Final Expiry Date (clause 3)	The 10th anniversary of the commencement date.
6.	Concession Fee	Annual Activity Fees:
	(clause 4)	\$500 + GST per day for commercials, feature films and dramas;
		\$300 + GST per day for documentaries, sports events and still photography;
		\$100 + GST per day for filming schools/ students and conservation/recreation promotion;
		PLUS \$25 + GST per person in the film crew per day for all of the above categories <u>except</u> for filming schools/ students."
		Annual Management Fee:
		\$500.00 per annum plus GST, payment of this fee is subject to special condition 34 in Schedule 3.
		Annual Environmental Monitoring Fee
		To be negotiated between Grantor and Concessionaire if required (see special condition 35)
7.	a. Activity Returns Dates b. Concession Fee Payment	a. Annually, in arrears upon commencement of the term set out in clause 3 for payment instalments.
	Instalments	b. Annually, in arrears upon commencement of the term set out in clause 3 for activity returns, using the form attached in
	(clauses 4 and 6.1)	Schedule 5.
8.	Concession Fee Payment Date(s) (clause 4)	On or before the date specified on the invoice issued by the Grantor.
9.	Penalty Interest Rate	Double the current Official Cash Rate (OCR).
-	(clause 4)	See Reserve Bank of New Zealand website

Concessionaire's initials Grantor's initials	Concessionaire's initials		Grantor's initials	
--	---------------------------	--	--------------------	--

Godley Peaks Pastoral Lease - filming concession - May 2015 (1)Licence Previous template docdm-1406816

10.	Concession Fee Review Date(s) (clause 5)	On the third anniversary of the date of commencement of this Concession set out in item 3 above and the corresponding date every 3 years thereafter until the expiry of the term.
11.	Health and Safety (clause 12)	Audited Safety Plan: Required (unless Qualmark or Outdoorsmark certified)Auditors certificate of approval to be provided to Grantor
12.	Concessionaire Identification (Clause 23)	Not Required
13.	Insurance (To be obtained by Concessionaire) (clause 11)	Types and amounts:Public Liability Insurance for:(a) General indemnity for an amount no less than\$2,000,000.00; and(b) Forest and Rural Fires Act extension for an amount noless than \$2,000,000.00; and(c) Statutory liability for an amount no less than\$500,000.00Subject to review on each Concession Fee Review Date
14.	Addresses for Notices (clause 21)	The Grantor's address is: Physical Address: Department of Conservation Conservation House 77 Lower Stuart Street Dunedin 9016 Postal Address: PO Box 5244 Moray Place Dunedin 9058
		Phone: (03) 477 0677 Email: permissionsdunedin@doc.govt.nz The Concessionaire's address in New Zealand is:
		Address: C/- Milnes Beatson Ld 29 Wallace Street Motueka 7120 Phone: (03) 528 8760 Email: david@mba.net.nz
		Director's Address: 210 Patons Rock Road RD2 Takaka 7182 Phone: 021 531 227

Concessionaire's initials

Grantor's initials

Godley Peaks Pastoral Lease - filming concession - May 2015 (1)Licence Previous template docdm-1406816

		Email: ann@verityfarms.co.nz
15.	Special Conditions (clause 25)	See Schedule 3.

Note: The clause references are to the Minister of Conservation's Standard Terms and Conditions for Guiding Licences set out in Schedule 2.

Note: Please initial each page of Schedule 1

Godley Peaks Pastoral Lease - filming concession - M	lay 2015 (1)Licence	
Previous template docdm-1406816			

Concessionaire's initials

Grantor's initials

SCHEDULE 2

STANDARD TERMS AND CONDITIONS FOR GUIDING LICENCES

1. Interpretation

1.1 **"Background"** means the matters referred to under the heading 'Background" on the first page of this Document, and words used in the Background have the meaning given to them in this clause 1.1.

"Commencement date" means the date that an approved plan is registered vesting the Land in the Crown as a conservation area or reserve.

"Concession" means this document and any subsequent amendments and all schedules, annexures, and plans attached to it.

"Concession Activity" means the activity described in item 2 of Schedule 1.

"Conservation Area" has the same meaning as "conservation area" in section 2 of the Conservation Act 1987.

"Director-General" means the Director-General of Conservation

"**Reserve**" has the same meaning as "reserve" in section 59A of the Reserves Act 1977.

"Term" means the period of time specified in Item 5 of Schedule 1 during which this Concession operates.

1.2 The Concessionaire is responsible for the acts and omissions of its employees, contractors, agents, clients and invitees (excluding other members of the public accessing the Land). The Concessionaire is liable under this Concession for any breach of the terms of the Concession by its employees, contractors, agents, clients and invitees (excluding other members of the public accessing the Land), as if the breach had been committed by the Concessionaire.

2. What is being authorised?

2.1 The Concessionaire is only allowed to use the Land for the Concession Activity.

3. How long is the Concession for - the Term?

- 3.1 This Concession commences on the date set out in Item 3 of Schedule 1 and ends on the Final Expiry Date specified in Item 5 of Schedule 1.
- 3.2 No renewals of this Concession are permitted.

4. What are the fees and when are they to be paid?

4.1 The Concessionaire must pay to the Grantor in the manner directed by the Grantor the Concession Fee (which includes the Annual Activity Fees, the Management Fee, and the Environmental Monitoring Fee) plus GST in the

instalments and on the Concession Fee Payment Dates specified in Items 6, 7 and 8 of Schedule 1.

4.2 If the Concessionaire fails to make payment within 14 days of the Concession Fee Payment Date then the Concessionaire is to pay interest on the unpaid Concession Fee from the Concession Fee Payment Date until the date of payment at the Penalty Interest Rate specified in Item 9 of Schedule 1.

5. When can the fees be reviewed?

- 5.1 The Grantor is to review the Concession Fee on the Concession Fee Review Date stated in Item 10 of Schedule 1. The new Concession Fee is to be the market value of the Concession Activity carried out on the Land having regard to the matters set out in section 17Y(2) of the Conservation Act 1987.
- 5.2 Both parties are to agree on the new fee within 30 working days of the Grantor giving the Concessionaire written notice of the review.
- 5.3 If the parties cannot so agree then each party is to appoint a Registered Valuer who must meet and agree on the new fee. If the Registered Valuers fail to reach agreement the new fee is to be determined by an umpire appointed by the two Registered Valuers. Each party is to bear that party's own costs and half the costs of the umpire (if any).

6. What about Activity return forms?

6.1 The Concessionaire must complete a Client Activity Return form in the format required by the Grantor, and return them to the Grantor on the Activity Return Dates stated in Item 7 of Schedule 1. The Grantor may request further or different activity related information to better monitor and determine any effects of the Concession Activity on the Land.

7. When can the Concession be assigned?

- 7.1 The Concessionaire must not transfer, sub licence, assign, mortgage or otherwise dispose of the Concessionaire's interest under this Concession or any part of it (which includes the Concessionaire entering into a contract or any other arrangement whatsoever whereby the Concession Activity would be carried out by a person (called the assignee) other than the Concessionaire) without the prior written consent of the Grantor.
- 7.2 The Grantor may in the Grantor's discretion decline any application for consent under clause 7.1.
- 7.3 Sections 17P, 17S, 17T, 17U, 17W, 17X, 17ZB and 17ZC of the Conservation Act 1987 apply to applications for consent under this clause unless the Grantor, in the Grantor's discretion, decides otherwise.
- 7.4 If the Grantor gives consent under this clause then the Concessionaire remains liable to observe and perform the terms and conditions of this Concession throughout the Term and is to procure from the Assignee a covenant to be bound by the terms and conditions of this Concession.

- 7.5 The Concessionaire must pay the costs reasonably incurred by the Grantor incidental to any application for consent, whether or not such consent is granted.
- 7.6 Any change in the shareholding of the Concessionaire altering the effective control of the Concessionaire is to be deemed to be an assignment and requires the consent of the Grantor.

8. What are the obligations to protect the environment?

- 8.1 The Concessionaire must not cut down or damage any vegetation; or damage any natural feature or historic resource on the Land; or light any fire on the Land without the prior consent of the Grantor.
- 8.2 The Concessionaire must ensure that it adheres to the international "Leave No Trace" Principles at all times (<u>www.leavenotrace.org.nz</u>).
- 8.3 The Concessionaire must not bury:
 - (a) any toilet waste within 50 metres of a water source on the Land; or
 - (b) any animal or fish or any part thereof within 50 metres of any water body, water source or public road or track.

9. When can structures be erected?

- 9.1 The Concessionaire must not place any structures on the Land without the prior written consent of the Grantor.
- 9.2 The Concessionaire must keep all structures, buildings, fences, gates, drains and other improvements now or hereafter upon the Land, in good order, condition and repair.

10. What if the Concessionaire wishes to surrender the Concession?

10.1 If the Concessionaire wishes to surrender this Concession during the currency of the Term, then the Grantor may accept that surrender on such conditions as the Grantor considers appropriate.

11. What are the liabilities and who insures?

- 11.1 The Concessionaire agrees to use the Land at the Concessionaire's own risk and releases to the full extent permitted by law the Grantor and the Grantor's employees and agents from all claims and demands of any kind and from all liability which may arise in respect of any accident, damage or injury occurring to any person or property in or about the Land.
- 11.2 The Concessionaire must indemnify the Grantor against all claims, actions, losses and expenses of any nature which the Grantor may suffer or incur or for which the Grantor may become liable arising from the Concessionaire's performance of the Concession Activity.
- 11.3 This indemnity is to continue after the expiry or termination of this Concession in respect of any acts or omissions occurring or arising before its expiry or termination.
- 11.4 Without prejudice to or in any way limiting its liability under this clause 11 the Concessionaire at the Concessionaire's expense must take out and keep current policies for insurance and for the amounts not less than the sums stated in Item 13 of Schedule 1 with a substantial and reputable insurer.
- 11.5 The Grantor may on each Concession Fee Review Date on giving 10 working day's notice to the Concessionaire alter the amounts of insurance required under clause 11.4. On receiving such notice the Concessionaire must within 10 working days take out and keep current policies for insurance and for the amounts not less than the sums specified in that notice.
- 11.6 The Concessionaire must, provide to the Grantor within 5 working days of the Grantor so requesting:
 - (a) details of any insurance policies required to be obtained under this Concession, including any renewal policies if such renewal occurs during the Term; and/ or;
 - (b) a copy of the current certificate of such policies.

12. What about Health and Safety?

- 12.1 The Concessionaire must exercise the rights granted by this Concession in a safe and reliable manner and must comply with the Health and Safety in Employment Act 1992 and its regulations and all other provisions or requirements of any competent authority relating to the exercise of this Concession. The Concessionaire must comply with its safety plan (if one is required in Item 11 of Schedule 1), and with any safety directions of the Grantor.
- 12.2 Before commencing the Concession Activity the Concessionaire must where the Concessionaire has Qualmark or Outdoorsmark certification provide the Grantor with a copy of that certification.
- 12.3 If the Concessionaire does not hold Qualmark or Outdoorsmark certification then before commencing the Concession Activity the Concessionaire must, if required by Item 11 of Schedule 1:
 - (a) prepare a safety plan;

- (b) have it audited by a suitably qualified person approved by the Grantor; and forward to the Grantor a certificate from the auditor certifying that the safety plan is suitable for the Concession Activity; and
- (c) The Concessionaire must obtain from the auditor details as to when the safety plan is to be re-audited. The Concessionaire must comply with any such requirement to re-audit and forward a copy of the re-audit certificate to the Grantor within 5 working days of the certificate being issued.
- 12.4 The Grantor may at any time request the Concessionaire to provide the Grantor with a copy of the current safety plan in which case the Concessionaire must provide the copy within 10 working days of receiving the request.

13. What are the compliance obligations of the Concessionaire?

- 13.1 The Concessionaire must comply where relevant:
 - (a) with the provisions of any conservation management strategy or conservation management plan under the Conservation Act 1987 or Part 2A of the Reserves Act 1977, or any general policy statement made under the Conservation Act 1987, Reserves Act 1977, National Parks Act 1980, or Wildlife Act 1953, or management plan under section 45 of the National Parks Act 1980, whichever is appropriate to the Land, together with any amendment or review of any policy, strategy or plan whether approved before, on, or after the date on which this Concession takes effect; and
 - (b) with the Conservation Act 1987, the Reserves Act 1977, the National Parks Act 1980, Wildlife Act 1953 and any other statute, ordinance, regulation, bylaw, or other enactment (collectively the "Legislation") affecting or relating to the Land or affecting or relating to the Concession Activity, including any regulations made under the Conservation Act 1987 and Wildlife Act 1953 or bylaws made under the Reserves Act 1977 or the National Parks Act 1980; and
 - (c) with all notices and requisitions of any competent authority affecting or relating to the land or affecting or relating to the conduct of the Concession Activity; and
 - (d) with all Department signs and notices placed on or affecting the Land
- 13.2 The Concessionaire must comply with this Concession.
- 13.3 A breach or contravention by the Concessionaire of a relevant conservation management strategy, conservation management plan, management plan or any statement of general policy referred to in clause 13.1 (a) is deemed to be a breach of this Concession.
- 13.4 A breach or contravention by the Concessionaire of any Legislation affecting or relating to the Land or affecting or relating to the Concession Activity is deemed to be a breach of this Concession.

14. When can the Concession be suspended?

14.1 If, in the Grantor's opinion, there is a temporary risk to any natural or historic resource on or in the vicinity of the Land or to public safety whether

arising from natural events such as earthquake, land slip, volcanic activity, flood, or arising in any other way, whether or not from any breach of the terms of this Concession on the part of the Concessionaire, then the Grantor may suspend this Concession.

- 14.2 If, in the Grantor's opinion, the activities of the Concessionaire is having or may have an adverse effect on the natural, historic or cultural values or resources of the Land and the Grantor considers that the effect can be avoided, remedied or mitigated to an extent satisfactory to the Grantor, then the Grantor may suspend this Concession until the Concessionaire avoids, remedies or mitigates the adverse effect to the Grantor's satisfaction.
- 14.3 The Grantor may suspend the Concession for such period as the Grantor determines where the Concessionaire has breached any terms of this Concession.
- 14.4 The Grantor may suspend this Concession while the Grantor investigates any of the circumstances contemplated in clauses 14.1 and 14.2 and also while the Grantor investigates any potential breach or possible offence by the Concessionaire, whether or not related to the Concession Activity under the Conservation Act 1987 or any of the Acts mentioned in the First Schedule of that Act.
- 14.5 The word "investigates" in clause 14.4 includes the laying of charges and awaiting the decision of the Court.
- 14.6 During any period of temporary suspension arising under clauses 14.1 or 14.2 the Concession Fee payable by the Concessionaire is to abate in fair proportion to the loss of use by the Concessionaire of the Land.
- 14.7 The Grantor is not to be liable to the Concessionaire for any loss sustained by the Concessionaire by reason of the suspension of the Concession under this clause 14 including loss of profits.

15. When can the Concession be terminated?

- 15.1 If:
 - (a) the Concessionaire breaches any of the conditions of this Concession; or
 - (b) the whole or any part of the Land is required for the Grantor's use

the Grantor may terminate this Concession at any time in respect of the whole or any part of the Land. Before so terminating the Grantor must give the Concessionaire either:

- (c) one calendar month's notice in writing; or
- (d) such other time period which in the sole opinion of the Grantor appears reasonable and necessary

of the Grantor's intention so to terminate this Concession. If this Concession is terminated then the Grantor, at the Grantor's sole discretion, may adjust the Concession Fee payable or refund any Concession Fee paid in advance.

16. What are the Grantor's Rights to remedy defaults?

16.1 The Grantor may choose to remedy at any time without notice any default by the Concessionaire under this Concession. Where that occurs, the Concessionaire must pay forthwith on demand all reasonable costs incurred by the Grantor in remedying such default.

17. What happens on termination or expiry of the Concession?

17.1 Upon the expiry or earlier termination of this Concession, either as in whole or in part, the Concessionaire is not entitled to compensation for any structure or other improvement erected or carried out by the Concessionaire. The Concessionaire must within such time as the Grantor determines, remove all such structures or other improvements making good at the Concessionaire's expense any damage caused by such removal and leaving the Land in a clean and tidy condition.

18. When is the Grantor's consent required?

18.1 Where the Grantor's consent or approval is expressly required under this Concession then the Concessionaire must seek that approval or consent for each separate time it is required even though the Grantor may have given approval or consent for a like purpose on a prior occasion. Any such consent or approval may be made on such conditions as the Grantor considers appropriate.

19. Are there limitations on public access and closure?

19.1 The Concessionaire acknowledges that the Land is open to the public for access and that the Grantor may close public access during periods of high fire hazard or for reasons of public safety or emergency.

20. How will disputes be resolved?

- 20.1 If a dispute arises between the parties in connection with this Concession the parties must, without prejudice to any other rights or entitlements they may have, attempt to resolve the dispute by agreement using informal dispute resolution techniques such as negotiation, mediation, independent expert appraisal or any other alternative dispute resolution technique. The rules governing any such technique adopted are to be agreed between the parties.
- 20.2 If the dispute cannot be resolved by agreement within 14 days of written notice by one party to the other (or such further period as the parties may agree to in writing) either party may refer the dispute to the Disputes Tribunal, where relevant, or to arbitration which arbitration is to be carried out in accordance with the provisions of the Arbitration Act 1996.
- 20.3 If the parties do not agree on an arbitrator within 10 working days of a party giving written notice of the requirement to appoint an arbitrator the President of the New Zealand Law Society is to appoint the arbitrator. In either case the arbitrator must not be a person who has participated in an informal dispute resolution procedure in respect of the dispute.
- 20.4 The arbitrator must include in the arbitration award reasons for the determination.
- 20.5 Despite the existence of a dispute, each party must continue to perform its obligations under this Concession.

Godley Peaks Pastoral Lease - filming concession - May 2015 (1) Previous template docdm-1406816

21. How are notices sent and when are they received?

- 21.1 Any notice to be given under this Concession is to be in writing and made by personal delivery, fax, by pre paid post or email to the receiving party at the address, fax number or email address specified in Item 14 of Schedule 1. Any such notice is to be deemed to have been received:
 - (a) in the case of personal delivery, on the date of delivery;
 - (b) in the case of post, on the 3rd working day after posting;
 - (c) in the case of email, on the date receipt of the email is acknowledged by tbe addressee by return email or otherwise in writing.
- 21.2 If either party's details stated out in Item 14 of Schedule 1 change then the party whose details change must within 5 working days of such change provide the other party with the changed details.

22. What about the payment of costs?

- 22.1 The Concessionaire must pay the Grantor's legal costs and expenses of and incidental to preparing and signing any extension or variation of this Concession.
- 22.2 The Concessionaire must pay in full immediately and on demand all costs and fees (including solicitor's costs and fees of debt collecting agencies engaged by the Grantor) arising out of and associated with steps taken by the Grantor to enforce or attempt to enforce the Grantor's rights and powers under this Concession. This includes the right to recover outstanding money owed to the Grantor.

23. What about Identification cards and the Grantor's Approved Label?

- 23.1 Before commencing the Concession Activity the Concessionaire must, if required by the Grantor in Item 12 of Schedule 1, obtain Concessionaire Identification cards from the Grantor. The Grantor is to supply such cards to the Concessionaire on a cost recovery basis.
- 23.2 The Concessionaire and any person acting under the authority of the Concession must carry and display a Concession Identification card when carrying out the Concession Activity.
- 23.3 The Concessionaire must obtain sufficient cards to ensure all people acting under the authority of the Concession can carry and display such cards when undertaking the Concession Activity.
- 23.4 The Concessionaire may also access, use and/ or display the Grantor's "Approved Label". This right only exists once the Concessionaire agrees to comply with the Grantor's Approved Label terms and conditions and while the Concession remains operative. When the Concessionaire so requests the Grantor is to forward the Concessionaire an electronic link to the Approved Label. This electronic link is to contain the Approved Label terms and conditions.
- 23.5 The right under this clause 23.4 does not affect the obligation in this clause 23 to carry and display a Concession Identification card.

24. When can the conditions of the Concession be varied?

- 24.1 The Grantor may on each Concession Fee Review Date, after first consulting with the Concessionaire, vary any condition of this Concession to make the condition more effective in addressing adverse effects resulting from the Concession Activity.
- 24.2 Nothing in clause 24.1 otherwise affects the Grantor's rights to vary the Concession under section 17ZC of the Conservation Act 1987.

25. Are there any Special Conditions?

25.1 Special conditions are specified in Schedule 3. If there is a conflict between this Schedule 2 and the Special Conditions in Schedule 3, the Special Conditions shall prevail.

26. The Law

26.1 This Concession is to be governed by, and interpreted in accordance with the laws of New Zealand.

SCHEDULE 3

SPECIAL CONDITIONS

Use of toilets

1. Toilets must be used when camping in the vicinity of huts.

Hut Use

- 2. Hut use is on a first come first served basis and the Concessionaire and their clients together with all other concessionaires and their clients must not be occupy more than half the bunk space of any hut unless the bunks would otherwise be unoccupied. The Concessionaire or their clients must carry alternative accommodation on all overnight trips operated under this Concession and must not use a hut for more than two consecutive nights unless authorised in writing by the Department's appropriate Manager.
- 3. Where the Concessionaire makes use of Department administered huts, the Concessionaire must enter relevant details of the activity into any hut book provided by the Grantor.

Hut fees/ permits

4. Standard hut fees and camping fees must be paid by both guides and clients either before the trip begins or to the hut warden or camp manager, or if unavailable, to the relevant District office on completion of the trip.

Private land

5. This Concession does not confer any right of access over any private land or public conservation land leased by the Grantor. Any arrangements necessary for access over private land or leased land are the responsibility of the Concessionaire. In granting this Concession the Grantor does not warrant that such access can be obtained.

DOC staff

6. The Grantor may send any officer of the Department on any of the activities authorised during the term of this Concession for the purpose of assessing the impact on conservation values, the standard of service offered and compliance with the terms and conditions of the Concession, at no expense to the Grantor.

Use of tracks

7. The Concessionaire must ensure that, where provided, clients remain on formed tracks or well-used routes designed to protect natural and historic features of the Land, do not enter caves and do not exceed any loading limitations placed on facilities and structures.

Camp sites

8. The Concessionaire must ensure that no permanent camp sites are created nor stores or cache of any equipment is left on the Land or in any hut without the specific authority of the Department's appropriate Area Manager.

Wāhi Tapu

9. The Concessionaire must recognise the sensitivity of wāhi tapu and urupa and seek guidance of iwi who claim mana whenua over any parts of the Land prior to providing interpretation on matters of iwi cultural significance and recognise the sensitivity of wāhi tapu and urupa.

In respect to Ngãi Tahu

- 10. The Concessionaire is requested to consult the relevant Papatipu Runanga (www.ngaitahu.iwi.nz) if they wish to use Ngāi Tahu cultural information. If the Concessionaire wishes to use the Tōpuni or statutory acknowledgement information contained in schedules 14-108 of the Ngāi Tahu Claims Settlement Act 1998, or any Department produced interpretative material in respect to Ngāi Tahu cultural information, they are requested to notify the relevant Papatipu Rūnanga, as a matter of courtesy.
- 11. The Concessionaire must, as far as practicable, attend any workshops held by the Department for the purpose of providing information to concessionaires, which is to include the Ngāi Tahu values associated with Tōpuni areas.
- 12. The Concessionaire must ensure any persons employed by the Concessionaire are requested to recognise and provide for Ngāi Tahu values in the conduct of their activities.

Vehicle use

13. The Concessionaire must ensure that none of its vehicles or vehicles of its clients are taken off formed roads.

Vehicle parking

14. The Concessionaire must ensure that its vehicles and the vehicles of its clients are only parked only in designated parking areas.

Animals

15. The Concessionaire must not take, and must ensure that its clients do not take, any animals, including dogs or any domestic pets, onto the Land.

Weeds

16. The Concessionaire must take all precautions to ensure weeds are not introduced to the Land; this includes ensuring that all tyres, footwear, gaiters and packs used by the Concessionaire, its staff and clients are clean before entering the Land.

Interpretation materials

- 17. The Concessionaire must consult with and seek the guidance of iwi claiming mana whenua over any parts of the Land prior to providing interpretation on matters of cultural significance to such iwi.
- 18. The Concessionaire must provide detailed information of any historical, cultural or natural science interpretation provided by the Concessionaire to its clients in the course of the Concession Activity, to the Grantor within thirty days of the date of any such written request by the Grantor.
- 19. If the Grantor considers the interpretative material provided by the Concessionaire above unsatisfactory, the Concessionaire must prepare an

interpretation plan for approval by the Grantor within 60 days of advice from the Grantor that this is required.

Recordings of bird songs

20. The Concessionaire must not and must ensure that its clients do not play recordings of bird songs on the Land.

Didymo

21. The Concessionaire must comply and ensure its clients comply with the Ministry for Primary Industry (MPI)'s "Check, Clean, Dry" cleaning methods to prevent the spread of didymo (*Didymosphenia geminata*) and other freshwater pests when moving between waterways. "Check, Clean, Dry" cleaning methods can be found at - <u>http://www.biosecurity.govt.nz/cleaning</u>. The Concessionaire must regularly check this website and update their precautions accordingly.

Review of Concession Activity

In this clause:

"Client Activity Return Form" means the Form required in clause 6.1 of Schedule 2.

"High Season" means the consecutive three month period when the highest use of the Conservation Activity occurs.

"Permitted Limit" means the frequency of trips and/or the numbers of clients the Concessionaire is permitted in carrying out the Concession Activity specified in Clause 1 of Schedule 3. It also includes the new limit set below.

- 22. On each Concession Fee Review Date the Grantor may review the Concessionaire's Client Activity Return Forms to determine the actual use of the Concession Activity, including the actual numbers of the Concessionaire's clients undertaking the Concession Activity in the twelve month period immediately preceding the Concession Fee Review Date.
- 23. If the Grantor's review of the Client Activity Return Forms shows that:
 - (a) the Concessionaire is not conducting the Concession Activity in all or any of the locations specified in clause 1 of Schedule 3; and/ or
 - (b) on average the Concessionaire has utilised less than 70% of the Permitted Limit over the High Season for that twelve month period,

the Grantor may:

- (c) exclude any such location from the Concession Activity; and/ or
- (d) reduce or eliminate significant under-utilisation over the life of the Concession.
- 24. Such a review must:
 - (a) examine the Concessionaire's Activity Returns for the preceding three years, or for such other period as the Grantor considers appropriate, to ascertain trends and in relation to the Permitted Limit component

identify the amount of the Permitted Limit that has not been utilised over the High Season/s; and

- (b) offer the Concessionaire an opportunity to present any business plans demonstrating any expected use of such location or an explanation as to why the Concession Activity has not been conducted at the location; and/ or
- (c) invite the Concessionaire to offer an explanation as to why the level of use has, on average, been below 70% of the Permitted Limit over the High Season/s; and to demonstrate that it has the capacity and the intention to increase utilisation. Capacity may be demonstrated by the existence of the equipment and infrastructure necessary to utilise the Permitted Limit. Intention may be demonstrated through business plans showing anticipated levels of operation up to the Permitted Limit.
- 25. Following this review the Grantor is to:
 - (a) determine if any location is to be excluded from the Concession; and /or
 - (b) determine if a reduction of the Permitted Limit is appropriate and the amount of that reduction; and
 - (c) advise the Concessionaire of the date the exclusion and/ or the reduced Permitted Limit becomes effective.
- 26. If any location is so excluded and or the Permitted Limit is so reduced the Grantor is not to be liable for any subsequent loss sustained by the Concessionaire (including loss of profits) even if that loss is wholly or partly a result of such exclusion and/or such a reduction in the Permitted Limit.
- 27. The Grantor may exercise the Grantor's right to exclude a location and/ or to reduce the Permitted Limit notwithstanding any prior waiver or failure to undertake action by the Grantor or any indulgence granted by the Grantor for any matter or default.

Filming Special Conditions

- 28. The Concessionaire is required to complete a filming application form for each proposed filming event to enable the Grantor to assess the potential of any significant effects on the environment for each particular event.
- 29. The Grantor may require the Concessionaire to pay a bond if it is established that there is potential for damage to the environment at the filming location. The bond will be determined by the Grantor and will be sufficient to cover the cost of restoring the location to its previous condition and will include monitoring of the restoration.
- 30. The Concessionaire must make adequate provision for suitable sanitary facilities for the site if required by the Grantor and for the disposal of all refuse material and is to comply with the reasonable directions of the Grantor in regard to these matters.
- 31. The Concessionaire must, upon request by the Grantor, submit written engineering or building plans and details to the Grantor for approval before:a. erecting or altering any structure on the site

- b. bringing any structure onto the site
- c. installing any facilities on the site
- d. altering the site in any way
- 32. The Concessionaire must not commence any work on the site until the Grantor has given written approval.
- 33. The Concessionaire must comply with all statutory requirements including building consents and code compliance certificates under the Building Act 1991 and must also apply for any other permits and resource consents required for the activities associated with the filming

Other Special Conditions

- 34. Provided that the Concessionaire for this concession being Verity Farms Limited is also the Concessionaire for the Guiding Licence concession being [insert reference], then the concession Annual Management Fee in Schedule 1, item 6 is not payable. This is on the basis that while the two concessions are both held by Verity Limited, one Annual Management Fee is sufficient.
- 35. The public have unrestricted access rights to the Land. The Grantor may erect signposts and mark the access routes through the Land for the benefit of the public.
- 36. This Licence authorises the Concessionaire to carry out the Concession Activity only, but does not authorise the Concessionaire to do anything which would not otherwise be permitted or which would require some other form of authority from the Grantor or any other person. Without limiting the effect of this clause and by way of example the Concessionaire is not authorised under this Concession to:
 - i. Access any part of the Land by air unless such access is with an air operator who is otherwise authorised to fly into the area concerned.
 - ii. Breach any law, regulation, by law, restriction, requirement, notice, sign or provision of any Conservation Management Strategy or management plan.
 - iii. Access any Land or water other than that listed in Schedule 1.
- 37. The Grantor reserves the right to apply restrictions on the Concession Activity, including but not limited to restricting the frequency of trips or reducing the maximum party size, or withdrawing all or part of the Land or Concession Activity approved, if in the opinion of the Grantor the Concession Activity is having, or may have, a significant adverse effect on the physical or social environment and the effect cannot be avoided, remedied or mitigated to an extent satisfactory to the Grantor. The Concessionaire shall not be entitled to any compensation in the event of such action being taken.
- 38. The Grantor reserves the right to request further or different activity related information in order to best monitor and determine any effects of the Concession Activity on the Land.
- 39. The Concessionaire shall ensure that guides brief clients on the historic and natural values of sites visited and take all practical measures to ensure that these values are not damaged or compromised in any way.

Godley Peaks Pastoral Lease - filming concession - May 2015 (1) Previous template docdm-1406816

- 40. This Concession does not confer any right for the Concessionaire and clients to take any part or parts of any indigenous plant on any land administered by the Department of Conservation.
- 41. The Concessionaire shall remove all rubbish, including waste food, associated with the Concession Activity from the Land at the conclusion of each trip.
- 42. The Concessionaire shall ensure that guided groups show courtesy and consideration to other users of public conservation land at all times.
- 43. The Concessionaire shall ensure that guided groups take all reasonable care to avoid dominating facilities and special interest sites to the detriment of other users.
- 44. The Concessionaire must report all incidents and accidents of what so ever nature and whether or not the subject of an official search and rescue operation to the Grantor. At the request of the Grantor the Concessionaire must supply a report in writing of any inquiry into the cause of an incident, or accident, and if in the opinion of the Grantor the inquiry reveals that a reasonable standard of safety was not maintained and/or the Concessionaire, their servants, or employees or agents were negligent then the Grantor may determine this Licence. The Concessionaire must at the request of the Grantor make available any employee, servant or agent who in the opinion of the Grantor might assist in any such enquiry.
- 45. If the Grantor determines that the conditions of this Document or the effects of Concession Activity should be monitored, the Concessionaire shall meet: either the full costs of any monitoring programme that is implemented; or, if the Grantor determines that the costs should be apportioned among several Concessionaires who use the same locations, part of the costs of the monitoring programme. These costs will include the Department's standard charge-out rates for staff time and the mileage rates for vehicle use associated with the monitoring programme.

SCHEDULE 4

Designations Plan or map

Any further plan if required

Godley Peaks Pastoral Lease - filming concession - May 2015 (1) Previous template docdm-1406816 .__

SCHEDULE 4 - Client Activity Return Form

Month/Year/.....

activities including the cumulative impacts on conservation and recreation values. We need to know how many trips you make to each site and the party size of each trip. If you make more than one trip to the same site on the same day please record each trip on a new line (see example below). Supplying this information to the Department not only helps us to collect the fees for your activity, more importantly it helps us to understand the visitor use patterns and trends on public conservation lands. This information is critically important and helps us to manage the effects of all We appreciate your time and efforts to supply this information accurately and promptly.

1 4 4 4 1 1 1 4 1 1 1 1		-						

Godley Peaks Pastoral Lease - filming concession - May 2015 (1) Previous template docdm-1406816

- 23 -

Appendix 8: Concession for telecommunications facility (DOC-2331607)



Department of Conservation Te Papa Atawhai

Concession Document (General Licence)

Concession Number:

THIS CONCESSION is made this day of					
PAR	PARTIES:				
Mini	ister of Conservation (the Grantor)				
Veri	ty Farms Limited (the Concessionaire)				
BAC	KGROUND				
А.	The Department of Conservation ("Department") Te Papa Atawhai is responsible for managing and promoting conservation of the natural and historic heritage of New Zealand on behalf of, and for the benefit of, present and future New Zealanders.				
в.	The land described in Item 1 of Schedule 1 as the Land is a Conservation Area or a Reserve under the management of the Grantor.				
C.	Sections 66 and 68 of the Crown Pastoral Land Act 1998 authorise the Grantor to grant a Concession for a Concession Activity in a Conservation Area and a Reserve under section $17Q(1)$ of the Conservation Act 1987 or section 59A of the Reserves Act 1977 (whichever is relevant in the circumstances).				
D.	The Concessionaire wishes to carry out the Concession Activity on the Land subject to the terms and conditions of this Concession.				
Е.	The parties wish to record the terms and conditions of this Concession and its Schedules.				
OPE	RATIVE PARTS				
F.	In exercise of the Grantor's powers under the Conservation legislation the Grantor GRANTS to the Concessionaire a LICENCE to carry out the Concession Activity on the Land subject to the terms and conditions contained in this Concession and its Schedules.				

Concession Number:

SIGNED on behalf of the Minister of Conservation by [insert name and title of delegate]	SIGNED by Verity Farms Limited by its Director Ann Poindexter:
acting under delegated authority in the presence of:	in the presence of: Witness Signature:
Witness Signature:	Witness Name:
Witness Name: Witness Occupation:	Witness Address:
Witness Address:	
A copy of the Instrument of Delegation may be inspected at the Director-General's office at 18-22 Manners Street, Wellington.	

Godley Peaks Pastoral Lease - telecommunication concession - May 2015 - DOC-2331607 Godley Peaks Pastoral Lease - telecommunication concession - May 2015 - DOC-2331607 Previous template docdm-1406816

SCHEDULE 1

1.	Land (clause 2)	Conservation Land situated in the Canterbury Land District and designated as Conservation Area on the Hall Range and Mistake Peak being labelled CA1 and shaded pink in the Plan attached to the Proposal.
2.	Concession Activity (clause 2)	To establish, operate, maintain and repair a telecommunications facility for the provision of communication services and the right of helicopter access to the site labelled as "TA" on the plan attached to the Proposal
3.	Term (clause 3)	30 years commencing on the date that an approved plan is registered, affecting Certificate of Title CB 30B/914 (Canterbury Registry), vesting the land in the Crown as a conservation area, being the day of 20 (the commencement date)
4.	Renewal(s) (clause 3)	One right of renewal from the 30 th anniversary of the commencement date for a further term of 30 years.
5.	Final Expiry Date (clause 3)	The 60th anniversary of the commencement date.
6.	Concession Fee (clause 4)	Activity Fee: \$1.00 per annum plus GST Management Fee: \$200.00 per annum plus GST Environmental Monitoring Fee (clause 9) To be negotiated between Grantor and Concessionaire if required (see Schedule 3 Special condition 8) Community Services Contribution (clause 6) Not required
7	Concession Fee Payment Instalments (clause 4)	Annually, in arrears upon commencement of the term set out in clause 4 for payment instalments.
8.	Concession Fee Payment Date(s) (clause 4)	On or before the date specified on the invoice issued by the Grantor.
9.	Penalty Interest Rate (clause 4)	Double the current Official Cash Rate (OCR). See Reserve Bank of New Zealand website
10.	Concession Fee Review Date(s) (clause 5)	On the third anniversary of the date of commencement of this Concession set out in item 3 above and the corresponding date every 3 years thereafter until the expiry of the term.
11.	Health and Safety (clause 13)	Not required
12.	Concessionaire Identification (clause 31)	Not required

Concessionaire's initials

Grantor's initials

Godley Peaks Pastoral Lease - telecommunication concession - May 2015 - DOC-2331607 Godley Peaks Pastoral Lease - telecommunication concession - May 2015 - DOC-2331607 Previous template docdm-1408400

13.	Insurance (To be obtained by Concessionaire) (clause 12)	 Types and amounts: Public Liability Insurance for: (a) General indemnity for an amount no less than \$1,000,000.00; and (b) Forest and Rural Fires Act extension for an amount no less than \$1,00,000.00; and Subject to review on each Concession Fee Review Date
14.	Addresses for Notices (clause 24)	The Grantor's address is: Physical Address: Department of Conservation Conservation House 77 Lower Stuart Street Dunedin 9016 Postal Address: PO Box 5244 Moray Place Dunedin 9058 Phone: (03) 477 0677 Email: permissionsdunedin@doc.govt.nz
		The Concessionaire's address in New Zealand is: Address: C/- Milnes Beatson Ld 29 Wallace Street Motueka 7120 Phone: (03) 528 8760 Email: david@mba.net.nz Director's Address:
		210 Patons Rock Road RD2 Takaka 7182 Phone: 021 531 227 Email: ann@verityfarms.co.nz
15.	Special Conditions (clause 34)	See Schedule 3.

Note: The clause references are to the Grantors Standard Terms and Conditions of Licence set out in Schedule 2.

Note: Please initial each page of Schedule 1

SCHEDULE 2

STANDARD TERMS AND CONDITIONS OF LICENCE

1. Interpretation

- 1.1 Where the Grantor's consent or approval is expressly required under a provision of this Concession, the Concessionaire must seek the consent or approval of the Grantor for each separate occasion it is required notwithstanding that the Grantor has granted consent or approval for a like purpose on a prior occasion.
- 1.2 The Concessionaire is responsible for the acts and omissions of its employees, contractors, agents, clients and invitees (excluding other members of the public accessing the Land). The Concessionaire is liable under this Concession for any breach of the terms of the Concession by its employees, contractors, agents, clients and invitees (excluding other members of the public accessing the Land), as if the breach had been committed by the Concessionaire.
- 1.3 Where this Concession requires the Grantor to exercise a discretion or give any approval or provides for any other actions by the Grantor, then the Grantor must act reasonably and within a reasonable time. When a consent is required under this Concession such consent must not be unreasonably withheld.
- 1.4 Where this Concession provides for approvals, directions, reports and consents to be given by one party to the other, those approvals, directions, reports and consents must be given by notice in writing and clause 24 is to apply.

2. What is being authorised?

- 2.1 The Concessionaire is only allowed to use the Land for the Concession Activity.
- 2.2 The Concessionaire must exercise reasonable skill, care and diligence in carrying out the Concession Activity, in accordance with standards of skill, care and diligence normally practised by suitably qualified and experienced people in carrying out such activities.
- 2.3 The Concessionaire must provide the Grantor with evidence of the competency and qualifications of its employees and contractors if the Grantor so requests.

3. How long is the Concession for - the Term?

- 3.1 This Concession commences on the date set out in Item 3 of Schedule 1 and ends on the Final Expiry Date specified in Item 5 of Schedule 1.
- 3.2 If there is a right of renewal then the Grantor at the Concessionaire's cost must renew the Term for a further period as set out in Item 4 of Schedule 1 provided the Concessionaire:
 - (a) gives the Grantor at least three month's written notice before the end of the Term, which notice is to be irrevocable, of the Concessionaire's intention to renew this Concession; and
 - (b) at the time notice is given in accordance with this clause the Concessionaire is not in breach of this Concession.

Godley Peaks Pastoral Lease - telecommunication concession - May 2015 - DOC-2331607 Previous template docdm-1408400 3.3 The renewal is to be on the same terms and conditions expressed or implied in this Concession except that the Term of this Concession plus all further renewal terms is to expire on or before the Final Expiry Date.

4. What are the fees and when are they to be paid?

- 4.1 The Concessionaire must pay to the Grantor in the manner directed by the Grantor the Concession Fee and any other payment comprised in the Total Payment specified in Item 6 of Schedule 1 in the instalments and on the Concession Fee Payment Date specified in Items 7, and 8 of Schedule 1.
- 4.2 If the Concessionaire fails to make payment within 14 days of the Concession Fee Payment Date then the Concessionaire is to pay interest on the unpaid Concession Fee from the Concession Fee Payment Date until the date of payment at the Penalty Interest Rate specified in Item 9 of Schedule 1.

5. When can the fee be reviewed?

- 5.1 The Grantor is to review the Concession Fee on the Concession Fee Review Dates in the following manner:
 - (a) The Grantor must commence the review not earlier than 3 months before a Concession Fee Review Date and no later than 9 months following the Concession Fee Review Date by giving notice to the Concessionaire.
 - (b) Subject to clause 5.1(e) the notice must specify the Concession Fee which the Grantor considers to be the market value for the Concession Activity as at the Concession Fee Review Date having regard to the matters specified in section 17Y(2) of the Conservation Act 1987.
 - (c) If, within 28 days of receipt of the Grantor's notice, the Concessionaire gives notice to the Grantor that the Concessionaire disputes the proposed new Concession Fee, the new Concession Fee is to be determined in accordance with clause 5.2(a) or (b).
 - (d) If the Concessionaire does not give notice to the Grantor under clause 5.1(c) the Concessionaire is to be deemed to have accepted the Concession Fee specified in the Grantor's notice.
 - (e) Notwithstanding clause 5.1(b) the new Concession Fee so determined or accepted must not be less than the Concession Fee payable during the year preceding the particular Concession Fee Review Date and is to be the Concession Fee payable by the Concessionaire from the Concession Fee Review Date.
 - (f) Until determination of the new Concession Fee, the Concession Fee payable by the Concessionaire from the Concession Fee Review Date is to be the Concession Fee specified in the Grantor's notice. On determination of the new Concession Fee an adjustment is to be made and paid, either by the Grantor or by the Concessionaire, whichever is applicable.
- 5.2 Immediately the Concessionaire gives notice to the Grantor under clause 5.1(c) the parties are to endeavour to agree on a new Concession Fee. If the parties are unable to reach agreement within 28 days the new Concession Fee is to be determined either:

- (a) By one party giving notice to the other requiring the new Concession Fee to be determined by the Disputes clause (clause 22) or, if the parties agree,
- (b) by registered valuers acting as experts and not as arbitrators as follows:
 - (i) Each party must appoint a valuer and give notice of the appointment to the other party within 14 days of the parties agreeing to determine the new Concession Fee by this means.
 - (ii) If the party receiving a notice does not appoint a valuer within the 14 day period the valuer appointed by the other party is to determine the new Concession Fee and that valuer's determination is to be binding on both parties.
 - (iii) Before commencing their determination the respective valuers must appoint an umpire who need not be a registered valuer.
 - (iv) The valuers are to determine the new Concession Fee which they consider to be the market value for the Concession Activity as at the Concession Fee Review Date having regard to the matters specified in section 17Y(2) of the Conservation Act 1987 but in no case is the new Concession Fee to be less than the Concession Fee payable during the year preceding the particular Concession Fee Review Date. If they fail to agree the Concession Fee is to be determined by the umpire.
 - (v) In determining the Concession Fee the valuers or umpire are to disregard the annual cost to the Concessionaire to maintain or provide access to the Land.
 - (vi) Each party is to be given the opportunity to make written or oral representations or submissions to the valuers or the umpire subject to such reasonable time and other limits as the valuers or the umpire may prescribe.
 - (vii) The valuers or the umpire must have regard to any such representations but are not bound by them.
- (c) The valuers or umpire must give written notice to the parties once they have determined the new Concession Fee. The notice is to be binding on the parties and is to provide how the costs of the determination are to be borne.
- (d) (i) If a Concession Fee Review Date is postponed because of a moratorium imposed by law the Concession Fee Review is to take place at the date the moratorium is lifted or so soon afterwards as is practicable; and
 - (ii) the Concession Fee Review is to establish the market value for the Concession Activity as at that date instead of the date fixed under clause 5.1 having regard to the matters specified in section 17Y(2) of the Conservation Act 1987 but in no case is the new Concession Fee to be less than the Concession Fee payable during the year preceding the particular Concession Fee Review Date; and
 - (iii) each subsequent Concession Fee Review is to take place in accordance with the procedure fixed in clause 5.1.

Godley Peaks Pastoral Lease - telecommunication concession - May 2015 - DOC-2331607 Previous template docdm-1408400

6. Are there any other charges?

- 6.1 The Concessionaire must pay all levies rates and other charges, including utility charges payable in respect of the Land or for the services provided to the Land which relate to the Concessionaire's use of the Land or the carrying on of the Concession Activity.
- 6.2 The Grantor is not liable for any cost incurred in re-establishing the supply of any utilities in the event of any of them becoming unavailable for any reason.
- 6.3 Where the Grantor has paid such levies, rates or other charges the Concessionaire must on receipt of an invoice from the Grantor pay such sum to the Grantor within 14 days of receiving the invoice. If payment is not made within the 14 days then the Concessionaire is to pay interest on the unpaid sum from the date payment was due until the date of payment at the Penalty Interest Rate specified in Item 9 of Schedule 1.
- 6.4 Where the Grantor or Director-General has provided a community service, benefit or facility for the benefit of the Concessionaire under section 17ZH of the Conservation Act 1987, the Concessionaire must pay the Grantor the amount specified in Item 7 of Schedule 1 as part of the Total Payment specified in Item 6 of Schedule 1 on the Concession Fee Payment Dates specified in Item 8 of Schedule 1.

7. When can the Concession be assigned?

- 7.1 The Concessionaire must not transfer, sub licence, assign, mortgage or otherwise dispose of the Concessionaire's interest under this Concession or any part of it (which includes the Concessionaire entering into a contract or any other arrangement whatsoever whereby the Concession Activity would be carried out by a person (called the assignee) other than the Concessionaire) without the prior written consent of the Grantor.
- 7.2 The Grantor may in the Grantor's discretion decline any application for consent under clause 7.1.
- 7.3 Sections 17P, 17S, 17T, 17U, 17W, 17X, 17ZB and 17ZC of the Conservation Act 1987 apply to applications for consent under this clause unless the Grantor, in the Grantor's discretion, decides otherwise.
- 7.4 If the Grantor gives consent under this clause then the Concessionaire remains liable to observe and perform the terms and conditions of this Concession throughout the Term and is to procure from the Assignee a covenant to be bound by the terms and conditions of this Concession.
- 7.5 The Concessionaire must pay the costs reasonably incurred by the Grantor incidental to any application for consent, whether or not such consent is granted.
- 7.6 If the Concessionaire is not a publicly listed company then any change in the shareholding of the Concessionaire altering the effective control of the Concessionaire is to be deemed to be an assignment and requires the consent of the Grantor.

8. What are the obligations to protect the environment?

8.1 The Concessionaire must not cut down or damage any vegetation; or damage any natural feature or historic resource on the Land; or light any fire on the Land without the prior consent of the Grantor.

- 8.2 The Concessionaire must at its cost keep the Land in a clean and tidy condition and free of weeds and all organisms specified as pests in a relevant pest management strategy.
- 8.3 The Concessionaire must not store hazardous materials on the Land nor store other materials on the Land where they may obstruct the public or create a nuisance
- 8.4 If directed by the Grantor, the Concessionaire must take all steps necessary to control, or, at the Grantor's option, contribute to the cost of controlling any pest, insect or rodent infestation occurring in or emanating from the Land or any structure or facility on the Land, and if directed by the Grantor, engage a pest exterminator approved by the Grantor.
- 8.5 The Concessionaire must make adequate provision for suitable sanitary facilities for the Land if directed by the Grantor and for the disposal of all refuse material and is to comply with the reasonable directions of the Grantor in regard to these matters.
- 8.6 The Concessionaire must keep all structures, facilities and land alterations and their surroundings in a clean and tidy condition. If reasonably directed by the Grantor the Concessionaire must paint all structures and facilities in colours approved by the Grantor and with paints of a type approved by the Grantor.
- 8.7 If, during the Term, the Concessionaire removes a structure or facility from the Land the Concessionaire must, unless the Grantor directs otherwise, repair and make good at its own expense all damage which may have been done by the removal and must leave the Land in a clean and tidy condition.
- 8.8 The Concessionaire must not bury:
 - (a) any toilet waste within 50 metres of a water source on the Land; or
 - (b) any animal or fish or any part thereof within 50 metres of any water body, water source or public road or track.

9. What about Environmental Monitoring?

- 9.1 The Concessionaire must, during the Term, if the Grantor so directs, design in consultation with the Grantor and undertake a programme to monitor and report on the environmental effects of the Concessionaire's use of the Land and conduct of the Concession Activity.
- 9.2 If the Grantor does not issue a direction under clause 9.1 the Concessionaire must, during the Term, pay to the Grantor the annual Environmental Monitoring Contribution specified in Item 6 of Schedule 1 to enable the Grantor to design and undertake a programme to monitor the environmental effects of the Concessionaire's use of the Land and conduct of the Concession Activity.

10. When can new structures be erected or land alterations occur?

- 10.1 The Concessionaire must not erect, alter or bring on to the Land any structure not authorised in Schedule 3 nor alter the Land in any way without the prior approval of the Grantor.
- 10.2 In giving approval under clause 10.1 the Grantor may, in the Grantor's sole and absolute discretion, impose any reasonable terms and conditions, including a review of the Concession Fee, as the Grantor considers appropriate under this clause; and may also decline the grant of such approval

after consideration of the relevant conservation and environmental issues.

- 10.3 The Concessionaire must pay to the Grantor all costs associated with applications for approval under this clause determined at the standard rates then applying in the Department for cost recovery of staff time and expenses.
- 10.4 The Concessionaire must, upon request by the Grantor, submit written engineering or building plans and details to the Grantor for approval before:
 - (a) erecting new structure or altering any structure on the Land
 - (b) altering the Land in any way.
- 10.5 The Concessionaire must at all times where a building warrant of fitness under the Building Act 2004 is required display a copy of the relevant current certificate showing the location of the compliance schedule in a place in each building (as defined in that Act) on the Land to which users of the building have ready access.
- 10.6 The Concessionaire must keep and maintain all building systems and any structure on the Land in accordance with the, requirements of any compliance schedule.
- 10.7 The Concessionaire must retain and make available to any territorial authority and any other person with a right to inspect any structures on the Land under the Building Act 2004 a copy of the compliance schedule, together with the written reports relating to compliance with the compliance schedule over the previous two year period.

11. What about advertising?

- 11.1 The Concessionaire must not erect or display any signs or advertising on the Land without the prior approval of the Grantor. At the expiry or termination of this Concession the Concessionaire must remove all signs and advertising material and make good any damage caused by the removal.
- 11.2 If directed by the Grantor, the Concessionaire must ensure that all its advertising and promotional material specifies that it is carrying out the Concession Activity under a Concession granted by the Grantor on land administered by the Department.
- 11.3 If directed by the Grantor, the Concessionaire must include information in its advertising and promotional material which assists its clients to understand the features and values of the natural and historic resources of the Land and the surrounding area.
- 11.4 The Concessionaire is encouraged to obtain information from and have regard to the views of tangata whenua.

12. What are the liabilities and who insures?

- 12.1 The Concessionaire agrees to use the Land at the Concessionaire's own risk and releases to the full extent permitted by law the Grantor and the Grantor's employees and agents from all claims and demands of any kind and from all liability which may arise in respect of any accident, damage or injury occurring to any person or property in or about the Land.
- 12.2 The Concessionaire must indemnify the Grantor against all claims, actions, losses and expenses of any nature which the Grantor may suffer or incur or for which the Grantor may become liable arising from the Concessionaire's performance of the Concession Activity.

- 12.3 This indemnity is to continue after the expiry or termination of this Concession in respect of any acts or omissions occurring or arising before its expiry or termination.
- 12.4 The Concessionaire has no responsibility or liability for costs, loss, or damage of whatsoever nature arising from any act or omission or lack of performance or any negligent or fraudulent act or omission by the Grantor, or any contractor or supplier to the Grantor, or any employee or agent of the Grantor.
- 12.5 Despite anything else in clause 12 the Concessionaire is not liable for any indirect or consequential damage or loss howsoever caused.
- 12.6 The Grantor is not liable and does not accept any responsibility for damage to or interference with the Land, the Concession Activity, or to any structures, equipment or facilities on the Land or any other indirect or consequential damage or loss due to any natural disaster, vandalism, sabotage, fire, or exposure to the elements except where, subject to clause 12.7, such damage or interference is caused by any wilful act or omission of the Grantor, the Grantor's employees, agents or contractors.
- 12.7 Where the Grantor is found to be liable in accordance with clause 12.6, the total extent of the Grantor's liability is limited to \$1,000,000 in respect of the Concessionaire's structures, equipment and facilities.
- 12.8 Despite anything else in clause 12 the Grantor is not liable for any indirect or consequential damage or loss howsoever caused.
- 12.9 Without prejudice to or in any way limiting its liability under this clause 12 the Concessionaire at the Concessionaire's expense must take out and keep current policies for insurance and for the amounts not less than the sums specified in Item 13 of Schedule 1 with a substantial and reputable insurer.
- 12.10 After every three year period of the Term the Grantor may, on giving 10 working day's notice to the Concessionaire, alter the amounts of insurance required under clause 12.9. On receiving such notice the Concessionaire must within 10 working days take out and keep current policies for insurance and for the amounts not less than the sums specified in that notice.
- 12.11 The Concessionaire must provide to the Grantor within 5 working days of the Grantor so requesting:
 - (a) details of any insurance policies required to be obtained under this Concession, including any renewal policies if such renewal occurs during the Term; and/ or;
 - (b) a copy of the current certificate of such policies.

13. What about Health and Safety?

13.1 The Concessionaire must exercise the rights granted by this Concession in a safe and reliable manner and must comply with the Health and Safety in Employment Act 1992 and its regulations and all other provisions or requirements of any competent authority relating to the exercise of this Concession. The Concessionaire must comply with its safety plan (if one is required in Item 11 of Schedule 1), and with any safety directions of the Grantor.

- 13.2 Before commencing the Concession Activity the Concessionaire must, where the Concessionaire has Qualmark or Outdoorsmark certification, provide the Grantor with a copy of that certification.
- 13.3 If the Concessionaire does not hold Qualmark or Outdoorsmark certification then before commencing the Concession Activity the Concessionaire must, if required by Item 11 of Schedule 1:
 - (a) prepare a safety plan;
 - (b) have it audited by a suitably qualified person approved by the Grantor; and forward to the Grantor a certificate from the auditor certifying that the safety plan is suitable for the Concession Activity; and
 - (c) the Concessionaire must obtain from the auditor details as to when the safety plan is to be re-audited. The Concessionaire must comply with any such requirement to re-audit and forward a copy of the reaudit certificate to the Grantor within 5 working days of the certificate being issued.
- 13.4 For any Concession Activity that is subject to the Health and Safety in Employment (Adventure Activities) Regulations 2011, proof of registration with WorkSafe New Zealand will satisfy the Grantor's requirement under clause 13.3(b).
- 13.4 The Grantor may at any time request the Concessionaire to provide the Grantor with a copy of the current safety plan in which case the Concessionaire must provide the copy within 10 working days of receiving the request.
- 13.5 Receipt of the safety plan/audit certificate by the Grantor is not in any way to limit the obligations of the Concessionaire under clause 13 and is not to be construed as implying any responsibility or liability on the part of the Grantor.
- 13.6 The Concessionaire must:
 - (a) notify the Grantor of any natural events or activities on the Land or the surrounding area which may endanger the public or the environment;
 - (b) take all practicable steps to protect the safety of all persons present on the Land and must, where necessary, erect signposts warning the public of any dangers they may encounter as a result of the Concessionaire's operations;
 - (c) take all practicable steps to eliminate any dangers to the public and must clearly and permanently mark any that remain and of which the Concessionaire is aware;
 - (d) record and report to the Grantor all accidents involving serious harm within 24 hours of their occurrence and forward an investigation report within 3 days of the accident occurring;
 - (e) ensure that all contracts between the Concessionaire and any contractors contain, at a minimum, the same requirements as clause 13;
 - (f) be satisfied that facilities or equipment provided by the Grantor to enable the Concession Activity to be carried out meet the safety requirements of the Concessionaire;

(g) not bring onto the Land or any land administered by the Department any dangerous or hazardous material or equipment which is not required for purposes of the Concession Activity; and if such material or equipment is required as part of the Concession Activity, the Concessionaire must take all practicable steps at all times to ensure that the material or equipment is treated with due and proper care.

14. What are the compliance obligations of the Concessionaire?

- 14.1 The Concessionaire must comply where relevant:
 - (a) with the provisions of any conservation management strategy or conservation management plan under the Conservation Act 1987 or Part IIA of the Reserves Act 1977, or any general policy statement made under the Conservation Act 1987, Reserves Act 1977, National Parks Act 1980, or Wildlife Act 1953, or management plan under section 45 of the National Parks Act 1980, whichever is appropriate to the Land, together with any amendment or review of any policy, strategy or plan whether approved before, on, or after the date on which this Concession takes effect; and
 - (b) with the Conservation Act 1987, the Reserves Act 1977, the National Parks Act 1980, Wildlife Act 1953 and any other statute, ordinance, regulation, bylaw, or other enactment (collectively the "Legislation") affecting or relating to the Land or affecting or relating to the Concession Activity, including any regulations made under the Conservation Act 1987 and Wildlife Act 1953 or bylaws made under the Reserves Act 1977 or the National Parks Act 1980; and
 - (c) with all notices and requisitions of any competent authority affecting or relating to the Land or affecting or relating to the conduct of the Concession Activity; and
 - (d) with all Department signs and notices placed on or affecting the Land; and
 - (e) with all reasonable notices and directions of the Grantor concerning the Concession Activity on the Land.
- 14.2 The Concessionaire must comply with this Concession.
- 14.3 A breach or contravention by the Concessionaire of a relevant conservation management strategy, conservation management plan, management plan or any statement of general policy referred to in clause 14.1(a) is deemed to be a breach of this Concession.
- 14.4 A breach or contravention by the Concessionaire of any Legislation affecting or relating to the Land or affecting or relating to the Concession Activity is deemed to be a breach of this Concession.

15. What are the Grantor's rights to remedy defaults?

- 15.1 The Grantor may elect to remedy at any time, after giving notice, if practicable, any default by the Concessionaire under this Concession. Before electing to so remedy in accordance with this clause, the Grantor must, if practicable, first give the Concessionaire notice of the default and a reasonable opportunity to remedy the default.
- 15.2 The Concessionaire must pay to the Grantor forthwith on demand all reasonable costs and expenses incurred by the Grantor, including legal costs

and expenses as between solicitor and client, in remedying such default. The Concessionaire is to pay interest on such costs and expenses if payment is not made within 14 day's of the Grantor's demand from the date of the demand until the date of payment at the Penalty Interest Rate specified in Item 12 of Schedule 1.

16. When can the Concession be suspended?

- 16.1 If, in the Grantor's opinion, there is a temporary risk to any natural or historic resource on or in the vicinity of the Land or to public safety whether arising from natural events such as earthquake, land slip, volcanic activity, flood, or arising in any other way, whether or not from any breach of the terms of this Concession on the part of the Concessionaire, then the Grantor may suspend this Concession.
- 16.2 If, in the Grantor's opinion, the activities of the Concessionaire are having or may have an adverse effect on the natural, historic or cultural values or resources of the Land and the Grantor considers that the effect can be avoided, remedied or mitigated to an extent satisfactory to the Grantor, then the Grantor may suspend this Concession until the Concessionaire avoids, remedies or mitigates the adverse effect to the Grantor's satisfaction.
- 16.3 The Grantor may suspend the Concession for such period as the Grantor determines where the Concessionaire has breached any terms of this Concession.
- 16.4 The Grantor may suspend this Concession while the Grantor investigates any of the circumstances contemplated in clauses 16.1 and 16.2 and also while the Grantor investigates any potential breach or possible offence by the Concessionaire, whether or not related to the Concession Activity under the Conservation Act 1987 or any of the Acts mentioned in the First Schedule of that Act.
- 16.5 The word "investigates" in clause 16.4 includes the laying of charges and awaiting the decision of the Court.
- 16.6 During any period of temporary suspension arising under clauses 16.1 or 16.2 the Concession Fee payable by the Concessionaire is to abate in fair proportion to the loss of use by the Concessionaire of the Land.
- 16.7 The Grantor is not to be liable to the Concessionaire for any loss sustained by the Concessionaire by reason of the suspension of the Concession under this clause 16 including loss of profits.

17. When can the Concession be terminated?

- 17.1 The Grantor may terminate this Concession either in whole or in part:
 - (a) by 14 days notice to the Concessionaire if the Concession Fee or any other money payable to the Grantor under this Concession is in arrears and unpaid for 14 days after any of the days appointed for payment whether it has been lawfully demanded or not; or
 - (b) by 14 days notice to the Concessionaire or such sooner period as it appears necessary and reasonable to the Grantor if.
 - (i) the Concessionaire breaches any terms of this Concession and in the Grantor's sole opinion the breach is able to be rectified; and

Godley Peaks Pastoral Lease - telecommunication concession - May 2015 - DOC-2331607 Previous template docdm-1408400

- (ii) the Grantor has notified the Concessionaire of the breach; and
- (iii) the Concessionaire does not rectify the breach within 7 days of receiving notification; or such earlier time as specified by the Grantor; or
- (c) by notice in writing to the Concessionaire where the Concessionaire breaches any terms of this Concession and in the sole opinion of the Grantor the breach is not capable of being rectified; or
- (d) immediately by notice in writing to the Concessionaire where the Concessionaire breaches clauses 12.9 and 13; or
- (e) by notice in writing to the Concessionaire if the Concessionaire ceases to conduct the Concession Activity or, in the reasonable opinion of the Grantor, the conduct of the Concession Activity is manifestly inadequate; or
- (f) by notice in writing to the Concessionaire if the Concessionaire is convicted of an offence under the Conservation Act 1987 or any of the Acts listed in the First Schedule to that Act or any statute, ordinance, regulation, bylaw, or other enactment affecting or relating to the Land or which in the Grantor's sole opinion affects or relates to the Concession Activity; or
- (g) by notice in writing to the Concessionaire if the Concessionaire or the Guarantor is dissolved; or enters into any composition with or assignment for the benefit of its creditors; or is adjudged bankrupt; or being a company, has a receiver appointed; or is put into liquidation; or is placed under statutory management; or has a petition for winding up presented against it; or is otherwise unable to pay its debts as they fall due; or the estate or interest of the Concessionaire is made subject to a Writ of Sale or charging order; or the Concessionaire ceases to function or operate; or
- (h) immediately if there is, in the opinion of the Grantor, a permanent risk to public safety or to the natural and historic resources of the Land whether arising from the conduct of the Concession Activity or from natural causes such as earthquake, land slip, volcanic activity, flood, or arising in any other way, whether or not from any breach of the terms of this Concession on the part of the Concessionaire.
- 17.2 The Grantor may exercise its power to terminate under 17.1(h) without giving notice.
- 17.3 The Grantor may exercise the Grantor's right under this clause to terminate the Concession notwithstanding any prior waiver or failure to take action by the Grantor or any indulgence granted by the Grantor for any matter or default.
- 17.4 Termination of the Concession is not to prejudice or affect the accrued rights or claims and liabilities of the parties.

18. What happens on termination or expiry of the Concession?

18.1 On expiry or termination of this Concession, either as to all or part of the Land, the Concessionaire is not entitled to compensation for any structures or other improvements placed or carried out by the Concessionaire on the Land.

- 18.2 The Concessionaire may, with the Grantor's written consent, remove any specified structures and other improvements on the Land. Removal under this clause must occur within the time specified by the Grantor and the Concessionaire is to make good any damage and leave the Land and other public conservation land affected by the removal in a clean and tidy condition.
- 18.3 The Concessionaire must, if the Grantor gives written notice, remove any specified structures and other improvements on the Land. Removal under this clause must occur within the time specified by the Grantor and the Concessionaire is to make good any damage and leave the Land and other public conservation land affected by the removal in a clean and tidy condition and replant the Land with indigenous vegetation of a similar abundance and diversity as at the commencement of the Term. If before the expiry of the Term the Concessionaire makes an application for a further concession in respect of the same Concession Activity on the Land then the Grantor can not require such removal and reinstatement until such time as that concession application has been determined. If a new concession is granted then removal and reinstatement can not be required until the expiry or termination of the new concession.

19. When is the Grantor's consent required?

19.1 Where the Grantor's consent or approval is expressly required under this Concession then the Concessionaire must seek that approval or consent for each separate time it is required even though the Grantor may have given approval or consent for a like purpose on a prior occasion. Any such consent or approval may be made on such conditions as the Grantor considers appropriate.

20. Are there limitations on public access and closure?

20.1 The Concessionaire acknowledges that the Land is open to the public for access and that the Grantor may close public access during periods of high fire hazard or for reasons of public safety or emergency.

21. What about other concessions?

21.1 Nothing expressed or implied in this Concession is to be construed as preventing the Grantor from granting other concessions, whether similar or not, to other persons provided that the Grantor must not grant another concession that would derogate in any material way from the Concessionaire's ability to carry out the Concession Activity.

22. How will disputes be resolved?

- 22.1 If a dispute arises between the parties in connection with this Concession the parties must, without prejudice to any other rights or entitlements they may have, attempt to resolve the dispute by agreement using informal dispute resolution techniques such as negotiation, mediation, independent expert appraisal or any other alternative dispute resolution technique. The rules governing any such technique adopted are to be agreed between the parties.
- 22.2 If the dispute cannot be resolved by agreement within 14 days of written notice by one party to the other (or such further period as the parties may agree to in writing) either party may refer the dispute to the Disputes Tribunal, where relevant, or to arbitration, which arbitration is to be carried out in accordance with the provisions of the Arbitration Act 1996.

- 22.3 If the parties do not agree on an arbitrator within 10 working days of a party giving written notice of the requirement to appoint an arbitrator the President of the New Zealand Law Society is to appoint the arbitrator. In either case the arbitrator must not be a person who has participated in an informal dispute resolution procedure in respect of the dispute.
- 22.4 The arbitrator must include in the arbitration award reasons for the determination.
- 22.5 Despite the existence of a dispute, each party must continue to perform its obligations under this Concession.

23. What about prosecution for offences?

- 23.1 Where any breach of this Concession by the Concessionaire also constitutes an offence under the Resource Management Act 1991, the Conservation Act 1987, or any of the Acts listed in the First Schedule to that Act:
 - (a) no waiver or failure to act by the Grantor under this Concession is to preclude the Grantor from prosecuting the Concessionaire; and
 - (b) no failure by the Grantor to prosecute the Concessionaire is to preclude the Grantor from exercising the Grantor's remedies under this Concession; and
 - (c) any action of the Grantor in prosecuting the Concessionaire is not to preclude the Grantor from exercising the Grantor's remedies under this Concession.

24. How are notices sent and when are they received?

- 24.1 Any notice to be given under this Concession is to be in writing and made by personal delivery, fax, by pre paid post or email to the receiving party at the address, fax number or email address specified in Item 14 or 15 of Schedule 1. Any such notice is to be deemed to have been received:
 - (a) in the case of personal delivery, on the date of delivery;
 - (b) in the case of post, on the 3rd working day after posting;
 - (c) in the case of email, on the day the email was sent provided it is sent before 5 pm, otherwise on the next working day.
- 24.2 If any party's details specified in Item 14 or 15 of Schedule 1 change then the party whose details change must within 5 working days of such change provide the other party with the changed details.

25. What is the scope of the Concession?

25.1 Except as provided by legislation, this Concession and any written variation agreed by the parties contain the entire understanding between the parties with reference to the subject matter of this Concession and there is no other agreement, representation or warranty whether it is expressed or implied which in any way extends, defines or otherwise relates to the provisions of this Concession.

26. Can provisions be severed?

26.1 Any illegality, or invalidity or unenforceability of any provision in this Concession is not to affect the legality, validity or enforceability of any other provisions.

27. What about the payment of costs?

- 27.1 The Concessionaire must pay the Grantor's legal costs and expenses of and incidental to preparing and signing any extension or variation of this Concession.
- 27.2 The Concessionaire must pay in full immediately and on demand all costs and fees (including solicitor's costs and fees of debt collecting agencies engaged by the Grantor) arising out of and associated with steps taken by the Grantor to enforce or attempt to enforce the Grantor's rights and powers under this Concession including the right to recover outstanding money owed to the Grantor.

28. What is the relationship of parties?

- 28.1 Nothing expressed or implied in this Concession is to be construed as:
 - (a) constituting the parties as partners or joint venturers;
 - (b) conferring on the Concessionaire any right of exclusive occupation or use of the Land;
 - (c) granting any exclusive estate or interest in the Land to the Concessionaire;
 - (d) affecting the rights of the Grantor and the public to have access across the Land.

29. What about a Guarantee?

- 29.1 Where the Grantor has in Item 15 of Schedule 1 required this Concession to be guaranteed by a third party the following clauses are to apply.
- 29.2 In consideration of the Grantor entering into this Concession at the Guarantor's request the Guarantor:
 - (a) guarantees payment of the Concession Fee and the performance by the Concessionaire of the covenants in this Concession; and
 - (b) indemnifies the Grantor against any loss the Grantor might suffer should the Concession be lawfully disclaimed or abandoned by any liquidator, receiver or other persons.
- 29.3 The Guarantor covenants with the Grantor that:
 - (a) no release, delay, or other indulgence given by the Grantor to the Concessionaire, to the Concessionaire's successors or assigns, or any other thing whereby the Guarantor would have been released had the Guarantor been merely a surety is to release, prejudice, or affect the liability of the Guarantor as a Guarantor or as indemnifier;
 - (b) as between the Guarantor and Grantor the Guarantor may, for all purposes, be treated as the Concessionaire and the Grantor is under no obligation to take proceedings against the Concessionaire before taking proceedings against the Guarantor;

- (c) the guarantee is for the benefit of and may be enforced by any person entitled for the time being to receive the Concession Fee;
- (d) any assignment of this Concession and any Concession Fee Review in accordance with this Concession are not to release the Guarantor from liability;
- (e) should there be more than one Guarantor the liability of each Guarantor under this Guarantee is to be joint and several.

30. What about Co-Siting?

- 30.1 In this clause "Co-Site" means the use of the Concessionaire's structures or facilities on the Land by a third party for an activity; and "Co-Sitee" and "Co-Siting" have corresponding meanings.
- 30.2 The Concessionaire must not allow Co-Siting on the Land without the prior written consent of the Grantor.
- 30.3 The Grantor's consent must not be unreasonably withheld but is at the Grantor's sole discretion and subject to such reasonable terms and conditions as the Grantor thinks fit including a requirement that the Co-Sitee be liable for direct payment to the Grantor of a concession fee and any environmental premium assessed in respect of the Co-Sitee's activity on the Land.
- 30.4 In addition, the Grantor must withhold consent if:
 - (a) the Co-Siting would result in a substantial change to the Concession Activity on the Land; or
 - (b) the Grantor considers the change to be detrimental to the environment of the Land.
- 30.5 Subject to clause 30.4 the Concessionaire must, if required by the Grantor, allow Co- Siting on the Land.
- 30.6 Where the Concessionaire maintains that Co-Siting by a third party on the Land would:
 - (a) detrimentally interfere physically or technically with the use by the Concessionaire of the Land; or
 - (b) materially prejudice any resource consents obtained by the Concessionaire or cause more onerous conditions to be imposed on it by the relevant authority; or
 - (c) obstruct or impair the Concessionaire's ability effectively to operate from the Land; or
 - (d) interfere with or prevent future forecast works of the Concessionaire,

the Grantor, must, as a pre-condition to consideration of an application to grant a concession to a third party, require that third party to obtain, at its own cost, a report prepared by an independent consultant acceptable to the Grantor confirming or rejecting the presence of the matters specified in this clause 30.6. The Grantor must not grant a concession to a third party where the report confirms that the proposed concession would give rise to one or more of the matters specified in this clause 30.6.

30.7 If the independent consultant report rejects the Concessionaire's concerns, the Concessionaire may dispute this in accordance with the procedure set out in clause 22 of Schedule 2.

- 30.8 Where the Concessionaire is required under clause 30.5 to allow Co-Siting on the Land, the Concessionaire is, subject to clause 30.10 entitled to enter into commercial agreements with third parties for them to conduct an activity on the Land and to receive a reasonable fee from them for any agreed activity they intend to carry out on the Land. If a dispute arises between the Concessionaire and a third party such dispute must be determined by the Grantor having regard to, but not limited to, the following matters:
 - (a) any written comments or submissions of the Concessionaire and third party;
 - (b) market value for the concession activity proposed by the third party having regard to the matters specified in Section 17Y(2) of the Conservation Act 1987;
 - (c) any other matters the Grantor considers relevant.
- 30.9 If the Concessionaire does not accept the Grantor's determination, the Concessionaire may dispute this in accordance with the procedure set out in clause 22 of Schedule 2.
- 30.10 For the avoidance of doubt, a Co-Sitee permitted on the Land must enter into a separate concession with the Grantor in terms of which the Co-Sitee may be required to pay to the Grantor a concession fee and environmental premium assessed in respect of the Co-Sitee's activity on the Land. This separate concession must not contain provisions that conflict with the Concessionaire's rights and obligations in relation to the Land.
- 30.11 The Grantor must not authorise the third party to commence work on the Land until all relevant resource consents are issued, an agreement is executed between the Concessionaire and third party, and any conditions imposed by the Concessionaire have been met.

31. What about Identification cards?

- 31.1 Before commencing the Concession Activity the Concessionaire must, if required by the Grantor in Item 12 of Schedule 1, obtain Concessionaire Identification cards from the Grantor. The Grantor is to supply such cards to the Concessionaire on a cost recovery basis.
- 31.2 The Concessionaire and any person acting under the authority of the Concession must carry and display a Concession Identification card when carrying out the Concession Activity.
- 31.3 The Concessionaire must obtain sufficient cards to ensure all people acting under the authority of the Concession can carry and display such cards when undertaking the Concession Activity.
- 31.4 The Concessionaire may also access, use and/ or display the Grantor's "Approved Label". This right only exists once the Concessionaire agrees to comply with the Grantor's Approved Label terms and conditions and while the Concession remains operative. When the Concessionaire so requests the Grantor is to forward the Concessionaire an electronic link to the Approved Label. This electronic link is to contain the Approved Label terms and conditions.

Godley Peaks Pastoral Lease - telecommunication concession - May 2015 - DOC-2331607 Previous template docdm-1408400

31.5 The right under this clause 31.4 does not affect the obligation in this clause 31 to carry and display a Concession Identification card.

32. Which clauses survive termination?

32.1 Clauses 12 and 24 survive the termination of this Concession.

33. When can the conditions of the Concession be varied?

- 33.1 The Grantor may on each Concession Fee Review Date, after first consulting with the Concessionaire, vary any condition of this Concession to make the condition more effective in addressing adverse effects resulting from the Concession Activity.
- 33.2 Nothing in clause 33.1 otherwise affects the Grantor's rights to vary the Concession under section 17ZC of the Conservation Act 1987.

34. Are there any Special Conditions?

34.1 Special conditions are specified in Schedule 3.

35. The Law

35.1 This Concession is to be governed by, and interpreted in accordance with the laws of New Zealand

SCHEDULE 3

SPECIAL CONDITIONS

1. The Concessionaire must ensure any persons employed by the Concessionaire are requested to recognise and provide for Ngāi Tahu values in the conduct of their activities.

Didymo

2. The Concessionaire must comply with all guidelines and notices put out by Biosecurity New Zealand regarding measures to avoid spreading the pest organism Didymosphenia geminate ("Didymo"), and or any other pest organism identified during the term of this Concession.

Special conditions

- 3. The Grantor acknowledges that the telecommunication equipment to be located on the site is the property of the Concessionaire.
- 4. The Concessionaire shall at all times keep the site in a tidy condition.
- 5. Aircraft landings are **only** permitted on the Land for purposes associated with operating maintaining and repairing the telecommunications facility.
- 6. The Concessionaire shall ensure that any aircraft attending the site and/or any machinery or equipment arriving at the site are clean and free of weeds or potential weed sources.
- 7. The Concessionaire shall ensure that any associated structures are of a colour consistent with the surrounding environment to the satisfaction of the relevant Manager, Twizel Office, Department of Conservation, or the closest Department office.
- 8. If the Grantor determines that the conditions of this Document or the effects of Concession Activity should be monitored, the Concessionaire shall meet: either the full costs of any monitoring programme that is implemented; or, if the Grantor determines that the costs should be apportioned among several Concessionaires who use the same locations, part of the costs of the monitoring programme. These costs will include the Department's standard charge-out rates for staff time and the mileage rates for vehicle use associated with the monitoring programme.

Concession Number:

SCHEDULE 4

Designations Plan or map Any further plan if required Godley Peaks Pastoral Lease - telecommunication concession - May 2015 - DOC-2331607 Previous template docdm-1408400

- 23 -

Appendix 9: Easement to be created over the freehold (DOCDM-1580421 (1))

TRANSFER GRANT OF EASEMENT IN GROSS

For Public Access and Management Purposes

Land Transfer Act 1952

This page does not form part of the Transfer.

RELEASED UNDER THE OFFICIAL INFORMATION ACT

TRANSFER

Land Transfer Act 1952

If there is not enough space in any of the panels below, cross-reference to and use the approved Annexure Schedule: no other format will be received.

Land Registration District

Certificate of Title No. A		All or Part?	Area and legal description - Insert only when part or Stratum, CT				
			· · · ·				

Transferor Surnames must be underlined

<u>COMMISSIONER OF CROWN LANDS</u>, acting pursuant to section 80 of the Crown Pastoral Land Act 1998

Transferee Surnames must be underlined

HER MAJESTY THE QUEEN, acting by and through the Minister of Conservation

Estate or Interest or Easement to be created: Insert e.g. Fee simple; Leasehold in Lease No.; Right of way etc.

Public Access and Management Purposes Easement in Gross under Section 7(2) of the Conservation Act 1987 (continued on pages 2, 3 and 4 of Annexure Schedule).

Consideration

The various considerations set out in a substantive proposal accepted under the Crown Pastoral Land Act 1998 on the day of

Operative Clause

For the above consideration (receipt of which is acknowledged) the GRANTOR TRANSFERS to the GRANTEE all the grantor's estate and interest in the land in the above Certificate(s) of Title and if an easement is described above such is granted or created.

Dated this day of

Attestation

Signed by acting under written delegation from the Commissioner of Crown Lands	Signed in my presence by the Grantor Signature of Witness Witness to complete in BLOCK letters (unless typewritten or legibly stamped) Witness name Occupation Address	(continued on page 7 of Annexure Schedule)
Signature. or common seal of Grantor	İ	

Certified correct for the purposes of the Land Transfer Act 1952

Annexure Schedule

Insert below "Mortgage", "Transfer", "Lease", etc

		 Deve	- 4	Deces
Transfer Easement	Dated	Page	ot	Pages

Definitions

- 1. In this transfer unless the context otherwise requires:
 - 1.1 "Easement Area" means that part of the Servient Land being 14 metres wide which is marked "[]" on Deposited Plan/S.O. Plan No []].
 - 1.2 "Management Purposes" means:
 - the protection of a significant inherent value of any land managed by the Grantee;
 - the ecological sustainable management of any land managed by the Grantee;
 - the management of the Easement Area consistent with the purposes for which the easement is held under the Conservation Act 1987 or the Reserves Act 1977.
 - 1.3 "Servient Land" means the land owned by the Grantor and described on page 1.
 - 1.4 "Grantee" means Her Majesty the Queen acting by and through the Minister of Conservation and includes tenants, agents, invitees, contractors, licensees and employees of the Minister of Conservation and the Director-General of Conservation; and for the purposes of clause 2.1 only, includes any member of the public.
 - 1.5 "Grantor" means the owner of the Servient Land described on page 1 and includes the Grantor's tenants and invitees. An invitee is somebody specifically invited onto the land by Her Majesty the Queen acting by and through the Minister of Conservation for Management Purposes.
 - 1.6 "Working day" means the period between any one midnight and the next, excluding Saturdays, Sundays and statutory holidays in the place where the Servient Land is located.

Standard Easement Terms

<u>Access</u>

- 2. The Grantee has the right in common with the Grantor:
 - 2.1 To pass and re-pass at any time over and along that part of the Easement Area marked "a-b", "c-d", "b-n", "o-p" and "q-r" on SO Plan ______ on foot, or by non-motorised vehicle powered by a person or persons.
 - 2.2 To pass and re-pass at any time over and along that part of the Easement Area marked ""a-b", "c-d", "e-f", "g-h", "g-i", "j-k", "I-m" "b-n", "o-p" and "q-r" on SO Plan ________ on foot, or on or accompanied by horses, or by non-motorised vehicle, or by motor vehicle, with or without machinery and implements of any kind, with or without guns and accompanied by dogs, for Management Purposes.
- 3. The Grantor must keep the Easement Area clear at all times of obstructions whether caused by parked vehicles, deposit of materials or unreasonable impediment to the use and

If this Annexure Schedule is used as an expansion of an instrument, all signing parties and either their witnesses or their solicitors must put their signatures or initials here.

Godley Peaks easement 2015 - DOCDM-1580421 (1)

enjoyment of the Easement Area, where such event or outcome is caused by or under the control of the Grantor.

Exclusion of Schedules

4. The rights and powers contained in Schedule 4 of the Land Transfer Regulations 2002 and the Fifth Schedule of the Property Law Act 2007 are expressly negatived.

<u>Term</u>

5. The easement is to be in perpetuity.

Temporary Closure of Easement Area

- 6.1 The Grantee may, at any time in exercise of her/his powers, temporarily close all or part of the Easement Area to the public for such period as she/he considers necessary.
- 6.2 For the avoidance of doubt, it is stated that any such temporary closure by the Grantee of all or part of the Easement Area to the public in accordance with clause 6.1 does not affect the rights of the Grantor to continue to access and use the Easement Area.

Dispute Resolution

- 7.1 If a dispute arises between the Grantor and Grantee concerning the rights, management and operation created by this transfer the parties are to enter into negotiations in good faith to resolve it.
- 7.2 If the dispute is not resolved within 14 days of written notice by one party to the other it is to be referred to mediation.
- 7.3 If the dispute is not resolved within 21 days or such other period as agreed to in writing between the parties after the appointment of the mediator, the parties must submit to the arbitration of an independent arbitrator appointed jointly by the parties or, if one cannot be agreed within 14 days, to an independent arbitrator appointed by the President for the time being of the New Zealand Law Society.
- 7.4 The arbitration is to be determined in accordance with the Arbitration Act 1996 and its amendments or any enactment passed in substitution.

If this Annexure Schedule is used as an expansion of an instrument, all signing parties and either their witnesses or their solicitors must put their signatures or initials here.

Godley Peaks easement 2015 - DOCDM-1580421 (1)

Notice

- 8.1 A notice to be given under this transfer by one party to the other is to be in writing and must:
 - (a) be hand delivered to the receiving party; or
 - (b) be sent by ordinary post to the receiving party; or
 - (c) be sent by email to the receiving party.
- 8.2 If clause 8.1(b) applies the notice will be deemed to be received by the receiving party on such date on which the ordinary post would be delivered.
- 8.3 If clause 8.1(c) applies the notice will be deemed to have been received on the day on which it is emailed if that day is a working day or, if dispatched after 5.00pm, on the next working day after the date of email.

Gates

- 9. Where the Grantor wishes to erect fences across the Easement Area, the Grantor shall install a gate not less than 3 metres wide, and either:
 - 9.1 Keep the gate unlocked at all times, or
 - 9.2 Ensure the Grantee is provided with a key to the gate; and
 - 9.3 Install a stile or appropriate facility to allow for public access by foot, horse or nonmotorised vehicle.
- 10. The Grantee (not being a member of the public) has the right:
 - 10.1 To mark the Easement Area as appropriate.
 - 10.2 To erect and maintain stiles and/or gates.
 - 10.3 To erect and maintain signs informing the public:
 - (a) of the location of the land managed by the Crown and available for public access and recreation; and
 - (b) of their rights and responsibilities in relation to the Easement Area.
 - 10.4 From time to time to modify the surface of the Easement Area so that it becomes and remains fit for the purpose of clauses 2.1 and 2.2.
 - 10.5 To use whatever reasonable means of access he/she thinks fit over the Easement Area to carry out the works in clause 10.1 to 10.4.

If this Annexure Schedule is used as an expansion of an instrument, all signing parties and either their witnesses or their solicitors must put their signatures or initials here.

Godley Pcaks easement 2015 - DOCDM-1580421 (1)

Special Easement Terms

- 11. The standard easement terms contained above must be read subject to any special easement terms set out below.
- 12. If the Grantee (being a member of the public) has a hunting permit, issued by the Director-General of Conservation for public conservation land to which the Easement Area provides access, they may carry a gun on the Easement Area for the purpose of gaining access to hunt on that land.
- 13. When the public is using the Easement Area in accordance with clause 2.1, no dogs are permitted.
- 14. When the Easement Area is to be used in accordance with clause 2.2, the Grantee will make her best efforts to give the Grantor notice prior to her use.
- 15. If the Grantor considers that for reasons of public safety or emergency, a temporary closure of the Easement Area or any part thereof is necessary, the Grantor shall advise the Grantee who may exercise her powers under clause 6.1. At the request of the Grantee, the Grantor may provide temporary signage indicating the Easement Area, or any part thereof, is temporarily closed.
- 16. The Grantor may close that part of the Easement Area marked "a-b" for stock management purposes, being the droving of stock along the easement only. Any stock management closures must be for no more than <u>four</u> hours barring any unforeseen circumstances. Prior to any closure for stock management purposes, the Grantor is to give the Grantee at least 48 hours notice of the closure and provide public information signage relating to the closure at both ends of that part of the Easement Area. Should emergency stock movements be required, e.g. for imminent severe weather or fire, on that part of the Easement Area marked "a-b" the Grantor shall give the Grantee as much notice as is reasonably practicable and provide public information signage about the closure at both ends.
- 17. In the event that that there is significant rock fall from the slopes adjoining Mistake Peak which causes obstruction of that part of the Easement Area marked "a-b" or a permanent shift in the course of the Cass River which causes degradation to the part of the Easement Area marked "c-d" which is immediately adjacent to the river bank at the date of this easement instrument (collectively called the "Affected Easement Areas") or to land adjoining the Affected Easement Areas such that clauses 2.1 and 2.2 cannot be satisfied, the parties are agreed that:
 - (a) in the event that it is the part of the Easement Area marked "a-b" that is affected, the Grantor may attempt to fix or recreate the track to a standard sufficient for farming operations by using a bulldozer and stabilising the ground up to 100 metres above that part of the Affected Easement Area.
 - (b) in respect of any part of the Affected Easement Areas, if the part of the Easement Area ceases to adequately provide access as required by clauses 2.1 and 2.2, then this Easement will be partially surrendered in relation to the part of the Affected Easement Areas, and
 - (c) the Grantor will grant a right of way to the Grantee on the same terms as are herein contained over the adjoining land of the Grantor ("New Easement").
- 18. In the event that any of the Management Easement Areas are modified away from their original alignment at the date of this easement instrument (collectively the "Management Affected Easement Areas") or to the land adjoining the Management Affected Easement Areas, such that clause 2.2 cannot be satisfied, the parties are agreed that:
 - (a) if part of the Management Affected Easement Area ceases to adequately provide access as required by clause 2.2, then this Easement will be partially surrendered in relation to the part of the Management Affected Easement Areas, and

If this Annexure Schedule is used as an expansion of an instrument, all signing parties and either their witnesses or their solicitors must put their signatures or initials here.

Godley Peaks easement 2015 - DOCDM-1580421 (1)

(b) the Grantor will grant a right of way to the Grantee on the same terms as are herein contained over the adjoining land of the Grantor ("New Management Easement").

)

- 19. That part of the Easement Area marked "g-h" is closed to the Grantee for the period between the 15th of October and the 1st of December inclusive if required for lambing purposes.
- 20. In the event the Grantee (not being a member of the public) wishes to use the Easement Area and part of it is temporarily obstructed (e.g. by irrigation machinery or because of stock yard use), the Grantee has the right to bypass the temporary obstruction by using an alternative route on the Servient Land which is as close as reasonably practicable to the Easement Area.

Continuation of "Attestation"

Signed for and on behalf of	
Her Majesty the Queen by	
[name]	
under a written delegation in the	
presence of:	

Witness (Signature)

Name

Address

Occupation

Footnote: In substitution of the SO Plan (which has yet to be prepared), the proposed easement described in clause 1 is marked on the Plan.

If this Annexure Schedule is used as an expansion of an instrument, all signing parties and either their witnesses or their solicitors must put their signatures or initials here.

Godley Peaks easement 2015 - DOCDM-1580421 (1)

TRANSFER GRANT OF EASEMENT IN GROSS

For Public Access and Management Purposes

Land Transfer Act 1952

Law Firm Acting

Solicitor Legal Services Department of Conservation Christchurch/Dunedin

! _____

> This page is for Land Registry Office use only. (except for "Law Firm Acting")

Appendix 10: Covenant to be created over the freehold – CC1 (DOCDM-1599758 (1))

DATED _____

Between

COMMISSIONER OF CROWN LANDS Pursuant to Section 80 of the Crown Pastoral Land Act 1998

and

MINISTER OF CONSERVATION ("the Minister")

COVENANT UNDER RESERVES ACT 1977 FOR CROWN PASTORAL LAND ACT 1998 PURPOSES



Godley Peaks Pastoral Lease - Conservation Covenant - Reserves Act - May 2015 - DOCDM-1599758 (1) DM1599758 Godley Peaks Conservation Covenant - Reserves Act - May 2015 Jan 2013

- 1 -

THIS DEED of COVENANT is made the day of

BETWEEN COMMISSIONER OF CROWN LANDS acting pursuant to section 80 of the Crown Pastoral Land Act 1998

AND MINISTER OF CONSERVATION

BACKGROUND

- A. The Commissioner of Crown Lands is deemed for the purposes of section 77 of the Reserves Act 1977 to be the owner of the Land under section 80(5) of the Crown Pastoral Land Act 1998.
- B. The Land contains certain Values specified in Schedule 1.
- C. The parties agree that the Land should be managed so as to preserve the particular Values specified in Schedule 1, and that such purpose can be achieved without the Minister acquiring a fee simple or leasehold interest in the Land.
- D. An approved plan designating the Land as land over which a Covenant under section 77 of the Reserves Act 1977 is to be created has been registered under section 64 of the Crown Pastoral Land Act 1998.
- E. The Commissioner of Crown Lands has agreed to grant the Minister a Covenant over the Land to preserve the particular Values specified in Schedule 1.

OPERATIVE PARTS

In accordance with section 77 of the Reserves Act 1977, and with the intent that the Covenant run with the Land and bind all subsequent Owners of the Land, the Commissioner of Crown Lands and the Minister agree as follows:

1. INTERPRETATION

1.1 In this Covenant unless the context otherwise requires:

"Act"	means the Reserves Act 1977.		
"Covenant"	means this Deed of Covenant made under section 77 of the Act.		
"Director-General"	means the Director-General of Conservation.		
"Fence"	includes a gate.		
"Fire Authority"	means a Fire Authority as defined in the Forest and Rural Fires Act 1977.		
"Land"	means the land described in Schedule 1.		
"Minerals"	means any mineral that is a Crown owned mineral under section 2 of the Crown Minerals Act 1991.		
"Minister"	means the Minister of Conservation.		
"Natural Water"	includes water contained in streams the banks of which have, from time to time, been realigned.		

"Owner" means the person or persons who from time to time is or are registered as the proprietor(s) of the Land.

"Party" or "Parties" means either the Minister or the Owner or both.

"Values" means any or all of the Land's natural environment, landscape amenity, wildlife, freshwater life, marine life habitat or historic values as specified in Schedule 1.

"Working Day" means the period between any one midnight and the next excluding Saturdays, Sundays, and statutory holidays in the place where the Land is located.

1.2 For avoidance of doubt:

- 1.2.1 the reference to any statute in this Covenant extends to and includes any amendment to or substitution of that statute;
- 1.2.2 expressions defined in clause 1.1 bear the defined meaning in the whole of this Covenant including the Background. Where the parties disagree over the interpretation of anything contained in this Covenant and in determining the issue, the parties must have regard to the matters contained in the Background;
- 1.2.3 clause and other headings are for ease of reference only and are not to be treated as forming any part of the context or to affect the interpretation of this Covenant;
- 1.2.4 words importing the singular number include the plural and vice versa;
- 1.2.5 words importing one gender include the other gender;
- 1.2.6 any obligation not to do anything must be treated to include an obligation not to suffer, permit or cause the thing to be done;
- 1.2.7 the agreements contained in this Covenant bind and benefit the parties and their administrators and executors, successors and assigns in perpetuity.

2. OBJECTIVE OF THE COVENANT

2.1 The Land must be managed so as to preserve the Values.

3. THE OWNER'S OBLIGATIONS

- 3.1 Unless agreed in writing by the parties, the Owner must not carry out or allow to be carried out on or in relation to the Land:
 - 3.1.1 grazing of the Land by livestock;
 - 3.1.2 subject to clauses 3.2.1 and 3.2.3, felling, removal or damage of any tree, shrub or other plant;
 - 3.1.3 the planting of any species of tree, shrub or other plant;
 - 3.1.4 the erection of any Fence, building, structure or other improvement for any purpose;

- 3.1.5 any burning, chemical spraying, top dressing or sowing of seed;
- 3.1.6 any cultivation, earth works or other soil disturbances;
- 3.1.7 any archaeological or other scientific research involving disturbance of the soil;
- 3.1.8 the damming, diverting or taking of Natural Water;
- 3.1.9 any action which will cause deterioration in the natural flow, supply, quantity, or quality of water of any stream, river, lake, pond, marsh, or any other water resource affecting the Land;
- 3.1.10 any other activity which might have an adverse effect on the Values;
- 3.1.11 any prospecting or mining for Minerals, coal or other deposit or moving or removal of rock of any kind on or under the Land;
- 3.1.12 the erection of utility transmission lines across the Land.
- 3.2 The Owner must:
 - 3.2.1 eradicate or control all weeds and pests on the Land to the extent required by any statute; and in particular comply with the provisions of, and any notices given under, the Biosecurity Act 1993;
 - 3.2.2 if it is safe to do so, assist the Fire Authority to extinguish any wildfire upon or threatening the Land;
 - 3.2.3 keep the Land free from exotic tree species;
 - 3.2.4 keep the Land free from rubbish or other unsightly or offensive material arising from the Owner's use of the Land;
 - 3.2.5 grant to the Minister or authorised agent of the Minister or any employee or contractor of the Director-General, a right of access at all times on and to the Land, with or without motor vehicles, machinery, and implements of any kind, for purposes associated with the management of this Covenant;
 - 3.2.6 keep all Fences on the boundary of the Land in good order and condition and, notwithstanding clause 3.1.4, rebuild or replace all such Fences when reasonably required except as provided in clause 4.2.

4. THE MINISTER'S OBLIGATIONS

- 4.1 The Minister must have regard to the objective specified in clause 2.1 when considering any requests for approval under this Covenant.
- 4.2 The Minister must repair and replace to its former condition any Fence or other improvement on the Land or on its boundary which may have been damaged in the course of the Minister or any person referred to in clause 3.2.5 exercising any of the rights conferred by this Covenant.

5. IMPLEMENTATION OF OBJECTIVES

5.1 The Minister may;

- 5.1.1 provide to the Owner technical advice or assistance as may be necessary or desirable to assist in meeting the objectives specified in clause 2.1;
- 5.1.2 prepare, in consultation with the Owner, a joint plan for the management of the Land to implement the objectives specified in clause 2.1;
- 5.1.3 prepare, in consultation with the Owner, a monitoring plan to assist the parties to meet the objectives specified in clause 2.1.

6. DURATION OF COVENANT

6.1 This Covenant binds the Minister and Owner in perpetuity to the rights and obligations contained in it.

7. OBLIGATIONS ON SALE, ASSIGNMENT OR OTHER DEPOSAL OF LAND

- 7.1 If the Owner sells, leases, or parts with possession of the Land, or hands over control of the Land to any other person, the Owner must ensure that the Owner obtains the agreement of the purchaser, lessee, assignee or manager to comply with the terms of this Covenant, including any agreement by the purchaser, lessee, assignee or manager to ensure that on any subsequent sale, lease, assignment, or change in control of the Land, any subsequent purchaser, lessee, assignee or manager must also comply with the terms of this Covenant including this clause.
- 7.2 A Transferee of the land will at law be bound by the registered Covenant. Such transfer is deemed to provide the agreement to comply with the terms of this covenant required by Clause 7.1.

8. MISCELLANEOUS MATTERS

8.1 Rights

8.1.1 The rights granted by this Covenant are expressly declared to be in the nature of a covenant.

8.2 **Trespass Act:**

- 8.2.1 Except as provided in this Covenant, the Covenant does not diminish or affect the rights of the Owner to exercise the Owner's rights under the Trespass Act 1980 or any other statute or generally at law or otherwise;
- 8.2.2 For avoidance of doubt these rights may be exercised by the Owner if the Owner reasonably considers that any person has breached the rights and/or restrictions of access conferred by this Covenant.

8.3 Reserves Act

8.3.1 Subject to the terms and conditions set out in this Covenant, sections 93 to 105 of the Reserves Act 1977, as far as they are applicable and with the necessary modifications, apply to the Land as if the Land were a reserve, notwithstanding that the Land may from time to time be sold or otherwise disposed of.

8.4 Titles

- 5 -
- 8.4.1 This Covenant must be signed by the Commissioner of Crown Lands and the Minister and registered against the Certificate of Title to the Land.

8.5 Acceptance of Covenant

8.5.1 The parties agree to be bound by the provisions of this Covenant including during the period prior to the Covenant's registration.

8.6 Fire

- 8.6.1 The Owner must notify, as soon as practicable, the appropriate Fire Authority and the Minister in the event of wild fire threatening the Land;
- 8.6.2 If the Minister is not the appropriate Fire Authority for the Land, the Minister will render assistance to the Fire Authority in suppressing the fire if:
 - 8.6.2.1 requested to do so; or
 - 8.6.2.2 if there is in place between the Minister and the Fire Authority a formalised fire agreement under section 14 of the Forest and Rural Fires Act 1977.

9. NOTICES

- 9.1 A notice to be given under this Covenant by one party to the other is to be in writing and made by personal delivery, by pre-paid post, or by email addressed to the receiving party at the address or email address set out in Schedule 1.
- 9.2 A notice given in accordance with clause 9.1 will be deemed to have been received:
 - (a) in the case of personal delivery, on the date of delivery;
 - (b) in the case of pre-paid post, on the third Working Day after posting;
 - (c) in the case of email, on the day on which it is dispatched if that is a Working Day or, if it is not a Working Day or if it is dispatched after 5.00pm, on the next Working Day after the date of dispatch.
- 9.3 The Owner must notify the Minister of any change of ownership or control of all or part of the Land and must supply the Minister with the name and address of the new owner or person in control.

10. DEFAULT

- 10.1 Where either the Minister or the Owner breaches any of the terms and conditions contained in this Covenant the other party:
 - 10.1.1 may take such action as may be necessary to remedy the breach or prevent any further damage occurring as a result of the breach; and
 - 10.1.2 will also be entitled to recover from the party responsible for the breach as a debt due all reasonable costs (including solicitor/client costs) incurred by the other party as a result of remedying the breach or preventing the damage.
- 10.2 Should either the Minister or the Owner become of the reasonable view that the other party (the defaulting party) has defaulted in performance of or observance of its obligations under this Covenant then that party (notifying party) may, by written notice:

- 10.2.2 state the action reasonably required of the defaulting party to perform or observe in accordance with this Covenant; and
- 10.2.3 state a reasonable period within which the defaulting party must take action to remedy the default.

11. DISPUTE RESOLUTION PROCESSES

11.1 If any dispute arises between the Minister and the Owner in connection with this Covenant, the parties must, without prejudice to any other rights they may have under this Covenant, attempt to resolve the dispute by negotiation or other informal dispute resolution technique agreed between the parties.

11.2 Mediation

- 11.2.1 if the dispute is not capable of resolution by agreement within 14 days of written notice by one party to the other (or such further period as the parties may agree to in writing) either party may refer the dispute to mediation with a mediator agreed between the parties;
- 11.2.2 if the parties do not agree on a mediator, the President of the New Zealand Law Society is to appoint the mediator.

11.3 Failure of Mediation

- 11.3.1 in the event that the dispute is not resolved by mediation within 2 months of the date of referral to mediation the parties agree that the provisions in the Arbitration Act 1996 will apply;
- 11.3.2 notwithstanding anything to the contrary in the Arbitration Act 1996, if the parties do not agree on the person to be appointed as arbitrator, the appointment is to be made by the President of the New Zealand Law Society;
- 11.3.3 the parties further agree that the results of arbitration are to be binding upon the parties.

12. JOINT OBLIGATIONS

12.1 The Owner or the Minister may, by mutual agreement, carry out any work or activity or improvements or take any action either jointly or individually to better preserve the Values.

13. FURTHER AGREEMENT

13.1 Where clauses in this Covenant require further agreement between the parties such agreement must not be unreasonably withheld.

14. SPECIAL CONDITIONS

- 14.1 Special conditions relating to this Covenant are set out in Schedule 2.
- 14.2 The standard conditions contained in this Covenant must be read subject to any special conditions.

Executed as a Deed

Signed by actin	ng under a)
delegation from the Commissioner of Ca	own Lands)
deemed pursuant to section 80(5) of the	Crown Pastoral)
Land Act 1998 to be the Owner of the L	and for the)
purposes of section 77 of the Reserves A	ct 1977)
in the presence of :)

Witness:	
----------	--

Address :

Occupation: _____

Signed by [name])
)
exercising his/her)
powers under section 117 of the Reserves Act 1977)
as designated Commissioner and acting for and on)
behalf of the Minister of Conservation)
in the presence of :)

Witness:

Address :

Occupation: _____

SCHEDULE 1

1. Description of Land

CC1 - Godley River Flats and Lower Faces.

All those pieces of land totalling 1269 hectares approximately labelled CC1 and shown shaded yellow on the designations plans attached to the Proposal.

2. Address for Service¹

The address for service (including email address) of the Minister is:

Minister of Conservation C/- Director Conservation Partnerships South and Eastern South Island Region Department of Conservation Conservation House 77 Stuart Street P O Box 5244 **DUNEDIN** PH: (03) 477 0677 E-mail: dunedinvc@doc.govt.nz

The address for service (including email address) of the Owner is: Verity Farms NZ Limited C/O Milnes Beatson Ltd 29 Wallace Street **MOTUEKA** 7120 Ph: 03 528-8760 Fax: 03 528-8762 Email: david@mba.net.nz

> Director: Ann Poindexter 210 Patons Rock Road RD 2 **TAKAKA 7182** Ph: 021 531-227 Email: ann@verityfarms.co.nz

¹ State street address not Post Office Box number.

Godley Peaks Pastoral Lease - Conservation Covenant - Reserves Act - May 2015 - DOCDM-1599758 (1) DM1599758 Godley Peaks Conservation Covenant - Reserves Act - May 2015

÷

3. Values of the Land to be Protected and Preserved (section 77 Reserves Act 1977)

<u>CC1 – Godley River Flats and Lower Faces– Landscape Amenity, Natural Environment, Wildlife</u> Habitat and Freshwater-life Habitat.

• The Land supports the threatened or uncommon plant species Olearia bullata, Coprosma intertexta and kowhai.

- The Land supports the threatened bird species black stilt, black-fronted tern, wrybill, banded dotterel and marsh crake and provides breeding sites for these species.
- The Land supports the threatened fish species upland longjaw galaxias and contains important aquatic habitats that support native fish species.
- The Land supports extensive red tussock wetland and tall matagouri shrubland, along with scattered shrublands on the hill slopes and gullies, which are representative of the original vegetation and contribute to the altitudinal sequence of plant species from the valley floor to the top of the Hall Range.
- The Land contains a largely unmodified wetland with intact hydrology that links to the Godley river floodplain.
- The Land is listed as a Wetland of Ecological and Regional Importance site and a Site of Significant Wildlife Interest of "outstanding" value for wildlife.
- The Land contains significant landscape values due to the high degree of naturalness, intactness, coherence and legibility alongside the Godley River.
- The Land forms an integral part of the high natural quality and integrity of the broad Godley River floodplain with the lower Range slopes contributing to the dramatic natural character of the mountain range.

- 10 -

SCHEDULE 2

Special Conditions

- 1 Notwithstanding the provisions of clause 3.1 the following shall apply;
 - 1.1 The Land may be grazed at any time by sheep and cattle only.
 - 1.2 The Owner may clear up to 30 hectares of matagouri per year within the Land with the use of chemical spraying or mechanical vegetation clearance. Matagouri clearance is restricted to areas of matagouri with an average canopy height of less than 1.5 metres tall.
 - 1.3 The Owner may, with the prior written consent of the Minister, plant shelterbelts within the Land. The Owner may not plant shelterbelts with species which the Minister considers have the potential to spread as wilding plant pests.
 - 1.4 The Owner may do routine maintenance of existing fences and structures and maintenance within the alignment of all existing tracks within the Land. Any maintenance undertaken outside the existing alignment or further upgrading of tracks requires the prior written consent of the Minister.
 - 1.5 The Owner may top dress and sow seed within the Land except for a 20 metre margin along the edge of any river or stream.

2 The Minister may design and undertake a monitoring programme:

- (a) to ensure that the ecological integrity of the Land is maintained; and
- (b) to enable the monitoring of any effects on the vegetation cover and conditions, faunal values and any other Values of the Land.
- 2.1 An initial possible monitoring programme for the Land is included in Schedule 3. This sets out the parties' respective responsibilities for undertaking monitoring, meeting costs, methods to be used and information requirements.
- 2.2 Should the results of monitoring show in the Minister's opinion a deterioration in the condition and extent of the ecological integrity of the Land, and/or the vegetative cover and conditions, faunal values or any other Values of the Land, the Minister reserves the right to take any necessary steps to protect species and/ or ecosystems or require the Owner to take such steps, including those further described under "monitoring results" in Schedule 3.
- 2.3 The Minister will liaise with the Owner in implementing any necessary steps, and the Owner will co-operate in giving effect to such steps as are considered necessary by the Minister.
- 3 The Minister may install a hut for the purpose of public shelter and safety within the Land. The hut is to be located at a site which is mutually agreed to between the Owner and the Minister and may include the potential to upgrade an existing farm hut on the Land for this purpose.

- 11 -

SCHEDULE 3

DESCRIPTION OF THE POSSIBLE MONITORING PROGRAMME TO BE ESTABLISHED.

1. Responsibilities:

An indigenous vegetation monitoring programme may be established on the Land by the Minister following the commencement of the covenant term. Re-monitoring will occur the year following the commencement of monitoring and then every 5 years after that. The re-monitoring is to be organised by the Owner with the assistance of the Minister.

The monitoring is intended to be a management tool, allowing the Minister and Owner to make informed decisions about the continued management of the land. This tool will develop and implement a system for assessing the impact of sheep and cattle grazing on the indigenous plant populations, species, communities, and ecosystems on the covenant of Godley Peaks Station.

The objectives are:

- 1. Establish a series of monitoring plots or points focussing on areas where stock grazing is likely to be most pronounced.
- 2. Use these points to establish baseline information about the indigenous plant populations, species, communities, and ecosystems the covenants of Godley Peaks.
- 3. After the first re-monitoring, at 5 yearly intervals, re-monitoring each of the plots or points and use this data to compare with previous monitoring occasions.

The Minister will be party to the re-monitoring by providing one staff member to assist with the physical monitoring. The Minister will be consulted as to the selection of a suitably qualified monitoring provider (which does not preclude the Owner undertaking this work to an acceptable standard). The Minister will be given a copy of the monitoring report in a format nominated by the Minister.

2. Costs:

The Minister will be responsible for the cost of establishing the monitoring. The Owner will be responsible for the cost of repeat monitoring and the report write up. The Minister will cover his/her own staff cost in assisting re-monitoring.

3. Monitoring Methods:

Photopoint Monitoring

A series of general repeatable photo-point sites will be established. The purpose of these photo points is to detect deterioration of the tussock, wetlands and shrublands being recorded as a consequence of stock impacts and other management practices.

This will be done using permanent landscape photo-points. (The number is to be decided by the Minister when setting up the monitoring). The location of each photo-point is to be recorded using GPS, and the pictures analysed to determine relative cover of tussock, *Hieracium*, rock/soil, and intertussock species. Site selection will occur when the monitoring is first established.

Within the Land photo points will consist of a series of general landscape photos to ensure that conditions of the covenant are complied with along with specific monitoring that may identify detrimental impacts as follows:

- within shrublands will include:
 - Observations of stock damage to the vegetation, fragmentation, tracking, gaps and canopy breakdown and impacts on regeneration, along with any loss of biodiversity that may occur as a direct result of grazing.
- within wetlands will include:
 - Observations of stock damage to the vegetation, fragmentation, pugging, tracking and impacts on regeneration, along with any loss of biodiversity that may occur as a direct result of grazing.

Godley Peaks Pastoral Lease - Conservation Covenant - Reserves Act - May 2015 - DOCDM-1599758 (1) DM1599758 Godley Peaks Conservation Covenant - Reserves Act - May 2015

- within tussock grassland areas will include:
 - observations of stock damage to the vegetation, impacts of trampling and browsing and impacts on regeneration, along with any loss of biodiversity that may occur as a direct result of grazing.

Permanent Plot Momtoring

A series of up to a total of 10 permanently marked 20 x 20m plots may be established across the Land; with the exact number to be decided after a detailed analysis of altitude, aspect and vegetation community factors. Site selection will occur as part of re-monitoring if a need for them is identified, but will use the criteria set out in this document. The location of these plots is to be targeted on fragile ecosystems where stock grazing is likely to be most pronounced; as decided by the Minister and in consultation with the owner. These sites may be paired with a second permanently marked 20 x 20m exclosure plot located nearby (within 50m). This paired plot will need to be fenced to exclude stock. This paired site will not be in addition to the 10 permanently marked plots

Each site will be marked using a warratah at the top NW corner and aluminium pegs used to mark the other corners.

For each plot, recording will include the aspect and slope, a compiled full list of vascular species and a cover estimate (not cover classes) for each species listed.

Within each plot monitoring will include height-frequency sampling using four equally spaced transects. The height-frequency of plants is to be sampled at 0.5m along each transect, and follow the methodology described by Wiser and Rose (1997).

Land Management

The Owner will maintain a record of stock numbers grazed within the Land, as well as any other significant factors affecting the stocking and stock management of the Land. This could include major stock trespass from broken fences.

The Owner will keep climate records as appropriate including rainfall.

4. Monitoring Results:

Following monitoring, results will be discussed between the Owner and the Minister.

Should it be noted as a result of monitoring that grazing or other management practices are having a detrimental impact on the values then the Minister will take significant steps to prevent this continuing, which may include such measures as fencing, or reducing stock numbers. These measures will be undertaken at the owner's cost and with no compensation payable

• For clarity, a 10% decline of a significant species in a permanent plot, relative to the exclosure plots would be considered significant enough to trigger this active management (i.e. the unfenced permanent plot going from 50% cover to 45% cover, and fenced going from 50% cover to 55%, would be a 10% difference). This way if both fenced and unfenced decline, as could happen, there'd be no reduction in grazing.

<u>GRANT</u> of

Correct for the purposes of the Land Transfer Act 1952 Solicitor for the Minister

.

CONSERVATION COVENANT UNDER SECTION 77 OF THE RESERVES ACT 1977 FOR CROWN PASTORAL LAND ACT 1998 PURPOSES

COMMISSIONER OF CROWN

to

MINISTER OF CONSERVATION

Solicitor

Department of Conservation DUNEDIN/CHRISTCHURCH

Godley Peaks Pastoral Lease - Conservation Covenant - Reserves Act - May 2015 - DOCDM-1599758 (1) DM1599758 Godley Peaks Conservation Covenant - Reserves Act - May 2015 Jan 2013

Execution Section

This Proposal (including the schedules and appendices) is signed by the Commissioner and the Holder as a binding agreement.

SIGNED for and on behalf of the Commissioner of Crown Lands by [] pursuant to a delegation under the Crown Pastoral Land Act 1998 in the presence of:

Witness

Occupation

Address

SIGNED for and on behalf of Verity Farms NZ by it's sole director:

Ann Harriet Poindexter

In the presence of:

Witness

Occupation

Address

3 Godley Peaks Preliminary Proposal 30092015 3 February 2014 A711868